Design, Fabrication, and Characterization of Germanium MOSFETs with High-k Gate Dielectric Stacks based on Nitride Interfacial Layers

by

Andrew P. Ritenour

B.S. University of Virginia (1997) S.M. Massachusetts Institute of Technology (1999)

Submitted to the Department of Electrical Engineering and Computer Science in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

May 2007 [June 2007] © Massachusetts Institute of Technology, 2007. All Rights Reserved.

Author	•••••	· • • • • • • • • • • • • • • • • • • •
	Elecurca Engineering	Science
		May 21, 2007

Certified by

Professor of Flectrical Erain

Dimitri A. Antoniadis ience visor

Accepted by

Arthur C. Smith

Chairman, Department Committee on Graduate Students

AUG 1 6 2007

BARKER

Design, Fabrication, and Characterization of Germanium MOSFETs with High-k Gate Dielectric Stacks based on Nitride Interfacial Layers by

Andrew P. Ritenour

Submitted to the Department of Electrical Engineering and Computer Science on May 21, 2007, in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Electrical Engineering

Abstract

To improve source injection velocity, and consequently MOSFET performance, high mobility semiconductors are being explored as possible replacements for silicon. Germanium offers enhanced electron mobility and superior hole mobility at high inversion charge density; however, the formation of a high quality germanium-dielectric interface remains a serious challenge. High-k dielectrics deposited directly on germanium exhibit poor physical and electrical properties so an interfacial layer is required. Proposed interlayers include GeON, Si, and metal nitrides such as AlN and Hf_3N_4 .

This work focuses on the fabrication and characterization of germanium MOSFETs with GeON, Hf_3N_4 , and AlN interlayers. $WN/Al_2O_3/AlN$ gate stacks deposited by atomic layer deposition (ALD) were investigated in detail. The impact of AlN interlayer thickness and post-metal anneal conditions on the electrical properties of $WN/Al_2O_3/AlN/Ge$ capacitors was determined. Optimal capacitance-voltage characteristics were achieved for an AlN thickness of 2.5 nm and 450-500 °C post-metal annealing. Ge n- and p-MOSFETs were fabricated with GeON, AlN, and Hf_3N_4 interlayers. The hole mobility of these devices generally matched or exceeded silicon universal hole mobility; however, Ge n-FETs showed poor electron mobility (50-100 cm²/Vs).

Many theories have been proposed to explain the low carrier mobility of Ge n-FETs. These theories were investigated and it was found that an asymmetric distribution of interface states in the bandgap is the primary cause of low electron mobility for germanium-AlN interfaces. The interface state density near the conduction band edge approaches 6×10^{13} cm⁻² eV⁻¹, approximately 5x higher than near the valence band edge. Low temperature characterization of n-FETs revealed degraded electron mobility due to carrier trapping and coulomb scattering from charged interface states. To reduce the interaction of carriers with interface states, n- and p-MOSFETs with reduced vertical effective field were fabricated using ion implantation. Devices exhibiting buried channel behavior showed electron and hole mobilities of 600 and 300 cm²/Vs respectively, confirming that mobility degradation is caused by interface states. Evidence for phosphorus passivation of the germanium-AlN interface is also presented.

Thesis Supervisor: Dimitri A. Antoniadis Title: Ray and Maria Stata Professor of Electrical Engineering

Acknowledgments

First I would like to thank Dimitri for serving as my advisor for eight out of the last ten years. He always gives his students the benefit of the doubt (and I certainly needed it a time or two). He was a great mentor and friend over the years and much of what I learned from him isn't related to the physics of MOSFETs. I clearly would not have reached this point without his help. I would also like to thank Tayo Akinwande and Gene Fitzgerald for serving on my thesis committee. I knew Tayo and Gene long before they were on my committee, and I certainly appreciate their help and guidance both before and during the thesis process. I've learned a great deal from both of them.

The completion of the Ph.D. process would not have been possible without the help, friendship, and support of many people. The MTL staff members (Bernard, Bob, Eric, Donal, Dave, Dennis, Paudely, Kurt, Vicky, Paul T., Tim, Ryan, Dan, Paul M., Pat B., Pat V., Scott, and Gary) were always there to help. When I was pressed to get something done, I could always count on them for a favor. All past and present members of the DAA group made things a lot easier, from the senior students that showed me around when I first arrived at MIT (Andy, Tony, and Mark) to the guys in the group now (John, Jae-kyu, Ali, and Osama). John deserves a lot of credit for helping with the installation and characterization of the ALD reactor. I would also like to acknowledge our outside collaborators, particularly Kevin Kim and Professor Roy Gordon of Harvard University for depositing ALD gate stacks for us and teaching us most of what we know about ALD. This work also wouldn't have been possible without the help of Douwe Monsma and Jill Becker at Cambridge NanoTech. They delivered a great ALD reactor and always provided us with tech support when we needed it. I also truly benefitted from the post-docs and visiting scientists in the group (Larry, Scott, and Ryan). Each of them made major contributions to the germanium research project and were also good friends. Speaking of friends, I'm sure I would have gone crazy without a few of those. Fanman, Lauer, Fletch, Hennessy, Wei, Dennis, Tonya, Jim, Little F, Niamh, Cait, Ingvar, Gregg, Mark, Tony, Ryan, Larry, Arthur, Mayank, and a bunch of other people helped keep things interesting. There are a few classic stories and those who were involved know exactly what I'm talking about. Marilyn Pierce also deserves special thanks for keeping track of me over the years. When returning from my two year leave of absence, I asked Marilyn what I needed to do to rejoin the department and she gave me a form to complete. All it required was my name, address, and signature. And of course, this thesis was not ready by the official due date so I benefitted from the Marilyn deadline.

Last but most importantly, I wouldn't be where I am without the support of a wonderful family. They gave me this opportunity and I'm glad I was able to make something of it. Hopefully I'll be able to repay all of the love, effort, time, (and money) that they've invested in me over the years. Mommy and Daddy, Kimberly, Bruce, Meredith, Brian, Lindsey, Chris, Alex, Mason, Sarah and all of my grandparents, aunts, and uncles deserve a lot of credit, much more than I can possibly include in these acknowledgments.

Table of Contents

1	1 Introduction17					
	1.1 Historical MOSFET Trends17					
	1.2 MOSFET Performance					
	1.3 The Advantages of Germanium					
	1.4	The Challenges of Germanium	.28			
	1.5	The Germanium-Dielectric Interface	.31			
	1.6	Historical Germanium MOSFET Performance	.38			
	1.7	Goals of Thesis	.41			
	1.8	Organization of Thesis	.41			
2	Atom	nic Layer Deposition of High-k Dielectric Stacks on Germanium	.43			
	2.1	Introduction	.43			
	2.2	Brief Review of Atomic Layer Deposition	.44			
	2.3	ALD Gate Stack Films	.47			
	2.4	Review of MOS Capacitor Electrical Characterization	.48			
	2.5	ALD Gate Stack Capacitor Fabrication and Characterization	.51			
	2.6	Summary	.76			
3	Surfa	ce-Channel Ge MOSFETs	79			
	3.1	Introduction	79			
	3.2	AlN Interlayer MOSFETs	80			
	3.3	Hf ₃ N ₄ Interlayer MOSFETs	87			
	3.4	RTP GeON Interlayer MOSFETs	90			
	3.5	MBE GeON Interlayer MOSFETs	96			
	3.6	Summary of MIT Ge MOSFET Mobility	96			
4	Inves	stigation of Carrier Transport in Ge MOSFETs with AlN Interlayers	99			
	4.1	The Ge NMOS Problem	99			
	4.2	Interface Distribution for Germanium-AlN Interfaces	101			
	4.3	Low Temperature Mobility Characterization	105			
	4.4	Buried-Channel MOSFETs	108			
	4.5	Summary	116			
5	Conc	lusions	117			
	5.1	Summary	117			
	5.2	Future Work	120			
$\mathbf{A}_{\mathbf{j}}$	ppend	lix A MOSFET Process Flow	123			
A	ppend	lix B Agilent 4294A IBASIC C-V Measurement Program	129			
A	ppend	lix C C-V Characteristics of Al/Al ₂ O ₃ /AlN/Ge Capacitors	131			
A]	Appendix D C-V Characteristics of WN/Al ₂ O ₃ /AIN/Ge Capacitors					
A	ppend	lix E C-V Characteristics of WN/GdScO ₃ /Hf ₃ N ₄ /Ge Capacitors	143			
B	ibliog	raphy	145			

List of Figures

Figure 1.1: Actual virtual source velocity, v_{xo} , for commercial CMOS technologies with gate lengths ranging from 480 nm to 35 nm. Source velocities required to maintain histor-Figure 1.2: Representative band structure for diamond and zincblende semiconductors (from [6]). The band structure depicted in this figure corresponds to Ge (conduction band Figure 1.3: Brillouin zone for diamond (C, Si, Ge) and zincblende semiconductors (GaAs). Conduction band minima for Si, Ge, and GaAs occur at Δ , L, and Γ points, respectively Figure 1.4: Equivalent circuit model for MOS inversion layer. Inversion layer capacitance (Cinv) is dominated by DOS at low inversion charge density and inversion layer thickness at high inversion charge density (adapted from [7])......23 Figure 1.5: Definition of axes for m_x , m_y , and m_z effective masses. The x-axis is oriented in the transport direction, the y-axis is oriented in the width direction, and the z-axis is ori-Figure 1.6: Schematic representation of band-to-band tunneling for a reverse-biased p-n Figure 1.7: Carrier mobility at 0.3 MV/cm for Ge MOSFETs reported since 2002. Silicon universal electron and hole mobility are shown for reference. Results that have been difficult to reproduce and are not consistent with other published data are denoted by black cir-Figure 2.1: Schematic representation of reactions that comprise an ALD cycle. The reaction between trimethylaluminum (TMA) and water to form Al₂O₃ is used as an example (adapted from [55])......46 Figure 2.2: Small-signal equivalent circuit model for MOS capacitor with interface states Figure 2.3: Common non-idealities observed in MOS capacitors. (A) Midgap D_{it} distortion, DC stretch-out, V_{FB} shift, and frequency dispersion. (B) Hysteresis and trap-assisted Figure 2.4: CET of Al/Al₂O₃/AlN/p-Ge capacitors vs. total number of ALD cycles. Y-in-Figure 2.5: Forward and reverse sweep C-V characteristics of as-deposited WN/(Al₂O₃, AIN)/Ge capacitors at 50 kHz, 100 kHz, and 1 MHz. (A) 60/00 Al₂O₃/AIN (B) 45/15 Figure 2.6: Gate leakage for WN/ 35 cycle Al₂O₃/ 25 cycle AlN gate stacks on n- and p-Figure 2.7: Metrics used to compare C-V characteristics of MOS capacitors. (A) Hysteresis extracted at 20% (H-20) and 80% (H-80) of C/C_{max} and (B) ΔQ extracted from separately normalized 50 kHz and 1 MHz forward-sweep C-V characteristics...... 59

Figure 2.8: ΔQ vs. anneal temperature for n- and p-Ge capacitors with 1.5 and 2.5 nm AlN
interlayers. (A) 60 s N ₂ annealing. (B) 30 min. FG annealing. n-Ge and p-Ge capacitors are
denoted by closed and open symbols, respectively. 1.5 nm and 2.5 nm AlN interlayer thick-
nesses are indicated by circles and squares, respectively. The N ₂ anneal data point at 350
°C is from the as-deposited sample
Figure 2.9: Hysteresis at 20% C/C _{max} vs. anneal temperature for n- and p-Ge capacitors
with 1.5 and 2.5 nm AlN interlayers. (A) 60 s N ₂ annealing. (B) 30 min. FG annealing. n-
Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. 1.5 nm and
2.5 nm AlN interlayer thicknesses are indicated by circles and squares, respectively. The
N ₂ anneal data point at 350 °C is from the as-deposited sample
Figure 2.10: Hysteresis at 80% C/C _{max} vs. anneal temperature for n- and p-Ge capacitors
with 1.5 and 2.5 nm AlN interlayers. (A) 60 s N ₂ annealing. (B) 30 min. FG annealing. n-
Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. 1.5 nm and
2.5 nm AlN interlayer thicknesses are indicated by circles and squares, respectively. The
N ₂ anneal data point at 350 °C is from the as-deposited sample
Figure 2.11: Difference in hysteresis at 80% and 20% C/C _{max} vs. anneal temperature for
n- and p-Ge capacitors with 1.5 and 2.5 nm AlN interlayers. (A) $60 \text{ s } N_2$ annealing. (B) 30
min. FG annealing. n-Ge and p-Ge capacitors are denoted by closed and open symbols, re-
spectively. 1.5 nm and 2.5 nm AlN interlayer thicknesses are indicated by circles and
squares, respectively. The N ₂ anneal data point at 350 °C is from the as-deposited sam-
ple64
Figure 2.12: Summary of important trends from PMA study on n- and p-Ge capacitors with
1.5 and 2.5 nm AlN interlayers. Metrics are shown in each column while row 1 and 2 sum-
marize the results for N_2 and FG annealing, respectively. The impact of AlN interlayer (IL)
thickness on the given metric is shown in the third row
Figure 2.13: C-V characteristics for WN/3.5nm Al ₂ O ₃ /2.5 nm AlN/Ge capacitors annealed
at 450 °C for 30 min. in FG66
Figure 2.14: Interface state density (D _{it}) determined using the conductance method for
WN/ 3.5 nm Al ₂ O ₃ / 2.5 nm AlN/Ge gate capacitors that received a 30 min. 450 °C FG an-
neal. This method is accurate from flatband to weak inversion, so n-Ge and p-Ge were used
to extract D _{it} in the lower and upper halves of the bandgap, respectively. The energy posi-
tion in the bandgap is referenced to the intrinsic Fermi level
Figure 2.15: Forward sweep 1 MHz C-V characteristics as a function of temperature for
WN/ 3.5 nm Al ₂ O ₃ / 2.5 nm AlN gate stack on p- and n-Ge. The n-Ge capacitor shows a
larger flatband shift indicating a higher density of states in the upper half of the band-
$gap \dots 71$
Figure 2.16: D_{it} distribution for WN/ 3.5 nm Al ₂ O ₃ / 2.5 nm AlN/Ge capacitors extracted
using conductance and Gray-Brown methods. The energy position in the bandgap is refer-
enced to the intrinsic Fermi level
Figure 2.17: C-V characteristics of WN/ 3.5 nm $Al_2O_3/2.5$ nm AlN/n-Ge capacitors at 297,
250, 200, 150, 100, and 80 K. Forward and reverse sweeps are shown at frequencies of 50
KHZ, $IUU KHZ$, and $I MHZ$
Figure 2.18: C-V characteristics of WN/3.5 nm $Al_2O_3/2.5$ nm AlN/p -Ge capacitors at 297,
250, 200, 150, 100, and 80 K. Forward and reverse sweeps are shown at frequencies of 50
KHz, 100 kHz, and 1 MHz

Figure 2.19: Direct comparison of low temperature C-V characteristics of WN/ 3.5 nm Al₂O₃/2.5 nm AlN gate stack on n- and p-Ge at 297, 250, and 200 K. n-Ge shows a significant reduction in peak capacitance at high frequency and/or low temperature due to trapping74 Figure 2.20: Integrated charge from 297 K 50 kHz and 1 MHz C-V characteristics of WN/ 3.5 nm Al₂O₃/ 2.5 nm AlN/n-Ge capacitors. The charge difference is attributed to carrier trapping. The percentage of trapped charge is plotted on the primary axis while integrated charge is plotted on the secondary axis.75 Figure 2.21: Quasistatic and high-frequency C-V data for WN/3.5 nm Al₂O₃/2.5 nm AlN Figure 3.1: Typical as-measured I_S-V_G characteristics for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. p-FETs exhibited a 95 mV/decade subthreshold swing and V_t of -0.05 V. n-FETs exhibited a V_t of +0.7 V, gradual roll-off of the Is-VG characteristic in the subthreshold regime, and increased device-to-device variabili-Figure 3.2: Typical as-measured I_S-V_D characteristics for Ge p- and n-MOSFETs with 40 Figure 3.3: Typical as-measured inversion-side split C-V characteristics for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stack. The peak inversion capacitance corresponds to a capacitance equivalent thickness (CET) of approximately 3 nm and is similar for both the n-FET and p-FET......84 Figure 3.4: Effective carrier mobility for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. p-FETs show slight enhancement compared to silicon universal while n-FETs show peak mobility of only 50 cm²/Vs......84 Figure 3.5: Comparison of effective electron mobility for (100) and (111) Ge n-MOSFETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. Mobility from five devices is shown Figure 3.6: Typical as-measured I_{S} -V_G characteristics for Ge p- and n-MOSFETs with 40 nm WN/2.5 nm GdScO₃/1.5 nm Hf₃N₄ gate stack. p-FETs exhibited a 114 mV/decade subthreshold swing and -0.65 V threshold voltage (V_t). The gradual I_S-V_G roll-off and high V_t indicate a large density of states in the lower half of the bandgap. The n-FETs exhibited a V_t of +0.45 V and steeper subthreshold swing of 103 mV/decade......88 Figure 3.7: Typical as-measured split C-V characteristics at 10 kHz for Ge p- and n-MOS-FETs with 40 nm WN/2.5 nm GdScO₃/1.5 nm Hf₃N₄ gate stack. The peak inversion capacitance corresponds to a capacitance equivalent thickness (CET) of approximately 1.5 nm Figure 3.8: Effective carrier mobility for Ge p- and n-MOSFETs with 40 nm WN/2.5 nm $GdScO_3/1.5$ nm Hf_3N_4 gate stack. n-FETs show promising peak electron mobility of 90 Figure 3.9: Schematic of Ge/SiGe heterostructure grown using UHVCVD. The strained germanium layer is approximately 55 nm thick. All SiGe layers are relaxed. After growing the 0-60% SiGe graded buffer, the wafers were polished using CMP to remove the crosshatch that arises during growth (from [61])......91

Figure 3.10: (A) XTEM of strained Ge after processing. The strain present in the film may be estimated from the dislocation density as seen in XTEM micrographs [64]. Applying this technique yields a strain of 0.8% in the germanium channel layer. (B) XTEM of gate stack on strained Ge after processing. The HfO_2 layer is approximately 5 nm thick. The ger-Figure 3.11: Typical as-measured I_{S} -V_G characteristics for bulk Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. These devices exhibited a -0.19 V threshold Figure 3.12: Typical as-measured split C-V characteristics for bulk Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. The inversion-side capacitance corre-Figure 3.13: Typical as-measured I_S-V_G characteristics for strained Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. These devices have a -0.23 V threshold voltage and 90 mV/decade subthreshold swing (from [61])......94 Figure 3.14: Typical as-measured split C-V characteristics for strained Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. The inversion-side capacitance cor-Figure 3.15: Comparison of effective hole mobility for bulk and strained Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. At high vertical effective field, strained Ge p-FETs show a 35% increase in mobility compared to the bulk Ge devices Figure 3.16: Summary of carrier mobility for nitride interlayer Ge MOSFETs fabricated at MIT. Representative data shown for p-FETs (open symbols) and n-FETs (closed sym-Figure 4.1: Comparison of C-V characteristics for electron inversion on p-Ge (MOSFET) and electron accumulation on n-Ge (MOSCAP). Both devices had WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stacks. MOSFET split C-V characteristics show higher baseline capacitance (dashed line) due to parasitic probe pad capacitance that is not modulated by the gate.100 Figure 4.2: I_S-V_G characteristics as a function of temperature (297, 250, 200, 150, 100, and 80 K) for Ge p- and n-FETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. The large V_t shift for n-FETs relative to p-FETs suggests higher D_{it} in the upper half of the bandgap, particularly near the conduction band edge.....104 Figure 4.3: Inversion-side split C-V characteristics (C_{GSD}) as a function of temperature (297, 250, 200, 150, 100, and 80 K) for Ge p- and n-FETs with 40 nm WN/3.5 nm Al₂O₃/ Figure 4.4: Interface state distribution for WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack on germanium. Conductance, Gray-Brown, and temperature dependence of MOSFET V, were used to map D_{it} across the bandgap. D_{it} rises rapidly near the conduction band edge and approaches 6×10^{13} cm⁻² eV⁻¹. The number of additional interface states near the CBE relative to the VBE (indicated by dotted triangle) is approximately $2x10^{12}$ cm⁻² which is similar to the trapped charge density estimated from the low temperature MOSCAP measurements in Chapter 2......105

Figure 4.5: Carrier mobility as a function of temperature (297, 250, 200, 150, 100, and 80 K) for Ge p- and n-FETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. Electron mobility decreased with decreasing temperature indicating significant charge trapping in interface states near the conduction band edge. Distortion of the electron mobility curves can be attributed to coulomb scattering from charged interface states......106 Figure 4.6: Hole mobility for SiGe surface-channel p-MOSFETs with ALD TiN/Al₂O₃/ HfAlO_x/Al₂O₃ gate stack. As-extracted mobility is compared with mobility corrected for carrier trapping (Q_i), carrier trapping plus coulomb scattering from 1×10^{12} cm⁻² interface states, and carrier trapping plus coulomb scattering from 4×10^{12} cm⁻² interface states (from Figure 4.7: Measured I_S -V_G characteristics for long-channel Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , 4×10^{12} , and 1×10^{13} cm⁻² channel implants. In contrast to p-FETs, n-FETs show decreasing subthreshold swing as implant dose increases from 1×10^{12} to 4×10^{12} cm⁻².....110 Figure 4.8: Measured inversion-side split C-V characteristics (C_{GSD}) for Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , 4×10^{12} , and 1×10^{13} cm⁻² channel implants. p-FETs show C-V stretch-out associated with transition from surface-channel to buriedchannel operation as implant dose increases from 1×10^{12} to 1×10^{13} cm⁻². In contrast, n-FETs show a simple shift of C_{GSD} characteristics for low implant doses $(1 \times 10^{12} - 4 \times 10^{12})$ cm⁻²) with only the highest implant dose $(1 \times 10^{13} \text{ cm}^{-2})$ showing evidence of buried-chan-Figure 4.9: Effective carrier mobility for Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , 4×10^{12} , and 1×10^{13} cm⁻² channel implants. Peak mobilities of 300 and 600 cm^2/Vs were observed for p- and n-FETs, respectively. Inset of (b) shows electron mobility on reduced vertical scale to provide a better view of carrier mobility in Ge n-FETs with 1×10^{12} - 4×10^{12} cm⁻² channel implants. Mobility was extracted using the standard split C-V method for devices that could be sufficiently turned off with gate bias......111 **Figure 4.10:** Effective mobility for 1×10^{13} cm⁻² dose buried-channel n-FETs with peak CETs of 4.5 and 6 nm. Mobility for 1×10^{12} cm⁻² dose n-FET with peak CET of 3 nm is shown for reference. For n-FET with 6 nm CET, inversion charge was determined using both split C-V and Hall measurements.....113 Figure 4.11: Capacitance equivalent thickness (CET) vs. inversion charge density for Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , and 4×10^{12} cm⁻² channel implants. p-FETs show evidence of reduced Eeff and/or buried-channel operation while n-FETs appear to be surface-channel devices. Plots were obtained by integrating the split C-Figure C.1: C-V characteristics for (A) 50/10, (B) 50/20, and (C) 50/30 Al₂O₃/AlN stacks on p-Ge at 100 kHz and 1MHz. Left column shows as-deposited characteristics while right Figure C.2: C-V characteristics for (A) 50/10, (B) 50/20, and (C) 50/30 Al₂O₃/AlN stacks on n-Ge at 100 kHz and 1MHz. Left column shows as-deposited characteristics while right Figure C.3: C-V characteristics of 50/20 n-Ge capacitors after (A) nitrogen and (B) form-

Figure D.1: Impact of PMA on C-V characteristics of WN/45 cycle $Al_2O_3/15$ cycle $AlN/$
p-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C
samples are shown. Left column shows characteristics from 60 s N2 anneal while right col-
umn shows 30 min. FG anneal
Figure D.2: Impact of PMA on C-V characteristics of WN/35 cycle $Al_2O_3/25$ cycle $AlN/$
p-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C
samples are shown. Left column shows characteristics from 60 s N2 anneal while right col-
umn shows 30 min. FG anneal
Figure D.3: Impact of PMA on C-V characteristics of WN/45 cycle $Al_2O_3/15$ cycle $AlN/$
n-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C
samples are shown. Left column shows characteristics from 60 s N ₂ anneal while right col-
umn shows 30 min. FG anneal 141
Figure D.4: Impact of PMA on C-V characteristics of WN/35 cycle $Al_2O_3/25$ cycle $AlN/$
n-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C
samples are shown. Left column shows characteristics from 60 s N ₂ anneal while right col-
umn shows 30 min. FG anneal142
Figure E.1: Quasistatic, 10 kHz, and 1 MHz C-V characteristics for WN/8 nm GdScO ₃ /
1.5 nm Hf_3N_4 /p-Gc capacitor (from [41])
Figure E.2: D _{it} extracted using both the conductance and quasistatic methods [59] for a
WN/8 nm GdScO ₃ /1.4 nm Hf ₃ N ₄ /p-Gc capacitor (from [41])

List of Tables

Table 1.1: Bulk carrier mobility for Si, Ge, GaAs, and InAs at 300K [5]	20
Table 1.2: Longitudinal and transverse effective masses for Δ and L valleys of Si and C	Зe
(adapted from [8])	25
Table 1.3: Conductivity (m_x) , quantization (m_z) , and width (m_y) effective masses of Δ and	nd
L valleys for (100), (111), and (110) substrate orientations (from [8]).	25
Table 1.4: Bandgap at 300K and dielectric constant of Si and Ge. 2	28
Table 1.5: State-of-the-art sheet resistance values for boron- and phosphorus-implanted	1
germanium [18]	31
Table 1.6: Gate stack materials and deposition method for mobility results summarized	in
Figure 1.7.	39
Table 2.1: Dielectric stack splits for Al/Al ₂ O ₃ /AlN/Ge capacitors. Stacks are defined b	у
number of Al ₂ O ₃ cycles and number of AlN cycles (Al ₂ O ₃ cycles/ AlN cycles)	52
Table 2.2: Capacitance equivalent thickness for Al/Al ₂ O ₃ /AlN/Ge capacitors	53
Table 2.3: Dielectric stack splits for WN/(Al ₂ O ₃ , AlN)/Ge capacitors. Stacks are define	ed
by number of Al ₂ O ₃ cycles and number of AlN cycles (Al ₂ O ₃ cycles/ AlN cycles)	55
Table 2.4: Post-metal anneal splits for WN/Al ₂ O ₃ /AlN/Ge capacitors. FG= 8:1	
N ₂ :H ₂	55
Table 3.1: Fabrication process summary for AlN interlayer Ge MOSFETs. Detailed pro	- C
cess description may be found in Appendix A.	31
Table 4.1: Simulated depth and peak concentration for buried-channel implants. Depth	is
given with respect to the germanium surface. P and BF2 implant energies were adjusted	to
give the same depth 1	.09
Table 4.2: Comparison of peak electron and hole mobility measured in buried-channel	
MOSFETs with bulk conductivity mobility of n- and p-Ge. Bulk conductivity mobility w	'as
calculated assuming a doping of 2×10^{18} cm ⁻³ 1	12
Table C.1: Dielectric stack splits for Al/Al ₂ O ₃ /AlN/Ge capacitors. Stacks are defined b	эу
number of Al ₂ O ₃ cycles and number of AlN cycles (Al ₂ O ₃ cycles/ AlN cycles) 1	31
Table C.2: Post-metal anneal splits for $Al/Al_2O_3/AlN/Ge$ capacitors. FG= 8:1	
N ₂ :H ₂ 1	32
Table D.1: Dielectric stack splits for WN/(Al ₂ O ₃ , AlN)/Ge capacitors. Stacks are defin	ed
by number of Al_2O_3 cycles and number of AlN cycles (Al_2O_3 cycles/ AlN cycles) 1	37
Table D.2: Post-metal anneal splits for $WN/Al_2O_3/AlN/Ge$ capacitors. FG= 8:1	
N ₂ :H ₂ 1	38

Chapter 1

Introduction

1.1 Historical MOSFET Trends

Since its inception, the integrated circuit (IC) industry has been able to maintain an impressive set of trends governing device density and performance. Silicon-based MOS-FET technology has provided an enabling platform for continuous improvement in density, switching speed, and cost per transistor. Since the 1970s, transistor density has doubled every 18 months. Dubbed Moore's Law, this trend reflects the tremendous impact of continuously reducing transistor dimensions and adjusting circuit parameters in a coordinated manner. In addition to gains in transistor density, the intrinsic switching delay for high-performance logic has historically decreased 17% per year [1]. The international semiconductor industry has developed a roadmap (*International Technology Roadmap for Semiconductors*) for extending these trends through 2020. Ultimately, technology R&D within the international semiconductor industry is geared towards maintaining these trends for as long as possible.

1.2 MOSFET Performance

Following the analysis of Antoniadis et al. [2], the intrinsic MOSFET switching delay, τ , is described by:

$$\tau = \frac{\Delta Q_G}{I_{eff}} \tag{1.1}$$

where ΔQ_G is the charge difference between logic states and I_{eff} is the effective switching current. The switching delay is inversely proportional to the *effective* virtual source velocity, v, which is defined as the effective carrier velocity at the virtual source. The virtual source is defined as the point at the top of the source-to-channel potential barrier where carriers are injected into the channel. The effective virtual source velocity may be extracted using a simple model that accurately describes the drain saturation current, I_D , of state-of-the-art MOSFETs [2]:

$$\frac{I_D}{W} = C'_{oxinv} (V_G - V_t) v = q \cdot n_{inv} \cdot v$$
(1.2)

where W is the device width, C'_{oxinv} is the inversion capacitance per unit area, n_{inv} is the inversion layer carrier concentration per unit area, V_G is the applied gate voltage, and V_t is the threshold voltage determined by linear extrapolation of the I_D-V_G curve in saturation. Due to series resistance between the source contact and virtual source, equation 1.2 only provides the effective source velocity. The *actual* virtual source velocity, v_{xo} , is related to the effective velocity through the following expression [2]:

$$v_{xo} = \frac{v}{[1 - C'_{oxinv} R_s W (1 + 2\delta) v]}$$
(1.3)

where R_s is the parasitic access resistance between the source contact and virtual source and δ is the drain induced barrier lowering (DIBL) coefficient.

Antoniadis et al. extracted v_{xo} for bulk Si MOSFETs with gate lengths (L_G) ranging from 480 nm to 35 nm and the results are shown in Figure 1.1.



Figure 1.1: Actual virtual source velocity, v_{xo} , for commercial CMOS technologies with gate lengths ranging from 480 nm to 35 nm. Source velocities required to maintain historical performance trends at 10 nm gate length are also shown (from [2]).

Figure 1.1 shows that source velocities have been increasing for recent MOSFET technology generations. It is clear that continuous improvement in source velocity has played a large role in maintaining the historical reduction in intrinsic MOSFET delay.

Lundstrom has developed a theory that relates v_{xo} to backscattering near the virtual source through the following expression [3]:

$$v_{xo} = v_T \left(\frac{1 - r_c}{1 + r_c}\right) \tag{1.4}$$

where r_c is the backscattering coefficient and v_T is the thermal velocity. Lundstrom showed that backscattering is reduced by increasing low-field carrier mobility or increasing the lateral electric field at the channel-side of the virtual source. Simple backscattering theory can be used to explain the source velocity trend shown in Figure 1.1. In this figure, three separate regimes can be identified: $L_G > 130$ nm, 60 nm $< L_G < 130$ nm, and $L_G <$ 60 nm [2]. For L_G greater than 130 nm, low-field carrier mobility remained constant at the unstrained bulk Si value so v_{xo} improvement was due to the increased lateral source field produced by device scaling. From 130 nm to 60 nm, v_{xo} saturated due to increased coulomb scattering from the high doping concentrations required to maintain electrostatic integrity. From 60 nm onwards, process-induced strain has boosted low-field mobility and consequently v_{xo} . Recent work has shown that the correlation ratio between virtual source velocity and low-field mobility is 0.85 [4]. As shown in Figure 1.1, very high source injection velocities will be required to maintain historical performance trends for future technology generations. Given the direct relationship between low-field mobility and v_{xo} , high mobility semiconductors show promise for future CMOS devices.

1.3 The Advantages of Germanium

As shown in Table 1.1, germanium offers superior bulk hole mobility (approximately 4x enhancement relative to Si) while still offering 2.4x enhancement in electron mobility [5].

Semiconductor	Mobility at 300K (cm ² /Vs)		
Semiconductor	Electrons	Holes	
Si	1500	450	
Ge	3900	1900	
GaAs	8500	400	
InAs	33000	460	

Table 1.1: Bulk carrier mobility for Si, Ge, GaAs, and InAs at 300K [5].

Additional advantages of germanium include high density of states (DOS) in the conduction band and relatively simple integration with silicon. Section 1.3.1 reviews the band structure of germanium and its relationship to DOS, inversion charge density, and mobility, Section 1.3.2 summarizes recent published simulation results comparing saturation drive current for germanium and III-V n-MOSFETs, and Section 1.3.3 briefly discusses practical issues related to incorporating new semiconductor materials into CMOS manufacturing.

1.3.1 Band Structure

Carrier mobility and inversion charge density are intimately linked to semiconductor band structure. Semiconductors with diamond and zincblende crystal structures are characterized by the representative band structure and Brillouin zone shown in Figures 1.2 and 1.3, respectively [6]. Conduction band minima occur at the center of the Brillouin zone (Γ point), at the zone edge along the <111> direction (L point), and near the zone edge along the <100> direction (Δ point). The band structure depicted in Figure 1.2 corresponds to Ge (conduction band minimum at the L point). Diamond and zincblende semiconductors have three valence bands, two of which are degenerate at the Γ point.



Figure 1.2: Representative band structure for diamond and zincblende semiconductors (from [6]). The band structure depicted in this figure corresponds to Ge (conduction band minimum at the L point).



Figure 1.3: Brillouin zone for diamond (C, Si, Ge) and zincblende semiconductors (GaAs). Conduction band minima for Si, Ge, and GaAs occur at Δ , L, and Γ points, respectively (from [6]).

1.3.2 Conduction Band

As described above, diamond and zincblende semiconductors have conduction band minima at the Γ , L, and Δ points in the Brillouin zone. At low electric field and in the absence of a confining potential well, carriers populate the lowest energy valley. At high electric field and/or in the presence of a confining potential well, energy separation between valleys and quantization effective mass determine the carrier population distribution. The conduction band minimum for III-V semiconductors occurs at the Γ point which is characterized by isotropic effective mass (m^*) . Germanium and silicon have conduction band minima at the L and Δ points, respectively. The effective mass in these valleys is anisotropic and may be characterized by an ellipsoidal constant energy surface with transverse (perpendicular to long axis) effective mass (m_i) and longitudinal (parallel to long axis) effective mass (m_l) . The DOS, conductivity, and quantization effective masses are functions of m_l and m_l , or just m^* for the Γ valley. For maximum drive current, low conductivity effective mass (high mobility), high DOS in the low effective mass valley (high n_{inv}), and high quantization mass (high n_{inv}) are desirable. Inversion charge density is a critical parameter for determining MOSFET performance and is related to both DOS and inversion layer thickness through the equivalent circuit shown in Figure 1.4 [7].



Figure 1.4: Equivalent circuit model for MOS inversion layer. Inversion layer capacitance (C_{inv}) is dominated by DOS at low inversion charge density and inversion layer thickness at high inversion charge density (adapted from [7]).

Inversion charge density is dominated by conduction band DOS at low charge density and inversion layer thickness at high charge density [7]. The crossover point occurs at a charge

density of approximately 1×10^{12} cm⁻². Inversion layer thickness is partly determined by energy level quantization in the 2D electron gas. Lower quantization mass translates into higher energy levels within the inversion layer potential well and consequently increased inversion layer thickness and decreased charge density.

The anisotropic effective mass of the Δ and L valleys in Si and Ge results in low conductivity effective mass and high DOS and quantization effective masses. Table 1.2 lists the transverse and longitudinal effective masses for the Δ and L valleys in Si and Ge [8]. Table 1.3 shows the effective mass in the x, y, and z directions (m_x, m_y, m_z) as a function of m_l and m_t for common substrate orientations [8]. As shown in Figure 1.5, the x-axis is oriented in the transport direction, the y-axis is oriented in the width direction, and the z-axis is oriented perpendicular to the wafer surface. Thus the conductivity, DOS, and quantization effective mass in Si and Ge leads to high DOS and quantization masses which gives them a substantial advantage over III-V semiconductors.



Figure 1.5: Definition of axes for m_x , m_y , and m_z effective masses. The x-axis is oriented in the transport direction, the y-axis is oriented in the width direction, and the z-axis is oriented perpendicular to the wafer surface (from [8]).

Material	Valley	m_l	m _t
Si	Δ	0.91	0.19
	L	1.70	0.12
Ge	Δ	0.95	0.20
	L	1.64	0.08

Table 1.2: Longitudinal and transverse effective masses for Δ and L valleys of Si and Ge (adapted from [8]).

(Wafer)/[transport] /[width]	Valley	m _x	mŗ	m _z	Degeneracy
	Δ	m,	<i>m</i> ,	m _i	2
		m _s	m,	m,	2
(001)/[100]/[010]		m,	m _i	ŝ	2
	Λ	$m_r \frac{(2m_l + m_s)}{(m_l - 2m_s)}$	$\frac{(m_i+2m_i)}{3}$	$\frac{3m_{t}m_{t}}{\left(2m_{t}+m_{t}\right)}$	4
	Δ	$\frac{(2m_i+m_{i})}{3}$	m.	$\frac{3m_im_i}{(2m_i+m_i)}$	2
		$\frac{2}{3}m_t\frac{(2m_t+m_t)}{(m_t+m_t)}$	$\frac{(m_t+m_t)}{2}$	$\frac{3m_{i}m_{r}}{\left(2m_{i}+m_{r}\right)}$	4
(111)/[211]/[011]	Δ	m,	m,	m _i	1
		$\frac{(8m_t+m_t)}{9}$	m,	$\frac{9m_im_i}{(8m_i+m_i)}$	1
		$\frac{m_t}{3} \frac{(8m_t+m_t)}{(2m_t+m_t)}$	$\frac{(2m_i+m_i)}{3}$		2
	Δ	m,	$\frac{(m_i+m_r)}{2}$	$\frac{2m_im_i}{(m_i+m_i)}$	4
		m,	m,	m,	2
(110)/[001]/[010]	Δ	$\frac{(m_i+2m_i)}{3}$	m,	$\frac{3m_im_i}{(m_i+2m_r)}$	2
	••	$\frac{3m_im_i}{(2m_i+m_i)}$	$\frac{(2m_i+m_i)}{3}$	m,	2

Table 1.3: Conductivity (m_x) , quantization (m_z) , and width (m_y) effective masses of Δ and L valleys for (100), (111), and (110) substrate orientations (from [8]).

1.3.3 Valence Band

As shown in Figure 1.2, diamond and zincblende semiconductors have three valence bands. Two are degenerate at the Γ point but have different curvatures and effective masses and are thus termed the light (upper) and heavy (lower) hole bands. The third band is split-off and lies at higher energy. Strong interaction between the light and heavy hole bands leads to non-parabolicity; therefore, simple parabolic band models fail to accurately describe hole transport [6] and the concept of effective mass is less straightforward [9]. Theoretical studies have shown that the germanium hole DOS effective mass is approximately half that of silicon [10]. Germanium also has a lower optical phonon energy than silicon (37 meV compared to 62 meV [11]) which limits carrier scattering to subbands within a narrower energy range [9]. Even without a simple effective mass model to describe hole transport in semiconductors, it is clear that germanium has superior hole mobility relative to other elemental and III-V semiconductors, and consequently is wellsuited for complementary logic.

1.3.4 Simulation Studies of Ge n-MOSFET Performance

Simulation studies of both bulk and double-gate ultra-thin body (DG UTB) n-MOSFETs with high mobility channels have recently been published [12], [13]. Ballistic drive current was compared for Si, Ge, GaAs, and other high electron mobility III-V semiconductors. The performance of III-V MOSFETs is limited by low inversion charge density and population of the lower mobility L valley [12]. The effects of low inversion layer capacitance are more pronounced for low equivalent oxide thickness (EOT). In general, germanium matches or outperforms III-V n-MOSFETs for EOT less than 1.5 nm for both bulk and DG UTB device geometrics. In addition, Ge MOSFETs fabricated on (111) substrates always outperform silicon MOSFETs [12]. The combination of improved mobility and high inversion charge density results in superior drive current and makes Ge MOSFETs an ideal candidate for future CMOS technologies.

1.3.5 Practical Considerations

In addition to matching or exceeding the device performance of III-V semiconductors, germanium also has practical advantages over these materials. Germanium is an elemental semiconductor and is processed similarly to silicon so existing infrastructure is well-suited for germanium device fabrication. Silicon and germanium use similar dopants and dopant activation methods and both form silicides/germanides which enables self-aligned device structures. Crystal growth techniques for germanium are also well established and largediameter germanium wafers are starting to become available; however, the world-wide supply of germanium is limited so germanium-on-insulator (GOI) is the most practical implementation. Any high mobility semiconductor will likely be integrated on a silicon platform so compatibility is an important consideration. Epitaxial growth of germanium on silicon results in a high quality interface without the risk of autodoping and cross-contamination that occurs when integrating III-V materials with Si or Ge. Furthermore, with the introduction of SiGe into the source-drain regions of Si p-MOSFETs, successful integration of germanium into high-volume CMOS manufacturing has already been demonstrated [14]. Unlike III-V semiconductors, germanium offers an evolutionary pathway to Ge CMOS, first as a source-drain material, and then perhaps as a buried-channel layer in p-FETs and finally a surface-channel layer for both n-FETs and p-FETs.

1.4 The Challenges of Germanium

In addition to the potential for improved drive current relative to silicon, germanium also presents new challenges due to differences in key material properties. Table 1.4 shows the bandgap and dielectric constant of Si and Ge [5].

Material Property	Si	Ge
Bandgap (cV)	1.12	0.66
Dielectric constant (ε ₀)	11.9	16

Table 1.4: Bandgap at 300K and dielectric constant of Si and Ge.

The narrow bandgap of germanium results in increased source-drain (S/D) junction leakage while its higher dielectric constant degrades electrostatic integrity. In addition, compared to Si, n-type dopants have lower solid solubilities and diffuse rapidly in germanium making it difficult to achieve shallow junctions with sufficiently low sheet resistance. But most importantly, native insulators on germanium (GeO₂ and GeO_xN_y) do not offer the same high quality interface provided by SiO₂ on Si. For CMOS applications, the success of any alternative semiconductor requires the development of a semiconductor-dielectric interface that matches the quality of the Si-SiO₂ interface.

1.4.1 Junction Leakage

The narrow bandgap of germanium leads to increased leakage in reverse-biased p-n junctions. Under low-field reverse bias, the large minority carrier concentration of germanium results in increased minority carrier drift current across the space charge region. Under high-field conditions, occupied states in the p-Ge valence band become energetically aligned with unoccupied states in the n-Ge conduction band. As shown schematically in Figure 1.6 [15], this condition allows electrons to tunnel from the p-side to the n-side, and hence is termed band-to-band tunneling.



Figure 1.6: Schematic representation of band-to-band tunneling for a reverse-biased p-n junction (from [15]).

Tunneling may either be direct or indirect depending on whether the conduction and valence band minima are aligned in k-space. The magnitude of the tunneling current is determined by the efficiency of the tunneling process, which is related to the carrier effective mass, barrier width, and density of states in each band. Germanium has both a narrower bandgap and smaller effective mass than silicon, and thus a larger band-to-band tunneling current (I_{BTBT}). I_{BTBT} is highly dependent on device geometry (channel and gate oxide thickness, gate configuration, etc.). Recent simulations have shown that I_{BTBT} for Ge MOSFETs is on the order of 0.1-1 μ A/ μ m [16], [17], which is at least several orders of magnitude higher than silicon but still comparable to the ITRS source-drain leakage target of 0.1-0.3 μ A/ μ m [1]. Furthermore, it has been suggested that band-to-band tunneling in germanium channel devices can be reduced through the use of strained Si/Ge heterostructures [16].

1.4.2 Electrostatic Integrity

Many narrow bandgap semiconductors, including germanium and most III-Vs, also have a higher dielectric constant than silicon which contributes to reduced electrostatic integrity in short-channel devices. Ultra-thin body and double-gate device geometries may be used

to reduce the impact of dielectric constant on short-channel electrostatics and thus improve scalability. For double-gate MOSFETs with reasonable geometry (15 nm gate length, 5-10 nm channel thickness, and 1 nm oxide thickness), the subthreshold swing of a germanium device is predicted to be 10-15% larger than a corresponding silicon device [17]. The degree of electrostatic degradation ultimately depends on the specific 3D device geometry; therefore, it is difficult to simply and universally speculate whether the higher dielectric constant of germanium will be a limiting factor for short-channel devices.

1.4.3 Dopant Diffusion and Activation

Dopant diffusion and activation also play a critical role in device performance, both in terms of electrostatic integrity and S/D series resistance (R_{SD}). Dopant diffusion must be controlled in order to reduce S/D junction depth and maintain sharp doping profiles. In addition, a high electrically-active dopant concentration is required to meet future R_{SD} requirements. Any new semiconductor must be capable of achieving R_{SD} values equal to or better than silicon (currently ~180 Ω -µm [1]).

Boron and phosphorus are the most common dopants used for germanium. While the solid solubility of boron in germanium is reportedly low $(5.5 \times 10^{18} \text{ cm}^{-3})$, active concentrations greater than 10^{20} cm^{-3} have been achieved [18]. Due to the low diffusivity of boron in germanium, as-implanted and post-anneal profiles match very closely enabling shallow junction formation. N-type dopants, on the other hand, exhibit higher diffusivity and lower peak active concentration. Phosphorus, the most promising n-type dopant in germanium, has a solid solubility of $2 \times 10^{20} \text{ cm}^{-3}$; however, to date, only peak active concentrations of $5 \times 10^{19} \text{ cm}^{-3}$ have been achieved [18]. Concentration-dependent diffusion and activation have also been reported for phosphorus activation at 600 °C [19]. At concentrations above $2 \times 10^{19} \text{ cm}^{-3}$, a box-like profile is observed and 20-50% of phosphorus atoms are electrically active. At concentrations below $2 \times 10^{19} \text{ cm}^{-3}$, no diffusion is

observed and 100% of phosphorus atoms are electrically active. Clustering is likely playing a role in the observed activation and diffusion behavior of phosphorus in germanium [18]. Table 1.5 shows state-of-the-art sheet resistance values for boron and phosphorus in Ge [18].

Dopant	Sheet Resistance (Ω/sq)	Junction Depth, x _j (nm)
В	85	70
Р	75	92

Table 1.5: State-of-the-art sheet resistance values for boron- and phosphorus-implanted germanium [18].

1.5 The Germanium-Dielectric Interface

Whether junction leakage, dopant activation, or electrostatic integrity ultimately limits germanium devices strongly depends on the specific device geometry and implementation. In contrast, a high quality germanium-dielectric interface is a fundamental requirement that is independent of device implementation. Despite the challenges associated with junction leakage, dopant activation, and electrostatic integrity, *the germanium-dielectric interface remains both the most important and most challenging aspect of germanium device engineering*.

1.5.1 Germanium Surface Preparation and GeO₂

Unlike SiO₂ on silicon, germanium oxide does not provide a stable, high quality passivation on germanium. Germanium oxide exists in a number of stoichiometrics (GeO_x) and crystal structures (amorphous, hexagonal, tetragonal) depending on oxidation conditions and processing history [20], [21]. Germanium oxide commonly exists as a mixture of these various forms and interconversion is possible during processing or exposure to ambient conditions [21], [22]. Hexagonal and amorphous germanium oxides are water soluble while the tetragonal form is insoluble. Because of the inherent instability of GeO_x, controlling the stoichiometry and phase is challenging. Attempts to control the properties of GeO₂ films have resulted in reasonable quality Ge-GeO₂ interfaces (reported $D_{it} \sim 5 \times 10^{11} \text{ cm}^{-2}$). However, these interfaces are still far from Si-SiO₂ interfaces in terms of quality and the proposed techniques are mainly applicable to thick dielectrics (greater than 10 nm) [20], [23]. Given the challenges associated with germanium oxide, a dielectric solution not relying on native oxide is preferable.

Due to the requirement of subnanometer EOT for future ultra-scaled MOSFETs [1], germanium oxide (or oxynitride) could only be used in a subnanometer interfacial layer between germanium and a high-k dielectric such as HfO₂. Control of ultra-thin germanium oxide layers is complicated by the lack of a stable surface termination. Unlike silicon, hydrogen-terminated germanium is not stable and is oxidized by exposure to ambient conditions [24]. Other surface terminations have been proposed (-S, -Cl, -Br, -I); however, they are also not completely stable against oxidation [24], [25] and have unknown consequences on the electrical properties of the germanium-dielectric interface.

1.5.2 High-k Dielectrics

To meet the EOT requirements for future MOSFETs, dielectrics with low permittivity (e.g. $SiO_2 \sim 3.9\varepsilon_0$) would need to be extremely thin (< 1 nm) and consequently would suffer from severe gate leakage due to direct tunneling [1]. For upcoming technology generations, the ITRS Roadmap calls for a maximum gate leakage of $\sim 1 \times 10^3$ A/cm² for an EOT of ~ 1 nm. Both of these specifications cannot be simultaneously achieved with SiO₂ or SiON, therefore high-k dielectrics are required [1]. A high dielectric constant allows low EOT with a physically thicker film, thus maintaining acceptable gate leakage. Of course, incorporating high-k dielectrics into CMOS devices also presents many challenges. These materials must be thermally stable in contact with the semiconductor and gate electrode, have large band offsets (> 1.0 eV) with respect to the semiconductor, and form a high

quality interface that does not degrade carrier mobility [74], [75]. To date, Hf-based materials have proven to be the most viable option [75]. HfSiO and HfSiON exhibit reduced remote phonon scattering and improved thermal stability (including reduced interfacial SiO₂ formation) compared to HfO₂ and are the leading candidates to replace SiO₂ [75]. However, an interfacial SiO₂ or SiON layer is still required to achieve a high quality semiconductor-dielectric interface. This layer may be purposefully introduced or can be an unintentional consequence of high-k deposition on silicon. SiON has several important advantages over SiO₂ including higher dielectric constant and improved thermal stability; however, the presence of nitrogen also introduces fixed charge that can degrade carrier mobility [75]. These issues have largely been resolved [75], and companies have recently announced the use of high-k dielectrics in production [76].

1.5.3 High-k Dielectrics Directly on Germanium

The development of deposited high-k gate dielectrics for silicon CMOS devices has fueled renewed interest in semiconductors like germanium that lack a high quality native insulator. Ideally, the high-k dielectric should be deposited directly on germanium without an interfacial layer; however, a thin GeO_x layer is often present due to incomplete oxide removal during cleaning, oxidation after cleaning, or oxidation during high-k dielectric deposition. The deposition of common high-k dielectrics such as HfO_2 and ZrO_2 on cleaned germanium oxide [26]. For atomic layer deposition (ALD) of HfO_2 on HF-last germanium, the resulting interfacial germanium oxide layer is approximately 0.3 nm [27]. To date, high-k dielectrics deposited directly on germanium have produced interfaces with very poor electrical characteristics including large hysteresis and frequency dispersion as well as degraded mobility [28], [29], [30]. In the case of HfO_2 , severe interfacial reactions [26] and germanium diffusion into the dielectric [28] have been observed during

subsequent thermal processing. For ZrO_2 , the interfacial layer disappears during annealing due to germanium diffusion into the dielectric layer [26]. Interfacial reactions, interdiffusion, and the presence of germanium oxide at the interface appear to be fundamental problems for HfO₂ and ZrO₂ deposited directly on cleaned germanium surfaces. Improved device characteristics have been reported for Ge MOSFETs with ZrSiO rather than ZrO₂ [30]; however, at present, incorporating an interfacial layer between the high-k dielectric and germanium is the most promising approach for achieving a high quality germaniumdielectric interface.

1.5.4 Interfacial Layers

The interlayer must address the shortcomings of $Ge-GeO_2$ and $Ge-(HfO_2, ZrO_2)$ inter-

faces. More specifically, the interlayer should:

- reduce interface state density,
- mitigate problems associated with interfacial GeO_x,
- prevent reactions and interdiffusion at the germanium-dielectric interface, and
- separate the inversion layer from the high-k dielectric to reduce remote coulomb and phonon scattering [31], [32].

Since it is difficult to passivate germanium surfaces against oxidation, a subnanometer layer of GeO_x is usually present on the surface prior to gate stack processing [25]. Furthermore, even if the initial surface is oxide-free, interfacial GeO_x can be formed during dielectric deposition or subsequent processing. To address the problems caused by interfacial GeO_x , the interlayer should:

- prevent GeO_x formation during processing,
- stabilize GeO_x, or
- chemically reduce GeO_x .

To date, germanium oxynitride (GeO_xN_y) [29], [33] and silicon [34] have been the most widely studied interlayers. In the case of GeON, nitrogen stabilizes the film making it

water insoluble and improving its electrical properties [35]. The use of a silicon capping layer effectively prevents GeO_x formation at the semiconductor-dielectric interface. Deposited nitrides (AIN and Hf₃N₄) and monolayer surface terminations (phosphorus and sulfur) have also been proposed as interlayers. Deposited nitride layers prevent oxidation during subsequent processing [36] but may also stabilize pre-existing oxide through nitrogen bonding. Monolayer surface treatments attempt to terminate all of the germanium surface bonds to prevent oxidation. The primary disadvantage of interlayers is the reduced EOT scalability of the dielectric stack. Aggressive EOT scaling will require reduction of the interlayer thickness and/or improved dielectric constant. GeON, silicon, metal nitride, and monolayer interlayers are discussed in detail in the following sections.

GeON

Thick germanium oxynitride was originally investigated as a standalone dielectric for germanium MOSFETs before EOT scaling demanded higher dielectric constant materials [35], [37]. More recently thin (~1 nm) GeON films have been used as interlayers between germanium and high-k dielectrics such as HfO₂ [33], [38]. Interfacial GeON improves both the physical and electrical properties of high-k dielectric stacks on germanium. HfO₂ deposited on HF-last germanium grows epitaxially and is polycrystalline while HfO₂ deposited on nitrided germanium yields an amorphous film [29]. Dielectric crystallization leads to increased gate leakage and is generally avoided. In addition, germanium oxynitride acts as a diffusion barrier and reduces germanium diffusion into the high-k dielectric [28]. MOS capacitors with GeON interfacial layers have shown significantly reduced leakage and more ideal C-V characteristics [29], [39]. Interface state densities as low as $5x10^{11}$ cm⁻² have been reported for Ge-GeON interfaces [40]. It has been shown, however, that dielectric stacks with GeON interlayers are not thermally stable and suffer from a large increase in gate leakage after annealing at temperatures above 500 °C [36]. This is particularly a problem for Ge n-FETs since the minimum activation temperature for n-type dopants is 500 °C. Furthermore, germanium oxynitride layers are not effective at preventing interfacial GeO_x formation during subsequent HfO₂ deposition [36].

Deposited Metal Nitrides

While nitridation of germanium results in a reasonable quality GeON native insulator, these films are still based on GeO_2 and suffer from poor thermal stability. As a consequence, deposited interlayers such as AlN [36], [41] and Hf₃N₄ [41] are being investigated. In principle, the composition of a deposited layer may be closely controlled and tailored to provide optimal interface characteristics. Metal nitride layers prevent oxidation during subsequent processing [36] and may also stabilize or chemically reduce pre-existing oxide. Like GeON, these interlayers suppress crystallization of the overlying high-k dielectric, reduce germanium in-diffusion, improve capacitor C-V characteristics, and reduce midgap D_{it} to reasonable levels (~1x10¹² cm⁻² eV⁻¹ [41]). However, metal nitrides also offer advantages over GeON, namely, higher dielectric constant, improved thermal stability, and reduced formation of interfacial germanium oxide during subsequent processing [36]. It has been shown that AlN interlayers are partially oxidized during subsequent high-k deposition but are still effective at reducing GeO_x formation at the interface [36]. The higher dielectric constant of metal nitrides, especially Hf_3N_4 ($\epsilon_r \sim 10$) [42], enables EOT scaling while enhanced thermal stability allows higher temperature n-type S/ D activation. AlN interlayers are stable to 600 °C while GeON starts to deteriorate at 500 °C [36]. However, as with Si₃N₄, AlN contains bulk traps and fixed charge that contribute to hysteresis and coulomb scattering [43], [77].

Silicon

In order to avoid the problems associated with germanium-dielectric interfaces, silicon capping layers have also been investigated [34], [44]. Very thin (~0.5 nm) silicon layers
may be epitaxially grown on germanium resulting in a high quality semiconductor-semiconductor interface. The large valence band offset (+0.51 eV [16]) between germanium and silicon forces the hole inversion layer into the underlying germanium. The conduction band edges of germanium and silicon are closely aligned (+0.05 eV [16]) so electron inversion is not confined to the underlying germanium. Silicon interlayers have several significant disadvantages. First, if inversion *does not* occur in the silicon layer it effectively acts as a dielectric and increases the effective oxide thickness. Second, if inversion *does* occur in the silicon layer, the structure does not take full advantage of germanium carrier mobility, and in the case of electrons, may suffer from intervalley scattering between the Si Δ and Ge L valleys [45]. Third, device performance will be very sensitive to small variations in silicon thickness so the manufacturability of this structure is questionable. Nevertheless, strained Si/Ge heterostructures show impressive electron and hole mobility enhancement [45] and are one of the primary options being considered for future CMOS devices.

Monolayer Surface Terminations

In an ideal germanium-dielectric interface, all germanium surface atoms are unperturbed and all bonds are properly terminated. In an attempt to achieve this ideal surface, monolayer surface terminations have been proposed for surface passivation prior to high-k dielectric deposition. It has been demonstrated that a few monolayers of sulfur terminate clean germanium surfaces and that these sulfur layers are stable up to ~200 °C [46]. Sulfur incorporation into germanium oxide has recently been reported [47]. Phosphorus-based surface treatments have also been investigated [48]. It was shown that phosphine plasma treatment at 400 °C provided similar benefits to AIN interfacial layers: suppression of germanium oxide formation, germanium in-diffusion, and high-k crystallization as well as improved thermal stability and carrier mobility. Nonetheless, monolayer termination approaches are highly speculative and it is unclear what effect they have on the electronic properties of germanium-dielectric interfaces.

1.6 Historical Germanium MOSFET Performance

While MOS C-V characteristics are important for evaluating germanium-dielectric interfaces, device performance is the ultimate metric for assessing Ge MOSFETs. Efforts to develop germanium MOSFETs with high-k gate dielectrics started in 2002. Figure 1.7 shows electron and hole mobility at constant vertical effective field for bulk germanium MOSFETs reported since that time. Silicon universal electron and hole mobility are shown for reference. Results that have been difficult to reproduce and are not consistent with other published data are denoted by black circles. These results are not included in the discussion that follows. Table 1.6 provides a more detailed description of the data points in Figure 1.7, including the gate stack materials and dielectric deposition method.



Figure 1.7: Carrier mobility at 0.3 MV/cm for Ge MOSFETs reported since 2002. Silicon universal electron and hole mobility are shown for reference. Results that have been difficult to reproduce and are not consistent with other published data are denoted by black circles. Results from silicon interlayers are denoted by the cross-hatched circle.

Year	Gate Stack Description (Dielectric Deposition Method)	Carrier Mobility at 0.3 MV/cm		Reference
		NMOS	PMOS	
2002	Pt/ZrO ₂ /Ge (PVD followed by UV ozone oxidation)	-	250	[49]
2004	Al/SiO ₂ /GeON/Ge (thermal oxidation followed by nitridation)	100	160	[40], [50]
2004	TaN/HfO ₂ /(AlN or PH ₃ treatment)/Ge (MOCVD and ALD)	150	80	[48]
2005	(IrO ₂ or IrO ₂ /Hf)/LaAlO ₃ /Ge (PVD)	350	175	[51]

Table 1.6: Gate stack materials and deposition method for mobility resultssummarized in Figure 1.7.

Ycar	Gate Stack Description (Dielectric Deposition Method)	Carrier Mobility at 0.3 MV/cm		Reference
		NMOS	PMOS	
2005	Mo/ZrO ₂ /Ge (PVD)	-	75	[30]
2005	Mo/ZrSiO/Ge (PVD)	-	125	[30]
2005	TaN/HfO ₂ /GeON/Ge (thermal nitridation followed by MOCVD)	-	140	[44]
2005	TaN/HfO ₂ /Si/Ge (MOCVD)	180	210	[44]
2006	TaN/HfO ₂ /Si/Ge (MOCVD)	175	-	[52]
2006	TiN/HfO ₂ /Si/Ge (ALD)	-	275	[34]

 Table 1.6: Gate stack materials and deposition method for mobility results summarized in Figure 1.7.

As shown in Figure 1.7, enhanced hole mobility in Ge MOSFETs has been demonstrated by a number of groups. The best results have been achieved with thin (~0.5 nm) silicon interlayers with 3x enhancement recently being demonstrated [34]. Silicon capping avoids the germanium-dielectric interface problem and the large valence band offset between germanium and silicon effectively results in a buried-channel device. Manufacturability and EOT scalability are major drawbacks of this approach. Surface-channel p-FETs with highk dielectrics have shown a much less impressive average enhancement of 0.4x. In contrast to the success of Ge p-FETs, n-FETs have shown mobilities far below silicon universal electron mobility, independent of gate stack composition and deposition method. Even n-FETs with silicon interlayers have shown disappointing results. The inversion layer in these devices is not buried due to the small conduction band offset and intervalley scattering may be occurring at the heterointerface [45]. While hole mobility enhancement has been seen for a wide variety of gate dielectric materials and processes, electron mobility has been universally poor. Successful demonstration of Ge n-MOSFETs will require better understanding of electron transport at germanium-dielectric interfaces and the development of new methods to passivate the interface.

1.7 Goals of Thesis

As demonstrated by recent published data, much progress has been made on Ge MOS-FETs, particularly p-FETs. Nonetheless, passivation of the germanium-dielectric interface remains a serious challenge and the low mobility observed in Ge n-FETs is poorly understood. In this work, the device characteristics of Ge MOSFETs with nitride interlayers are evaluated. ALD metal nitrides (AlN and Hf_3N_4) are the primary focus; however, germanium oxynitride is also investigated. The specific goals of this thesis are to:

- develop, optimize, and characterize AlN interlayer gate stacks,
- assess the device performance of Ge MOSFETs with nitride interlayers, and
- understand the impact of the germanium-dielectric interface on carrier transport.

1.8 Organization of Thesis

The remainder of this thesis is organized into four chapters. Chapter 2 reviews atomic layer deposition and describes the development of ALD gate stacks based on AlN interlayers. Optimization and characterization of WN/Al₂O₃/AlN gate stacks are discussed. Chapter 3 presents the device characteristics of Ge n- and p-MOSFETs with AlN, Hf₃N₄, and GeON interlayers. Carrier mobility is extracted for each interlayer and summarized at the end of the chapter. Chapter 4 discusses the impact of interface states on carrier transport in WN/Al₂O₃/AlN Ge MOSFETs. Low temperature device characterization is used to examine the temperature dependence of mobility and also extract interface state density near the band edge. Buried-channel MOSFETs are used to reduce the interaction of carriers with interface states. Chapter 4 also presents evidence of phosphorus passivation of the germanium-dielectric interface. And last, Chapter 5 summarizes the conclusions and contributions of this thesis, and discusses opportunities for future work.

Chapter 2

Atomic Layer Deposition of High-k Dielectric Stacks on Germanium

2.1 Introduction

Atomic layer deposition (ALD) allows a wide variety of materials to be deposited with layer-by-layer precision. This exquisite control over thickness and composition makes ALD particularly well-suited for the deposition of thin interfacial layers (0.5-2.0 nm) and high-k dielectrics for MOSFET applications. A model dielectric stack process consisting of an RCA-based clean followed by an aluminum nitride (AlN) interfacial layer and aluminum oxide (Al₂O₃) capping layer was selected for this work. AlN is one of only a few dielectric layers (GeON [50], AlN [48], Hf₃N₄ [41], LaAlO₃ [51]) that have produced functional Ge n-MOSFETs. In addition to reducing Ge diffusion into the high-k dielectric and suppressing interfacial GeO_x formation [36]), AlN also has a medium dielectric constant ($\epsilon_r \sim 7-10$) which makes it suitable for use as an interlayer in scaled dielectric stacks. Al₂O₃ was chosen as a capping layer for practical reasons. It is easily deposited using the same precursor and substrate temperature as AlN, and also exhibits low leakage at the thicknesses used in this work (3.5- 6 nm). To achieve low EOT, the Al_2O_3 capping layer could be replaced by a high dielectric constant material such as HfO2. This chapter describes the optimization and characterization of Al₂O₃/AlN dielectric stacks on germanium with particular focus on AlN thickness and post-metal anneal (PMA) optimization, and interface state density (D_{it}) characterization. The purpose of this study was to understand the basic properties of Al₂O₃/AlN dielectric stacks, including thermal stability, while developing a gate stack suitable for investigating carrier transport in Ge MOSFETs. This chapter begins with a brief review of atomic layer deposition and then presents the results of the Al_2O_3/AlN study. The conclusions drawn in this chapter are specific to the model dielectric stack process described above. Surface preparation and interfacial layer formation are tightly coupled, dependent processes. Recent evidence that will be discussed at the end of this chapter suggests that alternative surface preparations may be superior to the preparation used in this work, and that the optimal dielectric stack for these surfaces may be different than for RCA-cleaned surfaces.

2.2 Brief Review of Atomic Layer Deposition

Atomic layer deposition utilizes alternating pulses of precursors that undergo self-limiting reactions with the substrate and each other. Unlike chemical vapor deposition (CVD), the precursors are introduced into the chamber sequentially rather than simultaneously [53]. Atomic layer deposition occurs in four distinct steps which together are called an *ALD cycle* [54]:

- 1. Reactant A is pulsed into the chamber and chemisorbs on the surface forming a monolayer (ideal case).
- 2. Unreacted molecules of reactant A and by-products are removed from the chamber.
- 3. Reactant B is pulsed into the chamber and undergoes a self-limiting reaction with the chemisorbed molecules of reactant A.
- 4. Unreacted molecules of reactant B and by-products are removed from the chamber.

Figure 2.1 shows a schematic representation of the ALD reaction between trimethylaluminum (TMA) and water to form Al_2O_3 [55]. ALD processes typically begin with a hydroxyl (-OH) terminated surface as shown Figure 2.1a. Figures 2.1b and 2.1c show TMA undergoing a self-limiting reaction with surface hydroxyl groups followed by the removal of excess TMA and reaction by-products (CH₄). Figures 2.1d and 2.1e show water reacting with the remaining methyl groups on the chemisorbed TMA molecules followed by the removal of excess water and reaction by-products (CH₄). Note that at the end of the cycle, the surface is again hydroxyl terminated. Substrate temperature affects substrate-precursor interactions as well as precursor stability and reactivity [53]. The substrate temperature range over which surface-limited ALD growth occurs is referred to as the *ALD window*. An ideal ALD process should be independent of precursor dose and exhibit a linear growth rate. Further information on ALD may be found in references 53 and 54.





Figure 2.1: Schematic representation of reactions that comprise an ALD cycle. The reaction between trimethylaluminum (TMA) and water to form Al_2O_3 is used as an example (adapted from [55]).

2.3 ALD Gate Stack Films

The ALD gate stacks in this work consisted primarily of Al_2O_3/AlN dielectric stacks capped with a tungsten nitride (WN) electrode. The majority of the these films were deposited using a Cambridge NanoTech Savannah 200 ALD reactor installed in the Microsystems Technology Laboratory at MIT. Other high-k dielectrics (Hf₃N₄, GdScO₃, LaAlO₃) were deposited by K. Kim and R. Gordon at Harvard University.

2.3.1 Atomic Layer Deposition of AlN and Al₂O₃

AlN and Al_2O_3 were deposited using tris(dimethylamido)aluminum (TDMAA) and either ammonia or water as the co-reactant, respectively [41], [56]. The TDMAA bubbler temperature was 115 °C and a nitrogen carrier stream delivered the reactants from the precursor manifold to the chamber. The substrate temperature was 200 °C. Detailed process parameters are contained in Appendix A. SIMS analysis of a thick (50 nm) AlN film was performed to quantify the carbon and oxygen concentration in the film. The carbon and oxygen concentrations were 3.8×10^{20} cm⁻³ and 4.5×10^{20} cm⁻³, respectively, or 0.8% and 0.9% assuming an atomic density of 5×10^{22} cm⁻³. The growth rate of AlN and Al_2O_3 was 0.9-1.0 Å/cycle, as measured by spectroscopic ellipsometry.

2.3.2 Atomic Layer Deposition of WN

WN was deposited using bis(tertbutylimido)-bis(dimethylamido)tungsten with ammonia as the co-reactant [57]. The tungsten bubbler temperature was 90 °C and a nitrogen carrier stream delivered the reactants from the precursor manifold to the chamber. The substrate temperature was 340-375 °C. Detailed process parameters are contained in Appendix A. WN resistivity was measured using the four-point probe technique and was consistent with values reported in the literature $(1.5 \times 10^{-3} - 4.0 \times 10^{-3} \Omega$ -cm [57]). The WN growth rate was approximately 0.5 Å/cycle.

2.4 Review of MOS Capacitor Electrical Characterization

Impedance analysis is the most common method for evaluating the electrical properties of semiconductor-dielectric interfaces. The impedance of an MOS capacitor is typically measured as a function of gate bias and frequency. Measurement frequencies range from DC (quasistatic) to 1 MHz. Using an appropriate equivalent circuit model, relevant parameters of the device under test (e.g. capacitance, conductance, etc.) can be extracted. Imperfect semiconductor-dielectric interfaces require a more complex equivalent circuit to capture the effects of interface states and other non-idealities. Figure 2.2 shows the small-signal equivalent circuit model for an MOS capacitor with interface states [59]. The interface state capacitance is represented by C_{il} while R_{il} models losses due to the interface states, τ_{il} , is simply $C_{il}R_{il}$. C_{ox} and C_s represent the oxide and semiconductor capacitance, respectively.



Figure 2.2: Small-signal equivalent circuit model for MOS capacitor with interface states (from [59]).



Figure 2.3: Common non-idealities observed in MOS capacitors. (A) Midgap D_{it} distortion, DC stretch-out, V_{FB} shift, and frequency dispersion. (B) Hysteresis and trap-assisted minority carrier generation.

At low frequency ($f < (2\pi\tau_{ii})^{-1}$), distortion of the C-V characteristics is typically observed due to interface states that are able to respond at that frequency. Figure 2.3a shows 50 kHz and 1 MHz C-V characteristics for an MOS capacitor with midgap interface states. An ideal simulated 1 MHz curve without interface states is included for reference. This curve was generated using MISFIT, a C-V simulation program for Si and Ge MIS structures [72]. At 50 kHz, C-V distortion is clearly visible however at 1 MHz the interface states are unable to respond and do not contribute to the measured capacitance. In addition to C-V distortion, interface states also produce DC "stretch-out". Interface states must be filled as the Fermi level is swept through the bandgap thus requiring a larger voltage sweep to go from depletion to accumulation. This also produces a shift in flatband voltage (V_{FB}). The capacitance corresponding to the flatband condition (C_{FB}) can be determined from simulation, and thus V_{FB} is simply the voltage at which the measured capacitance equals C_{FB} . Figure 2.3a also shows frequency dispersion (frequency dependence of capacitance) in accumulation. This can be caused by problems intrinsic to the dielectric and semiconductor-dielectric interface as well as external parasitics (e.g. series resistance due to the gate and/or substrate). Another common problem in MOS capacitors is C-V hysteresis which is shown in Figure 2.3b. This plot shows both forward (depletion to accumulation) and reverse (accumulation to depletion) voltage sweeps. During measurement, charge can be injected into bulk traps in the dielectric or stored in slow interface states which produces a flatband voltage shift. This shift is clearly evident during the reverse voltage sweep. Figure 2.3b also shows the effect of minority carrier generation. At typical measurement frequencies (10 kHz to 1 MHz), inversion capacitance increase is not observed in MOS capacitors due to the low generation/recombination rate of minority carriers that prevents them from responding at these frequencies. However, for narrow bandgap semiconductors or in the case of significant trap-assisted generation/recombination, some level of capacitance increase in inversion can be observed in MOS capacitor C-V characteristics as shown in Figure 2.3b. For Si-SiO₂ interfaces, minority carrier generation has been observed due to bulk trap states in the semiconductor [73].

Since many of the non-idealities described above are very sensitive to sweep conditions, it is important to perform the measurements in a consistent manner. In this work, all C-V characteristics were measured using an Agilent IBASIC program that holds the gate voltage at the first sweep point for 1 s before starting the forward and reverse sweeps. Also, the maximum bandwidth setting of the Agilent 4294A Impedance Analyzer was used to minimize measurement time (1-10 ms/point) thereby capturing transient effects. By holding the forward and reverse sweep start voltage for 1 s and minimizing total measurement time, the measured hysteresis is "worst case" because the interface has little time to relax. In addition, the use of an automated program ensures that measurements are repeatable and allows fair comparison across different samples. The Agilent IBASIC program is listed in Appendix B.

2.5 ALD Gate Stack Capacitor Fabrication and Characterization

Germanium capacitors were fabricated to optimize and characterize Al₂O₃/AlN dielectric stacks. Initial experiments used an *ex-situ* sputtered aluminum gate electrode while later experiments used an *in-situ* ALD WN gate electrode. Al electrode capacitors were used to verify the film growth mechanism and also to estimate the dielectric constant of ALD AlN and Al₂O₃. Gate stack optimization and interface state characterization were performed on WN electrode capacitors. From a process integration perspective, aluminum is not a practical gate material because of its low melting point and poor chemical resistance, so WN was used in MOSFET fabrication. In addition, Al-gated MOS capacitors exhibited poor thermal stability characterized by a large increase in hysteresis during 400- 500 °C anneals. WN-gated capacitors were stable up to 500 °C and in general, the electrical characteristics improved with increasing PMA temperature from 350 to 500 °C.

2.5.1 Al/Al₂O₃/AlN/Ge capacitors

Experimental Description

Capacitors were fabricated on (100) Ga-doped p-type substrates with resistivity of 0.12-0.17 Ω -cm and (100) Sb-doped n-type substrates with resistivity of 0.13-0.16 Ω -cm. The wafers were cleaned using the modified RCA clean described in Appendix A. The dielectric stacks shown in Table 2.1 were deposited on both n- and p-Ge followed by *ex-situ* Al sputtering (200 nm) on the frontside of the wafer and Ti/Au sputtering (50 nm/ 200 nm) on the backside of the wafer. The aluminum gate electrodes were patterned using photolithography and wet etching. In this document, Al₂O₃/AlN dielectric stacks are denoted by the number of Al₂O₃ cycles followed by the number of AlN cycles (e.g. 50/20).

ALD Run No.	Al ₂ O ₃ cycles/AlN cycles
41	50/20
42	50/30
43	100/30
44	50/10

Table 2.1: Dielectric stack splits for Al/Al₂O₃/AlN/Ge capacitors. Stacks are defined by number of Al₂O₃ cycles and number of AlN cycles (Al₂O₃ cycles/ AlN cycles)

Electrical Results

The first step in characterizing any ALD process is to verify that surface-limited ALD growth is actually occurring. In the case of a dielectric, this can be accomplished by measuring capacitance as a function of the number of ALD cycles. Table 2.2 summarizes the capacitance equivalent thickness (CET) for n- and p-Ge capacitors after 350 °C 30 min. N₂ post-metal annealing. CET is simply the equivalent SiO₂ thickness corresponding to the peak measured capacitance. Figure 2.4 shows CET vs. total number of ALD cycles for the p-Ge capacitors in Table 2.2. The linear relationship between CET and total number of cycles verifies ALD deposition and also suggests that the deposition rate and dielectric constant of AIN and Al₂O₃ result in similar effective thickness per cycle. Since the deposition rates are similar (0.9-1.0 Å/cycle), the relative dielectric constants are also similar ($\epsilon_{\rm r} \sim$ 6-7 for both films). The y-intercept in Figure 2.4 at zero cycles indicates the presence of an ultra-thin (0.38 nm CET) interfacial oxide. The composition of this layer is a matter of speculation but likely includes Ge, Al, O, and N.

These capacitors exhibited poor thermal stability, as evidenced by a large increase in hysteresis, at post-metal anneal temperatures above 350 °C. Furthermore, aluminum is not suitable for use as a MOSFET gate electrode and *ex-situ* gate electrode deposition exposes the gate dielectric to ambient conditions after deposition. For these reasons, most optimi-

zation and characterization was performed on WN electrode capacitors, and is discussed in the following sections. C-V characteristics of the Al-gated capacitors are summarized in Appendix C.

Dielectric Stack	n-Ge CET (nm)	p-Ge CET (nm)
10/50	3.97	3.82
20/50	4.45	4.40
30/50	4.65	4.91
30/100	7.63	7.81

Table 2.2: Capacitance equivalent thickness for Al/Al₂O₃/AlN/Ge capacitors.



Figure 2.4: CET of $Al/Al_2O_3/AlN/p$ -Ge capacitors vs. total number of ALD cycles. Y-intercept of 0.38 nm indicates the presence of an ultra-thin interfacial oxide.

2.5.2 WN/(Al₂O₃,AlN)/Ge Capacitors

 Al_2O_3/AlN dielectric stacks were primarily evaluated using *in-situ* ALD WN gate electrodes. *In-situ* gate electrode deposition prevents the dielectric from being exposed to ambient conditions and the thermal stability of WN allows higher temperature post-metal annealing. In this section, the C-V characteristics of Al_2O_3/AlN stacks are compared to pure Al_2O_3 and AlN films to clearly demonstrate the benefits of thin (~2 nm) AlN interlayers. The impact of post-metal annealing on capacitors with 1.5 nm and 2.5 nm AlN interlayers is then discussed. And finally, the results of interface state characterization are presented.

Experimental Description

Capacitors were fabricated on (100) Ga-doped p-type substrates with resistivity of 0.12-0.17 Ω -cm and (100) Sb-doped n-type substrates with resistivity of 0.13-0.16 Ω -cm. The wafers were cleaned using the modified RCA clean described in Appendix A. The WN/ (Al₂O₃,AlN) gate stacks shown in Table 2.3 were deposited on both n- and p-Ge followed by *ex-situ* Al sputtering (250 nm) on the frontside of the wafer and Ti/Al sputtering (20 nm/ 1 µm) on the backside of the wafer. The aluminum metallization was patterned using photolithography and wet etching. Using the same resist mask, the WN gate electrode was etched using CF₄ RIE. The capacitors were then annealed as shown in Table 2.4. The "asdeposited" samples did not receive a post-metal anneal but did receive the thermal budget associated with WN deposition (approximately 350 °C for 6 hrs. in N₂).

ALD Run No.	Al ₂ O ₃ cycles/AlN cycles
104	45/15
105	35/25
111	60/00
113	00/60

Table 2.3: Dielectric stack splits for WN/(Al₂O₃, AlN)/Ge capacitors. Stacks are defined by number of Al₂O₃ cycles and number of AlN cycles (Al₂O₃ cycles/ AlN cycles).

	····	
Temperature (°C)	Time (min.)	Ambient
as-deposited	-	-
350	30	FG
400	1	N ₂
450	30	FG
450	1	N ₂
500	30	FG
500	1	N ₂
550	30	FG
550	1	N ₂

Table 2.4: Post-metal anneal splits for WN/Al₂O₃/AlN/Ge capacitors. FG= 8:1 N₂:H₂.

Impact of AIN Interlayer Thickness on C-V Characteristics

Figure 2.5 shows C-V characteristics at 50 kHz, 100 kHz, and 1MHz for as-deposited gate stacks. The left column shows p-Ge while the right column shows n-Ge. The CET for these capacitors is approximately 3 nm with p-Ge having slightly higher peak capacitance than n-Ge. This CET corresponds to an average relative dielectric constant of 7-8. This value is slightly higher than for Al-gated capacitors, probably due to the enhanced stability of WN/Al₂O₃ interfaces compared to Al/Al₂O₃ interfaces. Figures 2.5a and 2.5d show results from gate dielectric stacks consisting of pure Al_2O_3 and AlN, respectively. Al_2O_3 directly on germanium produces significant hysteresis for both p- and n-Ge (250 mV and 500 mV). DC stretch-out and distortion due to interface states are also evident. AlN directly on germanium produces poor C-V characteristics for both p- and n-Ge and large hysteresis for n-Ge. The WN/AlN/p-Ge capacitor shows inversion behavior due to trapassisted minority carrier generation. The WN/AlN/n-Ge capacitor shows severe distortion, particularly near accumulation. In contrast, the capacitors in Figures 2.5b and 2.5c with WN/Al₂O₃/AlN gate stacks show significantly improved C-V characteristics with reduced stretch-out, distortion, and hysteresis. In addition to better-behaved C-V characteristics, WN/Al₂O₃/AlN stacks also exhibit low gate leakage ($< 10^{-6}$ A/cm²) as shown in Figure 2.6. These results clearly demonstrate the benefit of using thin (1.5-2.5 nm) AlN interlayers.



Figure 2.5: Forward and reverse sweep C-V characteristics of as-deposited WN/(Al₂O₃, AlN)/Ge capacitors at 50 kHz, 100 kHz, and 1 MHz. (A) 60/00 Al₂O₃/AlN (B) 45/15 Al₂O₃/AlN (C) 35/25 Al₂O₃/AlN (D) 00/60 Al₂O₃/AlN.



Figure 2.6: Gate leakage for WN/ 35 cycle $Al_2O_3/25$ cycle AlN gate stacks on n- and p-Ge.

Impact of Post-Metal Annealing on C-V Characteristics

Post-metal annealing is a critical step for high-k dielectrics, particularly ALD films deposited at low temperature (200 °C), that is used to improve the electrical properties of asdeposited gate stacks. However, degradation of germanium-dielectric interfaces has also been observed during subsequent thermal processing. The purpose of this experiment was to determine the optimal post-metal anneal and investigate the thermal stability of these gate stacks at n-type S/D activation temperatures (500- 550 °C).

Due to the large number of unique samples (approximately forty), several simple metrics were developed to facilitate sample comparison. Hysteresis is a common metric for C-V characteristics; however, some capacitors in this work exhibit variable hysteresis depending on the magnitude of the capacitance, so a single value does not sufficiently describe the hysteretic behavior. To address this issue, hysteresis values were extracted at 20% (H-20) and 80% (H-80) of peak normalized capacitance as demonstrated in Figure 2.7a. The difference between these two hysteresis values can also provide useful information. For example, slow detrapping of charge trapped during accumulation will result in larger hysteresis at 80% C/C_{max} than at 20% C/C_{max}. This difference provides insight into the severity and time constant of interface trapping.

In addition to hysteresis, C-V distortion due to midgap interface states is also a useful metric for comparing C-V characteristics. While hysteresis provides information on charge injection and trapping, C-V distortion provides complementary information on midgap D_{it} . In this work, the area difference between the 50 kHz and 1 MHz curves (shown in Figure 2.7b as ΔQ) serves as the quantitative metric for midgap D_{it} . 50 kHz and 1 MHz curves sometimes show frequency dispersion in accumulation so each curve is normalized separately to ensure that ΔQ is due to midgap D_{it} and not to frequency dispersion in accumulation. Trap-assisted minority carrier generation can also produce a deviation in low and high frequency C-V characteristics in the depletion/inversion regime. While this effect is not due to midgap interface states [73], it is still undesirable and will get included in the ΔQ metric. Thus for samples with high ΔQ , the C-V characteristics must be examined to determine whether it results from minority carrier generation or mid-gap D_{it} .



Figure 2.7: Metrics used to compare C-V characteristics of MOS capacitors. (A) Hysteresis extracted at 20% (H-20) and 80% (H-80) of C/C_{max} and (B) ΔQ extracted from separately normalized 50 kHz and 1 MHz forward-sweep C-V characteristics.

Since dielectric stacks with AlN interlayers exhibit the most promising as-deposited electrical characteristics, these samples were used to study the impact of post-metal annealing. The WN/45 cycle $Al_2O_3/15$ cycle AlN and WN/35 cycle $Al_2O_3/25$ cycle AlN gate stacks will simply be denoted by their AlN thickness (1.5 nm and 2.5 nm, respectively). Both stacks have similar total thickness and peak capacitance, so AlN thickness is the differentiating parameter for these stacks. These capacitors were subjected to the post-metal annealing conditions listed in Table 2.4.

Figure 2.8 shows ΔQ vs. anneal temperature for (A) 60 s N₂ and (B) 30 min. FG annealing. As will be the case for all plots in this section, n-Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. Additionally, 1.5 nm and 2.5 nm AIN interlayer thicknesses are indicated by circles and squares, respectively. The actual C-V characteristics used to extract ΔQ and hysteresis are shown in Appendix D. For most samples, ΔQ decreases with increasing anneal temperature. For 60 s N₂ annealing, reduction in ΔQ is seen up to 550 °C. For 30 min. FG annealing, ΔQ decreases up to 450 °C but no additional reduction is observed at 500 and 550 °C. The ultimate reduction in ΔQ is similar for 60 s 550 °C N_2 and 30 min. 450 °C FG annealing suggesting that thermal budget, rather than anneal ambient, is responsible for the decrease in midgap Dit. Most samples followed the general trends outlined above, however, there was one notable exception. The 2.5 nm AlN n-Ge capacitor shows a large increase in ΔQ when annealed in nitrogen. Examination of the C-V characteristics for this sample (see Appendix D) shows trap-assisted minority carrier generation. The precise physical mechanism responsible for enhanced minority carrier generation in these samples is not known; however, similar behavior in Si MOS capacitors has been attributed to bulk semiconductor traps [73].



Figure 2.8: ΔQ vs. anneal temperature for n- and p-Ge capacitors with 1.5 and 2.5 nm AlN interlayers. (A) 60 s N₂ annealing. (B) 30 min. FG annealing. n-Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. 1.5 nm and 2.5 nm AlN interlayer thicknesses are indicated by circles and squares, respectively. The N₂ anneal data point at 350 °C is from the as-deposited sample.

Figure 2.9 shows hysteresis at 20% C/C_{max} vs. anneal temperature for (A) 60 s N₂ and (B) 30 min. FG annealing. At temperatures below 500 °C, 1.5 nm and 2.5 nm AlN interlayers show hysteresis values of 150- 225 mV and 100- 150 mV, respectively. For N₂ annealing, hysteresis reaches a minimum at 500 °C before increasing slightly for p-Ge and rapidly for n-Ge at 550 °C. For FG annealing, n-Ge hysteresis reaches a minimum at 400- 450 °C before increasing rapidly at 500 and 550 °C while p-Ge reaches a minimum at 500 °C and increases rapidly at 550 °C. Again, differences between N₂ and FG annealing appear to be due to thermal budget (i.e. anneal time and temperature) rather than anneal ambient. In general, n-Ge capacitors exhibit less thermal stability than p-Ge capacitors. In N₂, both are stable up to 500 °C, however hysteresis increases more rapidly in n-Ge at 550 °C, and in FG, n-Ge is stable to 450 °C while p-Ge is stable to 500 °C. It is interesting to note that n- and p-Ge have similar H-20 values before degradation at high temperature. Capacitors with 2.5 nm interlayers also tend to have lower hysteresis than capacitors with 1.5 nm interlayers. Since the interfacial region is quite complex, this observation is difficult to rationalize with simple bulk/interface trap theory.



Figure 2.9: Hysteresis at 20% C/C_{max} vs. anneal temperature for n- and p-Ge capacitors with 1.5 and 2.5 nm AlN interlayers. (A) 60 s N₂ annealing. (B) 30 min. FG annealing. n-Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. 1.5 nm and 2.5 nm AlN interlayer thicknesses are indicated by circles and squares, respectively. The N₂ anneal data point at 350 °C is from the as-deposited sample.

Figure 2.10 shows hysteresis at 80% C/C_{max} vs. anneal temperature for (A) 60 s N₂ and (B) 30 min. FG annealing. The values generally range from 300- 450 mV for n-Ge and 100- 200 mV for p-Ge. In contrast to the similar hysteresis values at 20% C/C_{max} for n- and p-Ge, n-Ge shows much larger hysteresis at 80% C/C_{max} due to greater trapping of carriers in slow states during accumulation. This trapping results in a large flatband shift for the reverse C-V characteristic. However, some carriers detrap during the reverse sweep time (~0.5 s for 50 bias points and measurement time of 10 ms/point). This effectively results in a time-dependent flatband shift that causes C-V hysteresis to decrease during the reverse sweep. By 20% C/C_{max}, most electrons have detrapped, therefore, 20% hysteresis is similar for n- and p-Ge. Hysteresis at 20% C/C_{max} is probably determined by bulk trapping in the AlN interlayer and hence less dependent on substrate type, while hysteresis at

80% C/C_{max} depends on carrier trapping near the conduction and valence band edges of nand p-Ge, respectively. For n-Ge, hysteresis at 80% C/C_{max} shows clear minima at 500 °C and 450 °C for N₂ and FG annealing, respectively. In addition, n-Ge capacitors with 2.5 nm interlayer show significantly less hysteresis than capacitors with 1.5 nm interlayer. For p-Ge, hysteresis at 80% C/C_{max} is relatively independent of anneal temperature but does increase at 550 °C.

The difference between hysteresis at 80% and 20% C/C_{max} is shown in Figure 2.11. p-Ge samples exhibit constant hysteresis, so as expected, the difference is approximately zero. The difference for n-Ge decreases at high anneal temperature partly due to reduced hysteresis at 80% C/C_{max} but also because of increased hysteresis at 20% C/C_{max}. Thus while the difference in hysteresis decreases at 550 °C, the average hysteresis increases substantially.



Figure 2.10: Hysteresis at 80% C/C_{max} vs. anneal temperature for n- and p-Ge capacitors with 1.5 and 2.5 nm AlN interlayers. (A) 60 s N₂ annealing. (B) 30 min. FG annealing. n-Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. 1.5 nm and 2.5 nm AlN interlayer thicknesses are indicated by circles and squares, respectively. The N₂ anneal data point at 350 °C is from the as-deposited sample.



Figure 2.11: Difference in hysteresis at 80% and 20% C/C_{max} vs. anneal temperature for n- and p-Ge capacitors with 1.5 and 2.5 nm AlN interlayers. (A) 60 s N₂ annealing. (B) 30 min. FG annealing. n-Ge and p-Ge capacitors are denoted by closed and open symbols, respectively. 1.5 nm and 2.5 nm AlN interlayer thicknesses are indicated by circles and squares, respectively. The N₂ anneal data point at 350 °C is from the as-deposited sample.

Summary of AIN Thickness and PMA Optimization

N- and p-Ge capacitors were fabricated with Al₂O₃, Al₂O₃/AlN, and AlN gate dielectrics. Al₂O₃/AlN dielectric stacks showed superior characteristics compared to pure Al₂O₃ and AlN. The impact of post-metal annealing on WN/Al₂O₃/AlN/Ge capacitors with 1.5 nm and 2.5 nm AlN interlayers was investigated for 60 s N₂ and 30 min. FG annealing at 350-550 °C. The C-V characteristics were compared using several simple metrics: area difference between 50 khz and 1 Mhz curves (Δ Q), hysteresis at 20% C/C_{max} (H-20), and hysteresis at 80% C/C_{max} (H-80). Figure 2.12 summarizes important trends from the postmetal annealing study on n- and p-Ge capacitors with 1.5 and 2.5 nm interlayers. Each column shows the results for a particular metric while row 1 and row 2 summarize the results for N₂ and FG annealing, respectively. The third row summarizes the impact of AlN thickness on the given metric.



Figure 2.12: Summary of important trends from PMA study on n- and p-Ge capacitors with 1.5 and 2.5 nm AlN interlayers. Metrics are shown in each column while row 1 and 2 summarize the results for N_2 and FG annealing, respectively. The impact of AlN interlayer (IL) thickness on the given metric is shown in the third row.

By examining the temperature dependence of these three metrics, it is clear that n- and p-Ge capacitors are both stable to 500 °C for short anneal times (60 s). At longer anneal times (30 min.), or higher temperature (550 °C), n-Ge C-V characteristics start to degrade more severely than p-Ge. For all of the data, the observed trends correlate with thermal budget rather than anneal ambient. It does not appear that hydrogen passivation plays the same role at germanium-AIN interfaces that it does for Si-SiO₂ interfaces, possibly due to lower Ge-H bond strength and the position of hydrogen energy levels in germanium [58]. Hysteresis at 20% and 80% C/C_{max} also shows an important trend. While hysteresis at 20% C/C_{max} is similar for n- and p-Ge, hysteresis at 80% C/C_{max} is much higher for n-Ge indicating greater carrier trapping. Less trapping occurs in capacitors with 2.5 nm interlay-

ers and in general, these capacitors outperform those with 1.5 nm interlayers. In this gate stack, the thicker AlN interlayer is probably more effective at reducing or stabilizing interfacial GeO_x. For the process space explored in this work, 2.5 nm AlN interlayers annealed at 500 °C in N₂ or 450 °C in FG produced the best C-V characteristics (see Figure 2.13). In the next section, interface state characterization is performed on 2.5 nm AlN interlayer capacitors annealed at 450 °C in FG.



Figure 2.13: C-V characteristics for WN/3.5nm $Al_2O_3/2.5$ nm AlN/Ge capacitors annealed at 450 °C for 30 min. in FG.

2.5.3 Interface State Characterization of WN/Al₂O₃/AlN/Ge Capacitors

Common capacitor-based techniques for extracting D_{it} include the conductance, quasistatic, and Gray-Brown methods [59]. MOSFET-based techniques include charge pumping [59] and V_t temperature dependence [79]. Challenges in measuring interface state density include: precise determination of D_{it} distribution (due to uncertainty in surface potential), measurement of D_{it} near the band edges, and limitations imposed by severely distorted C-V characteristics. The conductance and charge pumping methods rely on direct measurement of dynamic charge exchange with the interface states. These techniques are not affected by C-V distortion but do require low minority carrier generation in the depletion regime (conductance method) and low gate leakage (conductance and charge pumping methods). For charge pumping, band edge states are difficult to detect and without the use of more complex procedures this technique yields an average D_{it} value rather than the distribution [59]. The conductance method is also inaccurate near the band edges and determination of the surface potential, and hence D_{it} distribution, is complicated by the presence of interface states. In contrast to the conductance and charge pumping methods, the quasistatic technique utilizes both low-frequency (quasistatic) and high-frequency C-V characteristics. In principle, this technique allows the D_{it} distribution to determined across the bandgap; however, interfaces with a high density of states may exhibit severe C-V distortion making it difficult to obtain true quasistatic and high frequency C-V characteristics [59]. Temperature-based techniques, such as the Gray-Brown method (temperature dependence of V_{FB}) and MOSFET V_t temperature dependence, offer several advantages over the techniques discussed thus far. First, flatband and threshold voltage shifts are easily measured. Second, for both the flatband and threshold voltage, the corresponding surface potential is known from basic theory which allows the D_{it} distribution to be determined. And third, temperature-based techniques are effective at measuring D_{it} near the band

edges. The Gray-Brown and low temperature MOSFET V_t methods also have important drawbacks that must be considered when using these techniques. The primary disadvantage of the Gray-Brown technique is that high measurement frequencies are required to ensure that the flatband shift is only due to the temperature dependence of the Fermi level and not trap freeze out at low temperature [73]. Additional flatband shift due to decreased trap response results in an overestimation of D_{it} [73]. The measurement frequency required to ensure true high-frequency C-V characteristics (i.e. no contribution from interface states) is sample specific; therefore, this technique is best suited for qualitative sample comparison [73]. To the best knowledge of the author, the accuracy of the low temperature MOSFET V_t method has not been studied in detail. This method relies heavily on the definitions used for threshold voltage and inversion, as well as the model used to determine the ideal V_t shift (i.e. without interface states). It assumes that any additional observed V_t shift is due to interface states.

In this work, interface state characterization was performed using the conductance, Gray-Brown, and MOSFET V_t (discussed in Chapter 4) methods. The conductance method allows rapid determination of midgap D_{it} while the temperature-based techniques measure near-band-edge D_{it} . This section discusses the characterization of WN/Al₂O₃/AlN capacitors using the conductance and Gray-Brown methods.

Interface state characterization was performed on n- and p-Ge capacitors with 2.5 nm AlN interlayer. The post-metal anneal condition for these samples was 450 °C for 30 min. in FG. The conductance method is accurate from flatband to weak inversion [59], so n-Ge and p-Ge were used to extract D_{it} in the lower and upper halves of the bandgap, respectively. The surface potential, ϕ_s , was determined by integrating same-device quasistatic C-V characteristics according to the following relation [59]:

$$\phi_{s} = \int_{V_{G1}}^{V_{G2}} \left(1 - \frac{C_{lf}}{C_{ox}}\right) dV_{G} + \Delta$$
(2.1)

where C_{lf} is the quasistatic capacitance and C_{ox} is the peak oxide capacitance. Integration was carried out from strong inversion to strong accumulation yielding a total change in surface potential of 0.70 V. This approximately matches the bandgap of germanium (0.66 eV) indicating that the method is valid [59]. In strong inversion (accumulation), the Fermi level for an n-Ge (p-Ge) capacitor is near the valence band edge, so the valence band edge is taken as the surface potential reference and Δ is assumed to be zero in equation 2.1. Figure 2.14 shows the resulting D_{it} distribution referenced to the intrinsic Fermi level. Midgap D_{it} is approximately $2x10^{12}$ cm⁻² eV⁻¹ and is slightly higher in the upper half of the bandgap. For comparison, midgap D_{it} of Si-SiO₂ interfaces is typically on the order of 10^{10} cm⁻² eV⁻¹.



Figure 2.14: Interface state density (D_{it}) determined using the conductance method for WN/ 3.5 nm Al₂O₃/ 2.5 nm AlN/Ge gate capacitors that received a 30 min. 450 °C FG anneal. This method is accurate from flatband to weak inversion, so n-Ge and p-Ge were used to extract D_{it} in the lower and upper halves of the bandgap, respectively. The energy position in the bandgap is referenced to the intrinsic Fermi level.

In addition to the conductance method, the Gray-Brown method was also used in order to extract D_{it} closer to the band edges [60]. In this method, the temperature dependence of the Fermi level is used to sweep the Fermi level position within the upper and lower halves of the bandgap in n-Ge and p-Ge, respectively. The corresponding change in flatband voltage is directly related to the interface state density within the energy range swept out by the Fermi level. At flatband, the surface potential is zero so the region of the bandgap swept out by the Fermi level position is easily calculated from basic theory thus allowing the D_{it} distribution to be determined. This technique assumes that true high-frequency curves (i.e. no contribution from interface states) are measured at each temperature [73]. This is most likely to be true at lower measurement temperatures due to trap freeze out [73]. Figure 2.15 shows forward sweep 1 MHz C-V characteristics as a function of temperature for n- and p-Ge. The flatband capacitance shown on each plot was determined by simulating an ideal germanium capacitor with similar substrate doping and CET. The flatband voltage is simply the voltage at which the capacitance equals the flatband capacitance. The n-Ge capacitor shows a larger flatband shift, qualitatively indicating a higher density of states in the upper half of the bandgap. Making the assumptions discussed above, the extracted D_{it} distribution is shown in Figure 2.16. D_{it} extracted from the conductance method has also been included for reference. This plot shows an asymmetric interface state distribution that rises more quickly near the conduction band edge. At E- E_i = 0.3 eV, D_{it} is approximately 3.5×10^{13} cm⁻² eV⁻¹. The D_{it} values extracted using the two techniques appear to be consistent.



Figure 2.15: Forward sweep 1 MHz C-V characteristics as a function of temperature for WN/ 3.5 nm $Al_2O_3/$ 2.5 nm AlN gate stack on p- and n-Ge. The n-Ge capacitor shows a larger flatband shift indicating a higher density of states in the upper half of the bandgap.



Figure 2.16: D_{it} distribution for WN/ 3.5 nm Al₂O₃/ 2.5 nm AlN/Ge capacitors extracted using conductance and Gray-Brown methods. The energy position in the bandgap is referenced to the intrinsic Fermi level.

In addition to extracting D_{it} from flatband shift, simple examination of the low temperature C-V characteristics also provides insight into the germanium-AIN interface. Figure 2.17 shows n-Ge C-V characteristics at 297, 250, 200, 150, 100, and 80 K. As discussed previously, and as shown at 297 K, hysteresis decreases during the reverse voltage sweep. The peak capacitance also decreases at high frequency and/or low-temperature which points to a large density of "trapped" electrons in accumulation. At high frequency, the trap time constant, τ_{it} , is long relative to the small-signal period and the trapped carriers cannot respond. At low temperature the trap emission time, and consequently τ_{it} , increase [73], thus producing the same effect. At 297 K and 10 kHz, n- and p-Ge have similar peak capacitance; however, at high frequency or low temperature, the "trapped" electrons do not contribute to small-signal AC capacitance. This frequency dispersion is not consistent with other possible explanations such as series resistance or capacitor perimeter effects. In addition to reduced peak capacitance, the hysteresis of n-Ge capacitors also changes significantly from 297 to 80 K. At low temperature, the emission time of trapped carriers becomes long resulting in constant hysteresis over the reverse sweep. For comparison, the low temperature C-V characteristics of p-Ge capacitors are shown in Figure 2.18. As expected, midgap interface states freeze out at low temperature producing more-ideal C-V characteristics [73]. A direct comparison of n- and p-Ge C-V characteristics at 297, 250, and 200 K is shown in Figure 2.19. From this data, it is very clear that carrier trapping severely impacts electrons at the germanium-AlN interface. The percentage of trapped electrons can be estimated by comparing the integrated charge from the 50 kHz and 1 MHz curves at 297 K. This estimate is most accurate at high inversion charge density. At low inversion charge density, small differences in integrated charge caused by factors not related to trapping may result in an inaccurate estimate. Figure 2.20 shows the integrated charge of both curves (secondary axis) along with the percentage of trapped charge (pri-
mary axis). For electron accumulation charge densities approaching 1×10^{13} cm⁻², approximately 15% of the carriers appear to be trapped. In contrast, p-Ge does not show evidence of significant charge trapping.



Figure 2.17: C-V characteristics of WN/ 3.5 nm $Al_2O_3/$ 2.5 nm AlN/n-Ge capacitors at 297, 250, 200, 150, 100, and 80 K. Forward and reverse sweeps are shown at frequencies of 50 kHz, 100 kHz, and 1 MHz



Figure 2.18: C-V characteristics of WN/ 3.5 nm $Al_2O_3/$ 2.5 nm AlN/p-Ge capacitors at 297, 250, 200, 150, 100, and 80 K. Forward and reverse sweeps are shown at frequencies of 50 kHz, 100 kHz, and 1 MHz.



Figure 2.19: Direct comparison of low temperature C-V characteristics of WN/ 3.5 nm $Al_2O_3/2.5$ nm AlN gate stack on n- and p-Ge at 297, 250, and 200 K. n-Ge shows a significant reduction in peak capacitance at high frequency and/or low temperature due to trapping.



Figure 2.20: Integrated charge from 297 K 50 kHz and 1 MHz C-V characteristics of WN/ 3.5 nm $Al_2O_3/$ 2.5 nm AlN/n-Ge capacitors. The charge difference is attributed to carrier trapping. The percentage of trapped charge is plotted on the primary axis while integrated charge is plotted on the secondary axis.

To further investigate the reduced small-signal capacitance in n-Ge, quasistatic C-V measurements were performed to compare the peak DC capacitance of n- and p-Ge capacitors. Figure 2.21 shows quasistatic and high-frequency data for n- and p-Ge. The quasistatic and high-frequency data show better agreement for hole accumulation than for electron accumulation. Furthermore, the peak DC electron capacitance is always higher than the peak DC hole capacitance suggesting that the reduction in n-Ge small-signal capacitance is significant and that electron trapping is occurring at the interface.



Figure 2.21: Quasistatic and high-frequency C-V data for WN/ 3.5 nm $Al_2O_3/2.5$ nm AlN gate stack on n- and p-Ge.

2.6 Summary

AlN is one of only a few dielectric layers that have produced functional Ge n-MOSFETs; however, AlN interlayers have yet to be studied in detail. In this study, the effects of AlN thickness and PMA conditions on Al_2O_3/AlN dielectric stacks were investigated. Interface state characterization was performed on Ge capacitors fabricated using an optimized gate stack and post-metal anneal.

Optimal C-V characteristics were achieved for an AlN interlayer thickness of 2.5 nm and either a 500 °C 60 s N₂ or 450 °C 30 min. FG anneal. As shown in Figure 2.13, these capacitors are still far from ideal with hysteresis values ranging from 100- 300 mV, midgap distortion for p-Ge, and evidence of electron trapping in n-Ge. Gate stacks annealed in N₂ were stable up to 500 °C with noticeable degradation observed at 550 °C. Interface state density was extracted for capacitors with 2.5 nm AlN interlayers using both the conductance and Gray-Brown methods. D_{it} was approximately $2x10^{12}$ cm⁻² eV⁻¹ near midgap and increased to $2x10^{13}$ and $3.5x10^{13}$ cm⁻² eV⁻¹ near the valence and conduction band edges, respectively. Low temperature C-V measurements also revealed a decrease in peak capacitance at low temperature and/or high measurement frequency for n-Ge capacitors. This behavior was attributed to electron trapping at the interface, and based on the observed frequency dispersion, the percentage of trapped carriers at high accumulation charge density was estimated to be approximately 15%.

Despite the non-idealities of the Al₂O₃/AlN dielectric stack, its properties are sufficient for fabricating functional Ge MOSFETs. Furthermore, additional optimization of this gate stack may be possible. The process space investigated in this work was limited to RCA-based cleaning followed by ALD deposition and post metal annealing in either N₂ for 60 s or 8:1 N₂:H₂ for 30 min. Each of these steps could be investigated in more detail. Other surface preparation techniques and more complex interlayers (e.g. AlON, HfON, HfAlON) could certainly be explored. As has been observed with silicon [75], interlayers with reduced nitrogen content may improve carrier mobility. Recent evidence also suggests that alternative surface preparations may be superior to the RCA-based cleans used in this work. Surface preparation and interlayer formation are dependent processes that must be co-optimized. It was recently discovered that dielectric stacks deposited on asreceived wafers from the supplier (Umicore) exhibit better electrical properties than on RCA-cleaned wafers. Furthermore, for as-received wafers, pure Al2O3 appears to perform better than Al₂O₃/AlN stacks. Since the supplier's surface preparation technique is proprietary, it is difficult to speculate why this may be the case; however, it seems likely that pre-existing germanium oxide is acting as an interlayer. The thermal stability of this apparent interlayer is unknown. Despite these new results, AlN interlayers are still a promising approach for improving the germanium-dielectric interface. In the next chapter, the electrical characteristics of surface-channel Ge MOSFETs with AIN interlayers are discussed.

Chapter 3

Surface-Channel Ge MOSFETs

3.1 Introduction

To evaluate device performance, Ge MOSFETs were fabricated with the optimized WN/ Al_2O_3/AlN gate stack described in Chapter 2. n-FETs were fabricated on both (100) and (111) bulk germanium substrates to examine the impact of substrate orientation on electron mobility. Ge MOSFETs with Hf₃N₄ and GeON interlayers were also investigated through collaborations with Harvard University [41], the University of Texas at Austin [33], [61], and the National Center for Scientific Research (NCSR) Demokritos in Athens, Greece [62].

Section 3.2 provides detailed electrical characteristics for devices with AlN interlayers. Sections 3.3, 3.4, and 3.5 describe device results for Hf_3N_4 , RTP GeON, and MBD GeON interlayers, respectively. In addition to bulk germanium, epitaxial strained germanium p-MOSFETs were also fabricated with RTP GeON interlayers and are described in Section 3.4. Lastly, Section 3.6 summarizes the carrier mobility for Ge MOSFETs fabricated at MIT with nitride-based interlayers.

3.2 AlN Interlayer MOSFETs

3.2.1 Fabrication Process

n- and p-MOSFETs were fabricated using a standard process that is summarized in Table 3.1. A detailed process description may be found in Appendix A. Annular devices were fabricated on bulk (100) Ge substrates (Ga: 0.12-0.17 Ω-cm, Sb: 0.13-0.16 Ω-cm). n-FETs were also fabricated on bulk (111) Ge substrates (Ga: 0.084-0.130 Ω -cm). The pre-gate stack cleaning procedure consisted of three steps: (1) 10 min. 6:1 DI:HCl followed by DI rinse, (2) 2 min. 5:1:1 DI: NH₄OH: H₂O₂ followed by DI rinse, and (3) 10 min. 6:1 DI:HCl followed by DI rinse. The first step removes native oxide, the second step etches germanium to provide a fresh surface prior to gate dielectric deposition, and the third step removes chemical oxide and chlorine-terminates the surface to reduce subsequent GeO_x formation [24]. Chlorine termination is not perfectly stable [24] so some oxide forms on the surface prior to gate stack deposition. The *in-situ* ALD gate stack consisted of a 2.5 nm AlN interlayer, 3.5 nm Al₂O₃ capping layer, and 40 nm WN gate electrode. The gate electrode was etched using CF₄-based reactive ion etching. Next S/D implantation was performed (P: 5×10^{14} cm⁻², 25 keV; BF₂: 1×10^{15} cm⁻², 40 keV), followed by PECVD oxide deposition, S/D activation, and contact photolithography and etching. Ti/Al was used for the metallization. S/D anneal conditions for p- and n-FETs were 450°C 60 s N_2 and 500°C 60 s N₂, respectively. The S/D anneal also served as the post-metallization anneal (PMA) for the gate stack.

Step	Process	Description
1	Pre-gate stack clean	Modified RCA clean: dilute HCl + SC1 (RT) + dilute HCl (DI rinse after each step)
2	ALD gate stack deposition	40 nm WN/3.5 nm Al ₂ O ₃ /2.5 nm AlN
3	Gate electrode patterning	CF ₄ RIE
4	S/D ion implantation	PMOS: BF ₂ , $1x10^{15}$ cm ⁻² , 40 keV NMOS: P, $5x10^{14}$ cm ⁻² , 25 keV
5	PECVD SiO ₂ deposition	200 nm
6	S/D activation	PMOS: 450 °C, N ₂ , 60 s NMOS: 500 °C, N ₂ , 60 s
7	Contact hole patterning	3:1 DI: BOE followed by 4:1 DI:HCl to insure high-k dielectric removal
8	Metal deposition	Frontside: 50nm Ti/ 400 nm Al Backside: 50nm Ti/ 2 µm Al
9	Metal patterning	Al: Al etchant Ti: dilute BOE

 Table 3.1: Fabrication process summary for AIN interlayer Ge MOSFETs. Detailed process description may be found in Appendix A.

3.2.2 Device Characterization and Results

(100) p- and n-FETs

Typical as-measured p- and n-FET I_S - V_G and I_S - V_D characteristics are shown in Figures

3.1 and 3.2. Source current is measured to eliminate the effect of drain junction leakage to

the substrate.

.



Figure 3.1: Typical as-measured I_S-V_G characteristics for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. p-FETs exhibited a 95 mV/decade sub-threshold swing and V_t of -0.05 V. n-FETs exhibited a V_t of +0.7 V, gradual roll-off of the I_S-V_G characteristic in the subthreshold regime, and increased device-to-device variability.



Figure 3.2: Typical as-measured I_S - V_D characteristics for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack.

The Ge p-FETs exhibited a 95 mV/decade subthreshold swing and -0.05 V threshold voltage (V_t). Device characteristics (V_t, source current, carrier mobility) were uniform across the wafer. The n-FETs exhibited a V_t of +0.7 V, gradual roll-off of the I_S-V_G characteristics in the subthreshold regime, and increased device-to-device variability. The work function of WN is 4.6 eV [63], which approximately aligns with the germanium valence band edge. The ideal threshold voltage (i.e. assuming zero fixed and interface state charge) for these devices is approximately +0.05 V for the p-FET and +0.5 V for the n-FET. The measured p-FET V_t is higher by |0.1 V| while the n-FET V_t is higher by |0.2 V|. These shifts are due to both fixed charge and interface states in the upper and lower halves of the bandgap for n- and p-FETs, respectively. The n-FET V_t shift combined with I_S-V_G subthreshold distortion suggests a higher interface state density in the upper half of the bandgap. Figure 3.3 shows the inversion-side split C-V characteristics (C_{GSD}) at low frequency (10 kHz). The forward sweep is from depletion to inversion, and vice versa for the reverse sweep. The peak inversion capacitance corresponds to a capacitance equivalent thickness (CET) of approximately 3 nm and is similar for both the n- and p-FET. Not surprisingly, the n- and p-FET inversion-side split C-V characteristics are similar to the n- and p-Ge MOSCAP results in Chapter 2. The p-FET C_{GSD} shows low hysteresis and slight distortion at midgap due to interface states. The n-FET C_{GSD} shows large hysteresis, distortion near inversion, and slow detrapping on the reverse sweep. The effective carrier mobility for these devices was extracted from same-device I_S-V_G and C_{GSD} characteristics and is shown in Figure 3.4. Silicon universal hole mobility at similar substrate doping (5x10¹⁶ cm⁻³) is included for reference. Silicon universal mobility is typically defined in terms of vertical effective field, which is calculated using empirically determined multiplying factors for electron and hole inversion charge density (1/2 and 1/3, respectively). These factors have not been verified for germanium, so germanium carrier mobility is often plotted versus inversion charge density. As was done for the silicon universal reference curve in Figure 3.4, vertical effective field can easily be translated into inversion charge density for a given substrate doping.



Figure 3.3: Typical as-measured inversion-side split C-V characteristics for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stack. The peak inversion capacitance corresponds to a capacitance equivalent thickness (CET) of approximately 3 nm and is similar for both the n-FET and p-FET.



Figure 3.4: Effective carrier mobility for Ge p- and n-MOSFETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stack. p-FETs show slight enhancement compared to silicon universal while n-FETs show peak mobility of only 50 cm²/Vs.

The Ge p-FETs show a peak mobility of 140 cm²/Vs and slight enhancement over silicon universal hole mobility. Given the 4x difference in bulk hole mobility, larger enhancement is expected. The bulk mobility enhancement factor is not realized in inversion layer mobil-

ity due to the non-ideal germanium-dielectric interface. Nonetheless, this result is still impressive given the early stage of this work. Mobility enhancement of up to 40% is typically reported for surface-channel Ge MOSFETs with high-k gate dielectrics and no silicon interlayer. In contrast, Ge n-FETs show a disappointing peak mobility of 50 cm²/Vs which is far below silicon universal electron mobility. The mobility peak is also pushed out to high inversion charge density due to coulomb scattering from charged interface states. The gradual roll-off of the I_S-V_G characteristics, early saturation behavior of the I_S-V_D characteristics, C-V distortion and hysteresis, and poor mobility all point toward problems with the germanium-dielectric interface that impact electrons more severely than holes. Chapter 4 investigates this topic in detail.

(111) n-FETs

Ge n-FETs were also simultaneously fabricated on (111) substrates to determine whether band structure was playing a significant role in the observed electron mobility. Figure 3.5 shows a comparison between the effective electron mobility for (100) and (111) Ge n-FETs. Because of high device-to-device variability, mobility from five devices is shown for each substrate orientation.



Figure 3.5: Comparison of effective electron mobility for (100) and (111) Ge n-MOS-FETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stack. Mobility from five devices is shown for each substrate orientation.

Ge (111) n-FETs show slightly lower mobility than (100) devices and the peak occurs at higher inversion charge density. As with silicon, the (111) plane has a higher density of atoms, and thus a larger number of bonds that must be terminated, so it is not surprising that (111) devices show degraded mobility relative to (100). While band structure certainly plays an important role in electron transport in germanium, the observed mobility in these devices is clearly limited by the germanium-dielectric interface.

3.3 Hf₃N₄ Interlayer MOSFETs

 Hf_3N_4 interlayers were explored through a collaboration with K. Kim and R. Gordon of Harvard University. Gate stack deposition and post-metal annealing were performed at Harvard, and device fabrication was completed at MIT. Device results from this collaboration are published in reference 62.

3.3.1 Fabrication Process

Annular Ge n- and p-MOSFETs were fabricated on bulk (100) Gc substrates (Ga: 0.2-0.4 Ω -cm, Sb: 0.025-0.04 Ω -cm). The substrates were cleaned in dilute HF (10%) for 5 min. and rinsed in DI water for 30 s. The *in-situ* ALD gate stack consisted of a 1.5 nm Hf₃N₄ interfacial layer, 2.5 nm GdScO₃ capping layer, and 40 nm WN gate electrode [62]. GdScO₃ has shown promise as a high-k dielectric on silicon and thus was used in this work [80]. After gate stack deposition, post-metal annealing was performed at 420°C for 90 s in 10:1 He:H₂ (10 torr). The gate electrode was then etched using CF₄-based reactive ion etching. Next S/D implantation was performed (P: 5×10¹⁴ cm⁻², 15 keV; BF₂: 1×10¹⁵ cm⁻², 24 keV), followed by PECVD oxide deposition, S/D activation, and contact photoli-thography and etching. Ti/Al was used for the metallization. S/D anneal conditions for n-and p-FETs were 500°C 60 s N₂ and 400°C 60 s N₂, respectively.

3.3.2 Device Characterization and Results

Typical as-measured p- and n-FET I_S-V_G characteristics are shown in Figure 3.6. The p-FETs exhibited a 114 mV/decade subthreshold swing and -0.65 V threshold voltage. The ideal p-FET threshold voltage is +0.0 V. The gradual I_S-V_G roll-off and high $|V_t|$ indicate a large density of states in the lower half of the bandgap. The n-FETs exhibited a V_t of +0.45 V and steeper subthreshold swing of 103 mV/decade. The ideal n-FET threshold voltage is approximately +0.5 V. The agreement between measured and ideal V_t is better than for WN/Al₂O₃/AlN Ge n-FETs, suggesting lower D_{it} in the upper half of the bandgap for Hf₃N₄ interlayers.



Figure 3.6: Typical as-measured I_S - V_G characteristics for Ge p- and n-MOSFETs with 40 nm WN/2.5 nm GdScO₃/1.5 nm Hf₃N₄ gate stack. p-FETs exhibited a 114 mV/decade subthreshold swing and -0.65 V threshold voltage (V_t). The gradual I_S - V_G roll-off and high V_t indicate a large density of states in the lower half of the bandgap. The n-FETs exhibited a V_t of +0.45 V and steeper subthreshold swing of 103 mV/decade.

Figure 3.7 shows inversion-side split C-V characteristics (C_{GSD}) at low frequency (10 kHz). The peak inversion capacitance corresponds to a capacitance equivalent thickness (CET) of approximately 1.5 nm and is similar for both the n-FET and p-FET. As expected from the I_S-V_G characteristics, the n-FET C_{GSD} characteristics shows less distortion and hysteresis. The effective carrier mobility for these devices is shown in Figure 3.8. Peak electron and hole mobilities are 90 and 17 cm²/Vs, respectively. While the electron mobility is promising, the hole mobility for Hf₃N₄ is significantly lower than for alternative interlayers such as GeON and AlN. It is certainly possible that gate stacks for Ge n- and p-FETs will need to be optimized separately.



Figure 3.7: Typical as-measured split C-V characteristics at 10 kHz for Ge p- and n-MOSFETs with 40 nm WN/2.5 nm $GdScO_3/1.5$ nm Hf_3N_4 gate stack. The peak inversion capacitance corresponds to a capacitance equivalent thickness (CET) of approximately 1.5 nm and is similar for both the n-FET and p-FET.



Figure 3.8: Effective carrier mobility for Ge p- and n-MOSFETs with 40 nm WN/2.5 nm GdScO₃/1.5 nm Hf₃N₄ gate stack. n-FETs show promising peak electron mobility of 90 cm²/Vs while p-FETs show severely degraded hole mobility.

3.4 RTP GeON Interlayer MOSFETs

Gate stacks with RTP GeON interlayers were explored through a collaboration with D.L. Kwong at the University of Texas at Austin. TaN/HfO₂/GeON gate stack deposition was performed at the University of Texas at Austin, and device fabrication was completed at MIT. In addition to bulk germanium substrates, p-FETs were also fabricated on epitaxial strained germanium. GeON interlayer p-FETs showed reasonable performance; however, n-FETs showed very low drive current (0.01 μ A/ μ m) and are not discussed further in this section. Device results from this collaboration are published in reference 61.

3.4.1 Strained Germanium Heterostructure

Figure 3.9 describes the Ge/SiGe heterostructure and Figure 3.10a shows an XTEM micrograph of the structure. The strained Ge channel is approximately 55 nm thick. This layer was intentionally grown thicker than its critical thickness ($h_c \sim 12$ nm) to withstand pre-gate cleaning steps. While the compressive strain is partially relaxed by misfit dislocations at the heterointerface, the strained Ge maintains approximately 0.8% elastic strain after device processing (the lattice mismatch between Ge and Si_{0.3}Ge_{0.7} is 1.3%). Device fabrication on strained germanium virtual substrates was performed at reduced temperature compared to bulk germanium substrates to prevent strain relaxation and interdiffusion.



Figure 3.9: Schematic of Ge/SiGe heterostructure grown using UHVCVD. The strained germanium layer is approximately 55 nm thick. All SiGe layers are relaxed. After growing the 0-60% SiGe graded buffer, the wafers were polished using CMP to remove the cross-hatch that arises during growth (from [61]).



Figure 3.10: (A) XTEM of strained Ge after processing. The strain present in the film may be estimated from the dislocation density as seen in XTEM micrographs [64]. Applying this technique yields a strain of 0.8% in the germanium channel layer. (B) XTEM of gate stack on strained Ge after processing. The HfO_2 layer is approximately 5 nm thick. The germanium oxynitride interfacial layer is clearly visible (from [61]).

3.4.2 Fabrication Process

Long-channel annular p-MOSFETs were fabricated on bulk (100) Ge substrates (Sb: 0.05-0.3 Ω -cm) and epitaxial strained Ge on Si_{0.3}Ge_{0.7} virtual substrates. The substrates were cleaned using cyclic 50:1 DI:HF [33]. The GeON interlayer was then formed by rapid thermal annealing in ammonia (NH₃) at 600 °C for 2 min. for bulk Ge and 500 °C for 2 min. for strained Ge. This process forms a thin (~0.7 nm) germanium oxynitride layer [33]. Next, 5 nm of HfO₂ was deposited by MOCVD followed by 140 nm of reactively sputtered TaN as the gate electrode. Figure 3.10b shows an XTEM micrograph of the gate stack. The gate electrode was etched using CF₄-based reactive ion etching (RIE). Following gate patterning, HfO₂ was removed from the S/D regions using CF₄ RIE. Next S/D implantation was performed (BF₂: 1 × 10¹⁶ cm⁻², 35 keV), followed by PECVD oxide deposition, S/D activation, and contact photolithography and etching. Ti/Al was used for the metallization. S/D anneal conditions for the bulk and strained Ge p-FETs were 450 °C for 30 min. in N₂ followed by 600 °C for 2 min. in N₂, and 450 °C for 30 min. in N₂, respectively.

3.4.3 Device Characterization and Results

Figures 3.11 and 3.12 show I_S-V_G and split C-V characteristics for the bulk Ge p-MOS-FETs. These devices have a -0.19 V threshold voltage, 80 mV/decade subthreshold swing, and 1.8 nm CET. I_S-V_G and split C-V characteristics for the strained Ge p-MOSFETs are shown in Figures 3.13 and 3.14. The strained Ge devices have a -0.23 V threshold voltage, 90 mV/decade subthreshold swing, and 1.6 nm CET.



Figure 3.11: Typical as-measured I_S - V_G characteristics for bulk Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. These devices exhibited a –0.19 V threshold voltage and 80 mV/decade subthreshold swing (from [61]).



Figure 3.12: Typical as-measured split C-V characteristics for bulk Ge p-MOSFETs with 140 nm TaN/5 nm $HfO_2/0.7$ nm GeON gate stack. The inversion-side capacitance corresponds to a CET of 1.8 nm (from [61]).



Figure 3.13: Typical as-measured I_S -V_G characteristics for strained Ge p-MOSFETs with 140 nm TaN/5 nm HfO₂/0.7 nm GeON gate stack. These devices have a -0.23 V threshold voltage and 90 mV/decade subthreshold swing (from [61]).



Figure 3.14: Typical as-measured split C-V characteristics for strained Ge p-MOSFETs with 140 nm TaN/5 nm $HfO_2/0.7$ nm GeON gate stack. The inversion-side capacitance corresponds to a CET of 1.6 nm (from [61]).

The bulk Ge devices exhibit slightly lower subthreshold swing than the strained Ge devices. This is not surprising since the bulk Ge nitridation was performed at 600 °C (vs. 500 °C for strained Ge) which results in a better interface with fewer traps [33]. Figure 3.15 compares the effective hole mobility for bulk and strained Ge p-FETs. The strained Ge devices show a 35% increase in hole mobility at high vertical effective field compared to bulk Ge devices. This enhancement is encouraging given the low level of strain present in the germanium channel and the imperfect germanium-dielectric interface.



Figure 3.15: Comparison of effective hole mobility for bulk and strained Ge p-MOSFETs with 140 nm TaN/5 nm $HfO_2/0.7$ nm GeON gate stack. At high vertical effective field, strained Ge p-FETs show a 35% increase in mobility compared to the bulk Ge devices (adapted from [61]).

3.5 MBE GEON Interlayer MOSFETs

In addition to RTP GeON, GeON interlayers deposited by molecular beam deposition (MBD) were also explored through a collaboration with A. Dimoulas of NCSR Demokritos in Athens, Greece. TaN/HfO₂/GeON gate stack deposition was performed at NCSR, and device fabrication was completed at MIT. MBD offers a completely different approach to surface preparation prior to gate stack deposition. Rather than ex-situ wet cleaning, MBD allows in-situ desorption of germanium surface oxides under UHV conditions immediately prior to gate stack deposition. In this work, native germanium oxide was desorbed at 360 °C for 15 min. while monitoring the RHEED surface reconstruction pattern [65]. The surface was then exposed to nitrogen and oxygen atomic beams while evaporating germanium at 0.05 Å/s to form an ultra-thin (less than 0.5 nm) GeON interlayer. The gate stack was completed by *in-situ* deposition of 5 nm HfO₂ and 70 nm TaN. Molecular beam deposition of the GeON interlayer allows precise thickness control and consequently, very thin interlayers. In this work, an equivalent oxide thickness of 0.7 nm was demonstrated indicating that aggressive EOT scaling is possible for gate stacks on germanium. In general, germanium forms thinner interfacial layers than silicon for high-k deposition directly on cleaned substrates and this is perhaps one benefit of GeO_x volatility [26]. As with RTP GeON, the p-FETs showed reasonable device characteristics with carrier mobility similar to silicon universal hole mobility while the n-FETs were essentially non-functional. Device results from this collaboration are published in reference 62.

3.6 Summary of MIT Ge MOSFET Mobility

Ge n- and p-MOSFETs were fabricated with GeON, AlN, and Hf_3N_4 interlayers that were formed using techniques ranging from thermal nitridation to atomic layer deposition and molecular beam deposition. Despite the range of materials and deposition methods investigated, device performance was relatively consistent indicating a degree of commonality in nitride-germanium interfaces. Figure 3.16 shows representative carrier mobility data for nitride interlayer Ge MOSFETs fabricated at MIT. p- and n-FET data are shown by open and closed symbols, respectively.



Figure 3.16: Summary of carrier mobility for nitride interlayer Ge MOSFETs fabricated at MIT. Representative data shown for p-FETs (open symbols) and n-FETs (closed symbols).

As shown in the figure, hole mobility generally matches or exceeds silicon universal hole mobility. In addition, p-FET device characteristics were generally distortion-free with subthreshold swing less than 100 mV/decade and reasonable C-V hysteresis (~100 mV). In contrast, ALD nitride interlayer n-FETs show poor mobility (typically 50-100 cm²/Vs), distorted device characteristics, and larger C-V hysteresis (~200 mV). n-FETs based on GeON interlayers were essentially non-functional. Amongst ALD interlayers, Hf₃N₄ produced higher electron mobility (100 cm²/Vs) while AlN produced higher hole mobility (150 cm²/Vs). The mobilities observed in this work are consistent with other published reports for surface-channel Ge MOSFETs with high-k dielectrics and no silicon interlayer. Based on all of the published data, the path towards high-mobility Ge p-FETs is clear; however, the same cannot be concluded for Ge n-FETs. To date, Ge n-FET device characteristics have been extremely disappointing and have contributed to renewed interest in III-V n-FETs. While many theories have been proposed to explain poor Ge n-FET performance, careful examination of carrier transport in surface-channel Ge MOSFETs has yet to be performed. In Chapter 4, carrier transport in Ge MOSFETs with WN/Al₂O₃/AlN gate stacks is experimentally investigated through low-temperature device characterization and the analysis of both surface-channel and buried-channel MOSFETs.

Chapter 4

Investigation of Carrier Transport in Ge MOSFETs with AlN Interlayers

4.1 The Ge NMOS Problem

As shown in Chapters 1 and 3, Ge MOSFETs show promising hole mobility but very poor electron mobility. Many theories have been proposed to explain poor Ge n-FET performance:

- high series resistance due to inadequate n-type activation in the S/D regions [78],
- electrochemical difference between dielectric formation on n- an p-Ge [66],
- poor thermal stability of the germanium-dielectric interface at n-type S/D activation temperatures [50],
- fundamental band structure issue (e.g. L- Δ intervalley electron scattering), and
- asymmetric distribution of interface states [38].

Although commonly cited as a reason for poor Ge n-FET performance, series resistance should not impact mobility extraction from long-channel (L_G = 20- 100 µm) devices. Active n-type dopant concentrations of 5x10¹⁹ cm⁻³ and sheet resistance values of 75 Ω/ sq. (x_j = 100 nm) have been demonstrated [18] and are more than adequate for mobility extraction from long-channel devices.

An electrochemical effect related to the presence of dopants in germanium substrates has also been suggested as a cause for poor n-FET performance. It has been theorized that Ga and Sb (common dopants in Ge wafers) affect interfacial layer formation, and thus CET, for identical gate stacks deposited on n- and p-Ge [66]. In this work, gate stacks deposited on n- and p-Ge [66]. In this work, gate stacks deposited on n- and p-Ge [66]. In this work, gate stacks deposited on n- and p-Ge (inversion-side of MOSFET split C-V characteristics) produces similar C-V characteristics to electron accumulation on n-Ge (MOSCAP C-V characteristics). The MOSFET split C-V characteristics show higher baseline capacitance due

to parasitic probe pad capacitance that is not modulated by the gate. At least for dielectric stacks with AlN interlayers, dielectric formation on n- and p-Ge does not produce significantly different C-V characteristics.



Figure 4.1: Comparison of C-V characteristics for electron inversion on p-Ge (MOSFET) and electron accumulation on n-Ge (MOSCAP). Both devices had WN/3.5 nm Al₂O₃/2.5 nm AlN gate stacks. MOSFET split C-V characteristics show higher baseline capacitance (dashed line) due to parasitic probe pad capacitance that is not modulated by the gate.

Poor gate stack thermal stability at n-type dopant activation temperatures (500- 600 °C) does affect interfacial quality and likely contributes to poor NMOS performance for some dielectrics. For example, GeON layers are known to exhibit poor thermal stability which likely degrades device performance [50]. AlN interlayers, however, exhibit stability up to at least 500 °C. In Chapter 2, the impact of post-metal annealing at 350-550 °C was explored and no significant change in C-V properties was observed at the n-type S/D activation temperature used in this work (500 °C).

Band structure related phenomena such as L-∆ intervalley scattering have also been proposed to explain the observed electron mobility in Ge n-FETs. In early published reports of high electron mobility in Ge n-FETs [e.g. 37], devices were often fabricated on

(111) substrates. In addition, recent simulations have shown that Ge n-FETs on (111) substrates should outperform those fabricated on (100) substrates [67]. However as discussed in Chapter 3, Ge n-FETs with WN/Al₂O₃/AlN gate stacks were fabricated on both (111) and (100) substrates and similar mobility was observed. This suggests that substrate orientation and consequently band structure are not responsible for the degraded mobility observed in AlN interlayer Ge n-FETs.

The last and most plausible explanation for poor n-FET mobility is the germaniumdielectric interface. More specifically, an asymmetric distribution of interface states with high density near the conduction band edge will result in severely degraded electron mobility while having less impact on hole mobility. Interface states in the upper half of the bandgap are acceptor-like (neutral when empty, negatively charged when filled) and therefore act as coulomb scattering sites in n-FETs but remain uncharged in p-FETs. Interestingly, an asymmetric D_{it} distribution resulting in electron mobility degradation was also observed during early work on HfO₂/SiON/Si gate stacks [68]. In the remainder of this chapter, the impact of interface states on carrier transport in Ge MOSFETs with WN/ Al₂O₃/AlN gate stacks is investigated. First, the interface state distribution within the bandgap is discussed. Second, the impact of this distribution on carrier mobility is determined through low-temperature device characterization and the analysis of both surfacechannel and buried-channel MOSFETs.

4.2 Interface Distribution for Germanium-AlN Interfaces

The interface state distribution for WN/Al₂O₃/AlN/Ge capacitors was extracted over most of the bandgap using the conductance and Gray-Brown methods as described in Chapter 2. The temperature dependence of MOSFET threshold voltage can be used to determine D_{it} even closer to the band edge (within 0.05 eV). At low temperature, the Fermi-Dirac distribution becomes more abrupt and consequently the Fermi level must move closer to the conduction band edge for the same inversion charge density. This results in the filling of additional interface states which produces a threshold voltage shift. In the ideal case of zero interface states, the temperature dependence of the threshold voltage is given by:

$$\frac{dV_t}{dT} = -\frac{1}{2q}\frac{dE_g}{dT} + (2m-1)\frac{d\psi_B}{dT}$$
(4.1)

where *m* is the MOSFET body-effect coefficient and ψ_B is the difference between the Fermi level and intrinsic level in the bulk of the semiconductor [15]. Any additional shift that is observed can be attributed to interface state charging. In this case, the interface density is simply:

$$D_{it} = \frac{(\Delta V_{t, meas} - \Delta V_{t, ideal}) \cdot C_{ox}}{q}$$
(4.2)

where $\Delta V_{t,meas}$ is the observed V_t shift and $\Delta V_{t,ideal}$ is the V_t shift in the absence of interface states. V_t was extracted at each temperature by linear extrapolation of the I_S-V_G characteristic at low $|V_{DS}|$ (50 mV). Using the popular definition for inversion, $\phi_s = 2\psi_B$, and knowing the temperature dependence of ψ_B , the position of the Fermi level relative to the intrinsic level is known at inversion. Accounting for the temperature dependence of E_g, the position of the Fermi level with respect to the band edge (E_c) can then be calculated at each temperature. Thus the V_t shift that occurs between temperatures T_1 and T_2 can be attributed to interface states between $(E_c - E_{F,i})_{T_1}$ and $(E_c - E_{F,i})_{T_2}$ where $E_{F,i}$ is the Fermi level position at inversion.

Low temperature measurements on Ge MOSFETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stacks were performed using a standard liquid nitrogen cryostat at temperatures of 297, 250, 200, 150, 100, and 80 K. The V_t shift at low temperature is qualitatively shown by the shift of the I_S-V_G and inversion-side split C-V characteristics in Figures 4.2 and 4.3, respectively. The positive threshold voltage shift for n-FETs and negative shift for

p-FETs confirms the acceptor-like nature of states in the upper half of the bandgap and donor-like nature of interface states in the lower half of the bandgap [60]. The large shift for n-FETs relative to p-FETs indicates a gross asymmetry in the interface state distribution near the band edge. Figure 4.4 shows the D_{it} distribution extracted from the temperature dependence of threshold voltage. D_{it} values extracted using the conductance and Gray-Brown methods are included for reference. As shown in the figure, the interface state density is clearly asymmetric. D_{it} near the valence band edge (VBE) rises slowly and saturates at approximately 1×10^{13} cm⁻² eV⁻¹. In contrast, D_{it} rises very rapidly near the conduction band edge (CBE) and approaches 6×10^{13} cm⁻² eV⁻¹. The number of additional interface states near the CBE relative to the VBE (indicated by dotted triangle in Figure 4.4) is approximately 2×10^{12} cm⁻² which is similar to the trapped charge density estimated from the low temperature MOSCAP measurements in Chapter 2. The clear asymmetry in near-band-edge D_{it} suggests that interface states are the principal cause of poor n-FET performance.

The next section discusses carrier transport in Ge MOSFETs with AlN interlayers. Low temperature mobility measurements were performed to better understand the mobility degradation mechanism. In addition, buried-channel MOSFETs were fabricated to investigate transport in devices with reduced interaction between carriers and interface states.



Figure 4.2: I_S - V_G characteristics as a function of temperature (297, 250, 200, 150, 100, and 80 K) for Ge p- and n-FETs with 40 nm WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack. The large V_t shift for n-FETs relative to p-FETs suggests higher D_{it} in the upper half of the bandgap, particularly near the conduction band edge.



Figure 4.3: Inversion-side split C-V characteristics (C_{GSD}) as a function of temperature (297, 250, 200, 150, 100, and 80 K) for Ge p- and n-FETs with 40 nm WN/3.5 nm Al₂O₃/ 2.5 nm AlN gate stack.



Figure 4.4: Interface state distribution for WN/3.5 nm Al₂O₃/2.5 nm AlN gate stack on germanium. Conductance, Gray-Brown, and temperature dependence of MOSFET V_t were used to map D_{it} across the bandgap. D_{it} rises rapidly near the conduction band edge and approaches 6×10^{13} cm⁻² eV⁻¹. The number of additional interface states near the CBE relative to the VBE (indicated by dotted triangle) is approximately 2×10^{12} cm⁻² which is similar to the trapped charge density estimated from the low temperature MOSCAP measurements in Chapter 2.

4.3 Low Temperature Mobility Characterization

As described in the previous section, low temperature measurements were performed on Ge MOSFETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stacks. Effective mobility was extracted from the same-device I_S-V_G and split C-V data shown in Figures 4.2 and 4.3, respectively. Figure 4.5 shows the effective electron and hole mobility as a function of temperature.



Figure 4.5: Carrier mobility as a function of temperature (297, 250, 200, 150, 100, and 80 K) for Ge p- and n-FETs with 40 nm WN/3.5 nm $Al_2O_3/2.5$ nm AlN gate stack. Electron mobility decreased with decreasing temperature indicating significant charge trapping in interface states near the conduction band edge. Distortion of the electron mobility curves can be attributed to coulomb scattering from charged interface states.

As expected, hole mobility increased with decreasing temperature due to reduced phonon scattering. In contrast, electron mobility decreased with decreasing temperature indicating significant charge trapping in interface states near the conduction band edge [69]. The low temperature n-FET mobility curves also show severe distortion due to coulomb scattering from charged interface states. This data shows that the degraded mobility of Ge n-MOS-FETs with AIN interfacial layers, and possibly Ge n-MOSFETs in general, is due to a large number of interface states near the conduction band edge that trap carriers and act as coulomb scattering sites. Trapping of carriers results in an overestimation of mobile charge and consequently the extracted mobility is lower than the true mobility. However, for small fractions of trapped charge (0.1- 0.2), the effective mobility (particularly at high inversion charge density) will only be reduced by a similar fraction. Assuming that the surface mobility ratio between germanium and silicon should be similar to the bulk mobility ratio, the best surface-channel electron mobility achieved in this work (160 cm²/Vs) is

still approximately 4x lower than expected. This severe reduction cannot be explained by carrier trapping and therefore must be due to carrier scattering (e.g. coulomb scattering from charged interface states and fixed charge). Published data on SiGe surface-channel p-MOSFETs with ALD TiN/Al₂O₃/HfAlO_x/Al₂O₃ gate stacks illustrates the relative impact of trapping and coulomb scattering on carrier mobility [70]. Figure 4.6 shows the mobility correction due to $1x10^{12}$ cm⁻² trapped carriers as well as coulomb scattering from $1x10^{12}$ cm⁻² and $4x10^{12}$ cm⁻² charged interface states. To correct for carrier trapping, mobile charge rather than total inversion charge is used for mobility extraction. To correct for coulomb scattering, the coulomb limited mobility (μ_{Cit}) is first estimated from:

$$\mu_{Cit} = 1.47 \times 10^{18} \frac{\sqrt{Q_i}}{N_{it, C}}$$
(4.3)

where Q_i is the mobile inversion charge density and $N_{il,C}$ is the density of charged interface states [70]. The corrected mobility (μ_{corr}) is then obtained from Mathiessen's Rule using the following expression:

$$\frac{1}{\mu_{corr}} = \frac{1}{\mu_{corr,Qi}} - \frac{1}{\mu_{Cii}}$$
(4.4)

where $\mu_{corr, Qi}$ is the mobility corrected only for trapped inversion charge [70].



Figure 4.6: Hole mobility for SiGe surface-channel p-MOSFETs with ALD TiN/Al₂O₃/ HfAlO_x/Al₂O₃ gate stack. As-extracted mobility is compared with mobility corrected for carrier trapping (Q_i), carrier trapping plus coulomb scattering from 1×10^{12} cm⁻² interface states, and carrier trapping plus coulomb scattering from 4×10^{12} cm⁻² interface states (from [70]).

As can be seen in the figure, the correction for trapped carriers is relatively small while coulomb scattering accounts for the majority of the observed mobility degradation. To confirm that interfacial trapping and scattering are primarily responsible for mobility degradation in Ge n-FETs with WN/Al₂O₃/AlN gate stacks, buried-channel devices were fabricated to reduce the interaction of carriers with the germanium-AlN interface.

4.4 Buried-Channel MOSFETs

4.4.1 Device Fabrication

Long-channel annular Ge p- and n-MOSFETs were fabricated on bulk (100) Ge substrates (Sb: 0.13-0.16 Ω -cm, Ga: 0.16-0.47 Ω -cm). First, an 11 nm SiO₂ screen oxide was deposited by low-temperature CVD followed by channel implantation. The n-FETs received phosphorus implants of 1×10¹² cm⁻² at 10 keV, and 2×10¹², 4×10¹², and 1×10¹³ cm⁻² at 15 keV while the p-FETs received BF₂ implants of 1×10¹² cm⁻² at 16 keV, and 2×10¹²,
4×10^{12} , and 1×10^{13} cm⁻² at 24 keV. Doses were chosen to be similar to typical MOSFET inversion charge densities. Dose and energy were also selected to ensure that some of the devices could be completely shutoff with gate bias. Table 4.1 shows the simulated implant depth and peak concentration for each of the doses. The depth is given with respect to the germanium surface. The P and BF₂ implant energies were adjusted to give the same implant depth.

Implant Dose (cm ⁻²)	N _{peak} (cm ⁻³)	Depth (nm)
1x10 ¹²	6x10 ¹⁷	at Ge-SiO ₂ interface
2x10 ¹²	9x10 ¹⁷	6
4x10 ¹²	2x10 ¹⁸	6
1x10 ¹³	4x10 ¹⁸	6

Table 4.1: Simulated depth and peak concentration for buried-channel implants. Depth is given with respect to the germanium surface. P and BF_2 implant energies were adjusted to give the same depth.

The BF₂ and P implants were activated at 400 °C and 500 °C, respectively, for 60 s in N₂. Following channel implant activation, the screen oxide was removed in 50:1 DI:HF and the wafers were then immersed in 6:1 DI:HCl for 5 min. and rinsed in DI water in preparation for gate stack deposition. An *in-situ* ALD WN/3.5 nm Al₂O₃/ 2.5 nm AlN gate stack was then deposited on all wafers [41]. After gate etching, S/D implantation was performed (BF₂: 1×10^{15} cm⁻², 40 keV; P: 1×10^{15} cm⁻², 25 keV), followed by PECVD oxide deposition, S/D activation, and contact patterning. Ti/Al was used for the metallization. S/D anneal conditions for the p- and n-FETs were 450 °C for 30 min. in forming gas and 500 °C for 60 s in N₂, respectively.

4.4.2 Experimental Results

Figure 4.7 shows I_S-V_G characteristics, Figure 4.8 shows inversion-side split C-V charac-

teristics (C_{GSD}), and Figure 4.9 shows effective carrier mobility, measured from the same p-FET or n-FET for each dose.



Figure 4.7: Measured I_S-V_G characteristics for long-channel Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , 4×10^{12} , and 1×10^{13} cm⁻² channel implants. In contrast to p-FETs, n-FETs show decreasing subthreshold swing as implant dose increases from 1×10^{12} to 4×10^{12} cm⁻²



Figure 4.8: Measured inversion-side split C-V characteristics (C_{GSD}) for Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , 4×10^{12} , and 1×10^{13} cm⁻² channel implants. p-FETs show C-V stretch-out associated with transition from surface-channel to buried-channel operation as implant dose increases from 1×10^{12} to 1×10^{13} cm⁻². In contrast, n-FETs show a simple shift of C_{GSD} characteristics for low implant doses (1×10^{12} - 4×10^{12} cm⁻²) with only the highest implant dose (1×10^{13} cm⁻²) showing evidence of buried-channel operation.



Figure 4.9: Effective carrier mobility for Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , 4×10^{12} , and 1×10^{13} cm⁻² channel implants. Peak mobilities of 300 and 600 cm²/Vs were observed for p- and n-FETs, respectively. Inset of (b) shows electron mobility on reduced vertical scale to provide a better view of carrier mobility in Ge n-FETs with 1×10^{12} - 4×10^{12} cm⁻² channel implants. Mobility was extracted using the standard split C-V method for devices that could be sufficiently turned off with gate bias.

The p-FET device characteristics show expected behavior as the devices transition from surface-channel to buried-channel operation with increasing channel implant dose. Figures 4.7a and 4.8a show increasing subthreshold swing and increased C-V stretch-out for higher implant doses. For implant doses up to 4×10^{12} cm⁻², the devices operate at reduced vertical effective field (E_{eff}) while the device that received the 1×10^{13} cm⁻² dose shows buried-channel behavior and significantly reduced capacitance equivalent oxide thickness (CET). Figure 4.9a shows the effective hole mobility as a function of implant dose. As expected, reduced E_{eff} and buried-channel operation results in increased mobility due to reduced interaction with interface states. These devices exhibit peak mobility of 300 cm²/ Vs and 2-3x improvement over Si universal hole mobility. At a dose of 4×10^{12} cm⁻², ionized impurity scattering starts to become significant and mobility improvement saturates. Figures 4.7b and 4.8b show the n-FET I_S-V_G and split C-V characteristics. n-FETs that

received the highest dose $(1 \times 10^{13} \text{ cm}^{-2})$ show clear buried-channel behavior and peak mobility of 600 cm²/Vs (Figure 4.8b). The 1×10^{12} cm⁻² dose devices show a distorted mobility curve with a peak value of 26 cm²/Vs (inset Figure 4.9b). While both n- and p-FETs show substantial mobility improvement for reduced E_{eff} and/or buried-channel operation, the impact on electron mobility is dramatic. This difference is due to the asymmetric distribution of interface states within the bandgap. Since the peak mobilities extracted in this work correspond to carriers in doped buried channels with low vertical effective field, these results are best compared against the bulk conductivity mobility of similarly doped n- and p-Ge. From resistivity data [5], the electron and hole mobility for a given doping concentration can easily be calculated. The buried-channel implants resulted in peak concentrations of 1- $4x10^{18}$ cm⁻³ (Table 4.1), so bulk electron and hole mobility were calculated assuming a doping level of 2×10^{18} cm⁻³. Table 4.2 compares the measured and calculated carrier mobilities. Given the assumptions involved in this comparison, the agreement is reasonable. The buried channel mobility is expected to be lower than the bulk conductivity mobility since the carriers are not completely isolated from the interface. Furthermore, the active doping concentration/profile is not precisely known which adds additional uncertainty to this comparison.

Parameter	Measured Values	Calculated Values (assuming doping of 2x10 ¹⁸ cm ⁻³)
Electron mobility (cm ² /Vs)	600	781
Hole mobility (cm ² /Vs)	300	481
Mobility ratio (μ_e/μ_h)	2.0	1.6

Table 4.2: Comparison of peak electron and hole mobility measured in buriedchannel MOSFETs with bulk conductivity mobility of n- and p-Ge. Bulk conductivity mobility was calculated assuming a doping of $2x10^{18}$ cm⁻³.

Due to process variation, some 1×10^{13} cm⁻² dose n-FETs exhibited different degrees of buried-channel operation as evidenced by differences in peak CET. Figure 4.10 shows effective mobility for 1×10^{13} cm⁻² dose n-FETs with peak CETs of 4.5 and 6 nm (mobility for 1×10^{12} cm⁻² dose device with peak CET of 3 nm is shown for comparison). As expected, devices that are "more buried" show higher CET, higher mobility, and cannot be shutoff with gate bias. For the 6 nm CET device, inversion charge was determined using both split C-V and Hall measurements. At an inversion charge density of 8×10^{12} cm⁻², this device shows an effective mobility of approximately ~350 cm²/Vs. Buried-channel p-FETs show a mobility of ~200 cm²/Vs at similar charge density. The buried-channel hole mobility observed in this work closely agrees with recently published data for Ge p-FETs with 0.5 nm silicon interlayers [34] which suggests that 2-3x enhancement over silicon universal could be achieved for surface-channel devices with a high quality germaniumdielectric interface.



Figure 4.10: Effective mobility for 1×10^{13} cm⁻² dose buried-channel n-FETs with peak CETs of 4.5 and 6 nm. Mobility for 1×10^{12} cm⁻² dose n-FET with peak CET of 3 nm is shown for reference. For n-FET with 6 nm CET, inversion charge was determined using both split C-V and Hall measurements.

4.4.3 Phosphorus Passivation Effect

Interestingly, n-FETs that received lower implant doses $(1 \times 10^{12} - 4 \times 10^{12} \text{ cm}^{-2})$ do not exhibit the signature behavior of reduced E_{eff} and/or buried-channel operation. The subthreshold swing of these devices decreases for doses up to $4 \times 10^{12} \text{ cm}^{-2}$ and the split C-V characteristics exhibit a simple shift rather than the expected stretch-out that was observed for p-FETs. In addition to reduced subthreshold swing, the I_S-V_G characteristics also show less distortion for doses up to $4 \times 10^{12} \text{ cm}^{-2}$. Fig. 4.11 shows CET vs. inversion charge (CET-n_{inv}) for both n- and p-FETs. By definition, reduced E_{eff} is manifested by increased CET for a given amount of inversion charge. As shown in Fig. 4.11, p-FETs clearly show this behavior for each of the implant conditions. In contrast, n-FETs that received 1×10^{12} - $4 \times 10^{12} \text{ cm}^{-2}$ doses show nearly identical CET-n_{inv} characteristics, which implies that they are not operating under reduced E_{eff} or as buried-channel devices.



Figure 4.11: Capacitance equivalent thickness (CET) vs. inversion charge density for Ge p-FETs (a) and n-FETs (b) that received 1×10^{12} , 2×10^{12} , and 4×10^{12} cm⁻² channel implants. p-FETs show evidence of reduced E_{eff} and/or buried-channel operation while n-FETs appear to be surface-channel devices. Plots were obtained by integrating the split C-V characteristics.

Despite this fact, the 2×10^{12} and 4×10^{12} cm⁻² dose devices still show substantially increased mobility accompanied by reduced subthreshold swing. The 4×10^{12} cm⁻² dose device has a peak electron mobility of 150 cm²/Vs and 6x increase at high field relative to the 1×10^{12} cm⁻² dose device. Since this improvement is not obviously explained by reduced E_{eff}, it is possible that phosphorus accumulates at the interface and either passivates interface states or prevents them by altering or suppressing GeO_x [48], [71]. Both lower subthreshold swing and reduced distortion of the I_S-V_G characteristics suggest a reduction in interface state density. Furthermore, the threshold voltage of the 4×10^{12} cm⁻² dose device is shifted closer to the ideal value of +0.5 V which also suggests that interface states have been passivated. This theory is consistent with published diffusion studies of phosphorus-implanted germanium in which out-diffusion and segregation of phosphorus at the germanium-dielectric interface was observed [19]. In addition, plasma PH₃ treatment [48] and AlP_xO_y dielectrics [71] have previously been reported to improve the electrical characteristics of Ge MOS devices.

4.5 Summary

Carrier transport in Ge n-MOSFETs with WN/3.5 nm Al₂O₃/2.5 nm AlN gate stacks was investigated through low temperature device characterization and buried-channel device operation. Low temperature characterization showed significant electron trapping and a high density of interface states within 0.05 eV of the conduction band edge. The density of states near the conduction band edge approached 6×10^{13} cm⁻² eV⁻¹, approximately 5x higher than near the valence band edge. Buried-channel Ge n- and p-FETs were fabricated and peak mobilities of 600 cm^2/Vs and 300 cm^2/Vs were observed, respectively. These values agree reasonably well with the bulk conductivity mobility of similarly doped n- and p-Ge and demonstrate that surface-channel mobility is limited by the quality of the germanium-dielectric interface. In contrast to buried-channel p-FETs, n-FETs that received phosphorus implants ranging from 1×10^{12} - 4×10^{12} cm⁻² did not exhibit reduced E_{eff}/ buried-channel behavior but still showed mobility improvement. Given previously published results, it seems plausible that phosphorus is accumulating at the germanium-dielectric interface and either passivating near-band-edge interface states or preventing interface state formation by altering or suppressing GeO_x formation. These results provide the first direct evidence that phosphorus incorporation at the interface leads to improved mobility and suggest that it is possible to achieve reasonable electron mobility with high-k dielectrics directly on germanium. Furthermore, phosphorus-based passivation techniques may be useful in improving other germanium-dielectric interfaces.

Chapter 5

Conclusions

5.1 Summary

Recent analysis has shown a strong correlation between intrinsic transistor delay, source injection velocity, and low-field carrier mobility. To maintain historical performance trends for ultra-scaled (L_G = 10 nm) CMOS devices, significant improvement in source injection velocity is required. Replacing silicon with a higher mobility semiconductor is one potential solution. Of the high mobility semiconductors, only germanium offers significant improvement in both electron ($\sim 2.4x$) and hole ($\sim 4x$) mobility. In addition, the band structure of germanium results in low transport effective mass and high inversion charge density for electrons on (111)-oriented substrates. High inversion charge density allows (111) Ge n-MOSFETs to match or exceed the performance of III-V MOSFETs for devices with aggressively scaled equivalent oxide thickness (0.5-1.0 nm). However, formation of a high quality germanium-dielectric interface remains a serious challenge that threatens to limit the ultimate performance of these devices. High-k dielectrics (HfO₂, ZrO₂) deposited directly on cleaned germanium result in a poor quality interface due to germanium in-diffusion, interfacial reactions, crystallization of the high-k dielectric, and GeO_x formation at the interface. Inserting an interfacial layer between the high-k dielectric and germanium surface mitigates these problems. To date, germanium oxide has been unable to provide a high quality, thermally stable interface with germanium, so alternative interlayers must be considered. Proposed interlayers include GeON, Si, and metal nitrides such as AlN and Hf₃N₄, with this work focusing on nitride-based interlayers.

A model dielectric stack process consisting of RCA-based surface preparation followed by ALD deposition of an aluminum nitride (AlN) interfacial layer and aluminum oxide (Al_2O_3) capping layer was studied in detail. Despite its limitations (bulk traps and fixed charge), AlN is one of only a few dielectric layers (GeON, AlN, Hf₃N₄, LaAlO₃) that have produced functional Ge n-MOSFETs. In this study, the effects of AlN thickness and PMA conditions on the electrical properties of Al₂O₃/AlN dielectric stacks were investigated. Interface state characterization was performed on Ge capacitors fabricated using an optimized gate stack and post-metal anneal. Optimal C-V characteristics were achieved for an AlN interlayer thickness of 2.5 nm and either a 500 °C 60 s N₂ or 450 °C 30 min. FG anneal. Gate stacks annealed in N₂ were stable up to 500 °C with noticeable degradation observed at 550 °C. Low temperature C-V measurements revealed a decrease in peak capacitance at low temperature and/or high measurement frequency for n-Ge capacitors. This behavior was attributed to electron trapping at the interface, and based on the observed frequency dispersion, the percentage of trapped carriers at high accumulation charge density was estimated to be approximately 15%.

Ge n- and p-MOSFETs were fabricated with GeON, AlN, and Hf₃N₄ interlayers deposited using techniques ranging from thermal nitridation to atomic layer deposition and molecular beam deposition. Surface preparation methods included standard wet cleans (dilute HF, modified RCA clean) as well as *in-situ* GeO_x desorption under UHV conditions. Despite the range of materials and methods investigated, device performance was relatively consistent indicating a degree of commonality in nitride-germanium interfaces. Hole mobility generally matched or exceeded silicon universal hole mobility. In addition, p-FET device characteristics were relatively distortion-free with subthreshold swing less than 100 mV/decade and reasonable C-V hysteresis (~100 mV). In contrast, ALD nitride interlayer n-FETs showed poor mobility (typically 50-100 cm²/Vs), distorted device characteristics, and larger C-V hysteresis (~200 mV). n-FETs based on GeON interlayers were essentially non-functional, probably due to poor thermal stability at the ntype S/D activation temperature used in this work (500 °C). Amongst ALD interlayers, Hf_3N_4 produced higher electron mobility (100 cm²/Vs) while AlN produced higher hole mobility (150 cm²/Vs). Based on these and other published results, a clear path towards high-mobility Ge p-FETs is apparent; however, the same cannot be concluded for Ge n-FETs. To date, Ge n-FET device characteristics have been extremely disappointing and have contributed to renewed interest in III-V n-FETs.

Many theories have been proposed to explain poor Ge n-FET performance. These theories were investigated and it was found that an asymmetric distribution of interface states in the bandgap is the primary cause of low electron mobility in AlN interlayer n-FETs. For germanium-AlN interfaces, the interface state density near the conduction band edge approaches 6×10^{13} cm⁻² eV⁻¹, approximately 5x higher than near the valence band edge.

Carrier transport in Ge n-MOSFETs with WN/3.5 nm Al₂O₃/2.5 nm AlN gate stacks was investigated through low temperature device characterization and buried-channel device operation. Low temperature mobility characterization showed significant electron trapping at the germanium-dielectric interface. To reduce the interaction of carriers with interface states, buried-channel Ge n- and p-FETs were fabricated and peak mobilities of 600 cm²/Vs and 300 cm²/Vs were observed, respectively. These values agree reasonably well with the bulk conductivity mobility of similarly doped n- and p-Ge and demonstrate that surface-channel mobility is limited by the quality of the germanium-dielectric interface. In contrast to buried-channel p-FETs, n-FETs that received phosphorus implants ranging from 1×10^{12} - 4×10^{12} cm⁻² did not exhibit reduced E_{eff} buried-channel behavior but still showed mobility improvement. These experiments provide the first direct evidence that phosphorus incorporation at the interface leads to improved mobility in Ge n-FETs.

In summary, the specific contributions of this work include:

- development and optimization of AIN interlayer MOS gate stacks,
- determination of the interface state distribution for WN/Al₂O₃/AlN gate stacks on germanium,
- performance evaluation of Ge MOSFETs with AlN, Hf₃N₄, and GeON interlayers prepared using atomic layer deposition, rapid thermal nitridation, and molecular beam deposition
- investigation of limiting factors in Ge n-FET surface mobility including an evaluation of carrier transport in surface-channel and buried-channel MOSFETs with WN/ Al₂O₃/AlN gate stacks, and
- discovery of a novel phosphorus-based passivation technique that significantly improves Ge n-FET device performance.

5.2 Future Work

Additional optimization of AlN and Hf₃N₄ interlayer gate stacks may be possible. The process space investigated in this work was limited to RCA-based cleaning followed by ALD deposition and post metal annealing in either N₂ for 60 s or 8:1 N₂:H₂ for 30 min. Each of these steps could be investigated in more detail. Other surface preparation techniques and more complex interlayers (e.g. AlON, HfON, HfAlON) could certainly be explored. As has been observed with silicon [75], interlayers with reduced nitrogen content may improve carrier mobility. Recent evidence also suggests that alternative surface preparations may be superior to the RCA-based cleans used in this work. Surface preparation and interlayer formation are dependent processes that must be co-optimized. It was recently discovered that dielectric stacks deposited on as-received wafers from the supplier (Umicore) exhibit better electrical properties than on RCA-cleaned wafers. Furthermore, for as-received wafers, pure Al2O3 appears to perform better than Al2O3/AlN stacks. Since the supplier's surface preparation technique is proprietary, it is difficult to speculate why this may be the case; however, it seems likely that pre-existing germanium oxide is acting as an interlayer. The thermal stability of this apparent interlayer is unknown. Understanding the conditions under which this layer forms and being able to controllably form this layer prior to high-k deposition are critical first steps. If the thermal stability of this layer is poor, nitrogen or other stabilizing elements (e.g. Si) could be added [75]. This thesis also provides a detailed description of symptoms associated with degraded NMOS mobility so rapid screening of new gate stack processes should be possible. Furthermore, phosphorus-based passivation techniques are a promising new direction for improving Ge n-FET performance.

Appendix A

MOSFET Process Flow

The process flow described in this appendix includes optional process modules for LTO device isolation and buried-channel MOSFETs.

Device Isolation Process Module (optional)

1. RCA Clean Machine: TRL RCA Recipe: see notes Notes: (1) 6:1 DI:HCl 10 min. (2) Rinser 1 (3) 5:1:1 DI:NH₄OH:H₂O₂ 30 s (4) Rinser 1 (5) 6:1 DI:HCl 10 min. (6) Rinser 2 (7) Dry with N_2 gun 2. LTO- Active Area- 130 nm Machine: ICL Tube 6C Recipe: "LTO 400C 53A SPK", ~30.6 Å/min., 43 min. Notes: 3. Photo- Active Area Machine: TRL HMDS, Coater, and EV1 Recipe: HMDS Setting 5, 4 s dispense, 4 KRPM 40 s, Expose 3.0 s, Develop 45 s Notes: 4. Etch- Active Area

- Machine: TRL Acidhood2 Recipe: 3:1 DI:BOE 80 s Notes: Etch rate~ 25 Å/s
- 5. Ash- Active Area Machine: TRL Asher Recipe: 1 hr. 15 min. Notes:

Buried-Channel Implant Module (optional)

6. RCA Clean Machine: TRL RCA Recipe: see notes (1) 6:1 DI:HCl 10 min. Notes: (2) Rinser 1 (3) 5:1:1 DI:NH₄OH:H₂O₂ *10 s* to minimize etching of germanium (4) Rinser 1 (5) 6:1 DI:HCl 10 min. (6) Rinser 2 (7) Dry with N_2 gun 7. LTO- Implant Screen Oxide- 10 nm Machine: ICL Tube 6C Recipe: "LTO-GATE1", ~5.25 Å/min., 21 min. Notes: Wafers loaded in 4" boat between 4" dummy wafers 8. Buried Channel Implant Machine: Innovion Recipe: variable Notes: 9. Post Implant Clean Machine: TRL Acidhood2 Recipe: 5 min. Nanostrip Notes: 10. RCA Clean Machine: TRL RCA Recipe: see below Notes: (1) 6:1 DI:HCl 10 min. (2) Rinser 1 (3) 5:1:1 DI:NH₄OH:H₂O₂ 30 s

- (4) Rinser 1 (5) 6:1 DI:HCl 10 min. (6) Rinser 2
 - (7) Dry with N_2 gun

11. Buried Channel Implant Activation Machine: ICL RTA2 Recipe: N₂ setpoint= 25, Emissivity= 55, TC control Notes: PMOS: 400 °C 60 s (APR400.rcp) NMOS: 500 °C 60 s (APR500.rcp)

Annular MOSFET Process Module

12. RCA Clean For Surface-Channel MOSFETs: Machine: TRL RCA Recipe: see notes Notes: (1) 6:1 DI:HCl 10 min. (2) Rinser 1 (3) 5:1:1 DI:NH₄OH:H₂O₂ 30 s (4) Rinser 1 (5) 6:1 DI:HCl 10 min. (6) Rinser 2 (7) Dry with N₂ gun

> For Buried-Channel MOSFETs: Machine: TRL Acidhood2 Recipe: clean + target 20 nm oxide removal Notes: (1) 5:1:1 DI:NH₄OH:H₂O₂ 30 s (2) Rinse in labware (3) 50 s 50:1 DI:HF, etch rate~ 4.0 Å/s (4) Rinse in labware (5) 6:1 DI:HCl 10 min. (6) Rinse in labware (7) Dry with N₂ gun

13. ALD Gate Stack

Machine: ALD reactor Recipe: WN/Al₂O₃/AlN stack (see below) Notes:

Film: AlN Cycles: 25 Carrier: 10 so Substrate T: Precursor TDMAA NH ₃	ccm 200 °C Precursor T 110 °C N/A	Pulse (s) 0.2 0.1	Expo (s) 0 0.1	Pump (s) 8 12
Film: Al_2O_3 Cycles: 35 Carrier: 10 set	ccm			
Substrate T:	200 °C			
Precursor TDMAA	Precursor I	Pulse (s) 0.2	Expo (s)	Pump (s)
H ₂ O	N/A	0.015	0	8

Film: WN				
Cycles: 1000				
Carrier: 10 scc	m			
Substrate T: 37	′5 ℃			
Precursor	Precursor T	Pulse (s)	Expo (s)	Pump (s)
N ₂ ASSIST	N/A	0.015	0	1
W	90 °C	0.1	0	8
NH ₂	N/Δ	0.2	0.2	12

14. Photo- Gate

Machine: TRL HMDS, Coater, and EV1 Recipe: HMDS Setting 5, 4 s dispense, 4 KRPM 40 s, Expose 3.0 s, Develop 45 s Notes:

15. Etch-Gate

Machine: TRL Plasmaquest

Recipe: TANETCH.rcp, 60 s (see below)

Notes: (A) Before etching, run ETCHCLN.rcp for 10 min., TANETCH for 10 min. (B) Reflected power: 22-24 W

(C) Wafers clear after 40-50 s of etching so 10-20 s of overetch

Etch recipes: TANETCH.RCP

	Step 1	Step 2
CF ₄ (sccm)	40	40
ECR (W)	0	300
RF (W)	0	25
Pressure (mT)	15	15
Time (s)	30	VAR
ETCHCLN.RCP		
	Step 1	Step 2
O_2 (sccm)	15	15
He (sccm)	15	15
CF_4 (sccm)	40	40
ECR (W)	0	300
RF (W)	0	50
Pressure (mT)	15	15
Time (s)	30	VAR

16. S/D Implant

Machine: Innovion

Recipe: see below

Notes: PMOS= BF_{2} ,40 keV,1x10¹⁵ cm⁻², 7 deg.

NMOS= P, 25 keV, 1×10^{15} cm⁻², 7 deg.

17. Post Implant Clean Machine: TRL Photowet Recipe: IPA rinse Notes: 18. Ash- Gate Machine: TRL Asher Recipe: 1 hr. 15 min. Notes: 19. PECVD SiO₂- Interlevel Dielectric- 200 nm Machine: TRL STSCVD Recipe: HFSIO, 4 min. 20 s Notes: Recipe: Pressure (mT): 900 Platen T (°C): 300 Showerhead T (°C): 250 N_2O (sccm): 1420 N₂ (sccm): 392 SIH₄ (sccm): 10 RF: 30 W, 13.56 Mhz on showerhead only 20. S/D Implant Activation Machine: TRL RTA35, Tube B1 Recipe: N_2 setpoint= 30, Emissivity= 55, TC control Notes: PMOS: 450 °C 8:1 N₂:H₂ 30 min. NMOS: 500 °C N₂ 60 s (GE500.RCP with Ge TC wafer) 21. Photo-Via Machine: TRL HMDS, Coater, and EV1 Recipe: HMDS Setting 5, 4 s dispense, 4 KRPM 40 s, Expose 3.0 s, Develop 45 s Notes: 22. Etch- Via Machine: TRL Acidhood Recipe: 3:1 DI:BOE, 60 s Notes: Etch rate~ 50 Å/s 23. Etch- High-k Dielectric Machine: TRL Acidhood Recipe: 4:1 DI:HCL 2 min. Notes:

24. Ash- Via Machine: TRL Asher Recipe: 1 hr. 15 min. Notes:

25. Pre-metal Clean Machine: TRL Acidhood Recipe: 50:1 DI:HF 15s Notes:

26. Metallization- Frontside and Backside Machine: TRL Perkin Elmer Recipe: see below Notes: Frontside= 50 nm Ti/ 400 nm Al Backside= 50 nm Ti/ 2μm Al

27. Photo- Metal

Machine: TRL HMDS, Coater, and EV1 Recipe: HMDS Setting 5, 4 s dispense, 4 KRPM 40 s, Expose 3.0 s, Develop 45 s Notes:

28. Etch- Metal Machine: TRL Acidhood Recipe: Al Etchant 50 s followed by Ti Etchant 90s Notes: Al Etchant: 55 °C Ti Etchant: 1000:15 DI:BOE

29. Ash- Metal Machine: TRL Asher Recipe: 1 hr. 15 min. Notes:

Appendix B

Agilent 4294A IBASIC C-V Measurement Program

The IBASIC program listed below performs forward and reverse C-V sweeps at the frequency specified in line 50. The sweep start voltage is held for 1 s before starting the forward and reverse sweeps. The DC bias sweep range is specified in line 210.

```
10 DIM Buff$[9],Command$[9]
20 REAL Freq(1:4)
30 ASSIGN @Agt4294 TO 800
40!
50Freq(1)=50000
60!
70 PRINT "Input file name (without Extension)"
80 INPUT "Name?", Inp char$
90 File$=UPC$(Inp char$)
100 !
110 PRINT "Input measurement number"
120 INPUT "Number?", Inp char2$
130 Num$=UPC$(Inp char2$)
140!
150 File1$=File$&Num$
160!
170 OUTPUT @Agt4294;";MEAS CPD"
180 OUTPUT @Agt4294;";SPLD OFF"
190 OUTPUT @Agt4294;";ACCUD OFF"
200 OUTPUT @Agt4294;";ACCUD ON"
210 OUTPUT @Agt4294;";STAR -1;STOP 1"
220 ! Set hold time before sweep to 1s
230 OUTPUT @Agt4294;";SDELT 1"
240 OUTPUT @Agt4294;";TRGS INT"
250!
260 ! Start Measurement
270!
280 OUTPUT @Agt4294;";CWFREQ ";Freq(1)
290 OUTPUT @Agt4294;";SWED UP"
300 OUTPUT @Agt4294;";DCO ON"
310 OUTPUT @Agt4294;";SING"
320!
```

330 ! Wait for measurement to finish 340 PRINT "Waiting..." 350 OUTPUT @Agt4294;"*OPC?" 360 ENTER @Agt4294;Buff\$ 370 PRINT "Sweep complete" 380 ! 390 ! Write data to memory 400 OUTPUT @Agt4294;";DATMEM" 410 ! 420 ! Prepare for next sweep 430 OUTPUT @Agt4294;";SWED DOWN" 440 OUTPUT @Agt4294;";SING" 450 ! 460 ! Wait for measurement to finish 470 PRINT "Waiting ... " 480 OUTPUT @Agt4294;"*OPC?" 490 ENTER @Agt4294;Buff\$ 500 PRINT "Sweep complete" 510! 520 OUTPUT @Agt4294;";DCO OFF" 530 OUTPUT @Agt4294;";STOD DISK" 540 OUTPUT @Agt4294;";SAVDTRC ON" 550 OUTPUT @Agt4294;";SAVCAL OFF" 560 OUTPUT @Agt4294;";SAVDAT OFF" 570 OUTPUT @Agt4294;";SAVMEM OFF" 580 OUTPUT @Agt4294;";SAVMTRC ON" 590 Command\$="SAVDASC" 600 OUTPUT @Agt4294;"*CLS" 610OUTPUT @Agt4294;Command\$&" """&File1\$&"""" 620 ! 630 !END

Appendix C

C-V Characteristics of Al/Al₂O₃/AlN/Ge Capacitors

Experimental Description

Capacitors were fabricated on (100) Ga-doped p-type substrates with resistivity of 0.12-0.17 Ω -cm and (100) Sb-doped n-type substrates with resistivity of 0.13-0.16 Ω -cm. The wafers were cleaned using the modified RCA clean described in Appendix A. The dielectric stacks shown in Table C.1 were deposited on both n- and p-Ge followed by *ex-situ* Al sputtering (200 nm) on the frontside of the wafer and Ti/Au sputtering (50 nm/ 200 nm) on the backside of the wafer. The aluminum gate electrodes were patterned using photolithography and wet etching. Capacitors were then annealed as shown in Table C.2. In this document, Al₂O₃/AlN dielectric stacks are denoted by the number of Al₂O₃ cycles followed by the number of AlN cycles (e.g. 50/20).

ALD Run No.	Al ₂ O ₃ cycles/AlN cycles
41	50/20
42	50/30
43	100/30
44	50/10

Table C.1: Dielectric stack splits for Al/Al₂O₃/AlN/Ge capacitors. Stacks are defined by number of Al₂O₃ cycles and number of AlN cycles (Al₂O₃ cycles/ AlN cycles)

Temperature (°C)	Time (min.)	Ambient
no anneal	-	-
350	30	FG
350	30	N ₂
400	1	N ₂
500	1	N ₂

Table C.2: Post-metal anneal splits for Al/Al₂O₃/AlN/Ge capacitors. FG= 8:1 N₂:H₂.

Electrical Results

Using these capacitors, the impact of AlN thickness was examined. Figure C.1 shows C-V characteristics for 50/10, 50/20, and 50/30 Al_2O_3/AlN stacks on p-Ge at 100 kHz and 1 MHz. The left column shows as-deposited characteristics while the right column shows the characteristics after 350 °C 30 min. forming gas (FG) annealing. In this work, the forming gas composition was 8:1 N₂:H₂. The forward sweep is from depletion to accumulation, and vice versa for the reverse sweep.



Figure C.1: C-V characteristics for (A) 50/10, (B) 50/20, and (C) 50/30 Al_2O_3/AlN stacks on p-Ge at 100 kHz and 1MHz. Left column shows as-deposited characteristics while right column shows characteristics after 350 °C 30 min. FG annealing.

Other than variation in peak capacitance due to varying AlN thickness, the as-deposited samples all exhibit similar behavior. The capacitance due to midgap interface states (C_{it}) features prominently in the as-deposited samples. For all three AlN interlayer thicknesses, post-metal annealing reduces interface state capacitance. Increasing AlN thickness, particularly from 10 to 20 cycles (approximately 1 nm to 2 nm), also reduces C_{it} ; however, less improvement is seen for the 30 cycle AlN interlayer. Like Si_3N_4 , AlN contains bulk traps that lead to C-V hysteresis [43]. As seen in Figure C.1, thicker AlN interlayers result in larger hysteresis in the PMA samples. Therefore, a trade-off between reduced C_{it} and increased hysteresis determines the optimal AlN thickness. From the data in Figure C.1, the optimal AlN thickness is 20 cycles or equivalently 2 nm.

The corresponding data for n-Ge is shown in Figure C.2. The trends are qualitatively similar to p-Ge; however, there are a few differences. The n-Ge C-V characteristics show more distortion near accumulation and also exhibit decreasing hysteresis during the reverse sweep. Both of these observations suggest a high density of interface states near the conduction band edge. Like p-Ge, the optimal AlN thickness for n-Ge is approximately 2 nm.

In addition to the 350 °C FG anneal, n and p-Ge capacitors were also annealed at 350 °C in N₂ (Table C.2). Figure C.3 shows the C-V characteristics for 50/20 n-Ge capacitors after nitrogen and forming gas annealing. The effect of anneal ambient on C-V characteristics was minimal. From this data, and the data presented in Chapter 2, it does not appear that hydrogen passivation plays the same role at germanium-dielectric interfaces that it does for Si-SiO₂ interfaces.



Figure C.2: C-V characteristics for (A) 50/10, (B) 50/20, and (C) 50/30 Al_2O_3/AlN stacks on n-Ge at 100 kHz and 1MHz. Left column shows as-deposited characteristics while right column shows characteristics after 350 °C 30 min. FG annealing.



Figure C.3: C-V characteristics of 50/20 n-Ge capacitors after (A) nitrogen and (B) forming gas annealing at 350 °C for 30 min.

Appendix D

C-V Characteristics of WN/Al₂O₃/AlN/Ge Capacitors

Experimental Description

Capacitors were fabricated on (100) Ga-doped p-type substrates with resistivity of 0.12 - 0.17 Ω -cm and (100) Sb-doped n-type substrates with resistivity of 0.13 to 0.16 Ω -cm. The wafers were cleaned using the modified RCA clean described in Appendix A. The WN/(Al₂O₃,AIN) gate stacks shown in Table D.1 were deposited on both n- and p-Ge followed by *ex-situ* Al sputtering (250 nm) on the frontside of the wafer and Ti/Al sputtering (20 nm/ 1 µm) on the backside of the wafer. The aluminum metallization was patterned using photolithography and wet etching. Using the same resist mask, the WN gate electrode was etched using CF₄ RIE. The capacitors were then annealed as shown in Table D.2. The "as-deposited" samples did not receive a post-metal anneal but did receive the thermal budget associated with WN deposition (approximately 350 °C for 6 hrs. in N₂).

ALD Run No.	Al ₂ O ₃ cycles/AlN cycles	
104	45/15	
105	35/25	
111	60/00	
113	00/60	

Table D.1: Dielectric stack splits for WN/(Al₂O₃, AlN)/Ge capacitors. Stacks are defined by number of Al₂O₃ cycles and number of AlN cycles (Al₂O₃ cycles/ AlN cycles).

Temperature (°C)	Time (min.)	Ambient
as-deposited	-	-
350	30	FG
400	1	N ₂
450	30	FG
450	1	N ₂
500	30	FG
500	1	N ₂
550	30	FG
550	1	N ₂

Table D.2: Post-metal anneal splits for WN/Al2O3/AlN/Ge capacitors. FG= 8:1N2:H2.

Impact of Post-Metal Annealing on C-V Characteristics

Post-metal annealing is a critical step for high-k dielectrics, particularly ALD films deposited at low temperature (200 °C), that is used to improve the electrical properties of asdeposited gate stacks. However, degradation of germanium-dielectric interfaces has also been observed during subsequent thermal processing. The purpose of this experiment was to determine the optimal anneal and also investigate the thermal stability of these gate stacks at n-type S/D activation temperatures (500- 550 °C).

WN/45 cycle Al₂O₃/15 cycle AlN/Ge (45/15) and WN/35 cycle Al₂O₃/25 cycle AlN/ Ge (35/25) capacitors were subjected to the post-metal annealing conditions listed in Table D.2. Figures D.1 and D.2 show the effect of 450 °C and 550 °C annealing in both N₂ and forming gas (FG) for p-Ge capacitors with 45/15 (Figure D.1) and 35/25 (Figure D.2) gate stacks. The corresponding data for 45/15 and 35/25 n-Ge capacitors is shown in Figures D.3 and D.4.



Figure D.1: Impact of PMA on C-V characteristics of WN/45 cycle $Al_2O_3/15$ cycle AlN/ p-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C samples are shown. Left column shows characteristics from 60 s N₂ anneal while right column shows 30 min. FG anneal.



Figure D.2: Impact of PMA on C-V characteristics of WN/35 cycle $Al_2O_3/25$ cycle AlN/ p-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C samples are shown. Left column shows characteristics from 60 s N₂ anneal while right column shows 30 min. FG anneal.



Figure D.3: Impact of PMA on C-V characteristics of WN/45 cycle $Al_2O_3/15$ cycle AlN/ n-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C samples are shown. Left column shows characteristics from 60 s N₂ anneal while right column shows 30 min. FG anneal.



Figure D.4: Impact of PMA on C-V characteristics of WN/35 cycle $Al_2O_3/25$ cycle AlN/ n-Ge capacitors. From top to bottom, characteristics from as-deposited, 450 °C, and 550 °C samples are shown. Left column shows characteristics from 60 s N₂ anneal while right column shows 30 min. FG anneal.

Appendix E

C-V Characteristics of WN/GdScO₃/Hf₃N₄/Ge Capacitors

Experimental Description

In addition to AlN, Hf₃N₄ interlayers were also explored through a collaboration with Harvard University [41]. ALD Hf₃N₄ layers were deposited using tetrakis-(ethylmethylamido)hafnium (TEMA-Hf) and NH₃ as reactants. The TEMA-Hf bubbler temperature was 80 °C and the substrate temperature was 200- 250 °C. ALD GdScO₃ was used as the dielectric capping layer and ALD WN was used as the gate electrode, both deposited *insitu*. In general, MOS capacitors with Hf₃N₄ behaved similarly to those with AlN. Figure E.1 shows measured quasistatic and high frequency C-V characteristics for a WN/8 nm GdScO₃/1.5 nm Hf₃N₄ /p-Ge capacitor [41]. Interface state density was extracted using both the quasistatic and conductance methods. Figure E.2 shows the resulting D_{it} distribution [41]. Midgap D_{it} for this gate stack was approximately 1×10^{12} cm⁻² eV⁻¹ which is slightly lower than the value obtained for AlN interfacial layers.



Figure E.1: Quasistatic, 10 kHz, and 1 MHz C-V characteristics for WN/8 nm $GdScO_3/$ 1.5 nm Hf_3N_4 /p-Ge capacitor (from [41]).



Figure E.2: D_{it} extracted using both the conductance and quasistatic methods [59] for a WN/8 nm GdScO₃/1.4 nm Hf₃N₄/p-Ge capacitor (from [41]).
References

- [1] International Technology Roadmap for Semiconductors, 2005 Ed., Semiconductor Industry Association, 2005 (http://www.itrs.org).
- [2] D.A. Antoniadis, I. Aberg, C. Ni Chleirigh, O.M. Nayfeh, A. Khakifirooz, and J.L. Hoyt, "Continuous MOSFET performance increase with device scaling: the role of strain and channel material innovations," *IBM J. Res. Develop.*, vol. 50, no. 4/5, pp. 363-376, Jul./Sept. 2006.
- [3] M. Lundstrom, "Elementary scattering theory of the Si MOSFET," *IEEE Electron Device Lett.*, vol. 18, no. 7, pp. 361-363, Jul. 1997.
- [4] A. Khakifirooz and D.A. Antoniadis, "Transistor performance scaling: the role of virtual source velocity and its mobility dependence," in *IEDM Tech. Dig.*, 2006, pp. 667-670.
- [5] S. Sze, *Physics of Semiconductor Devices, 2nd Ed.*, New York: Wiley, 1981.
- [6] M. Lundstrom, *Fundamentals of Carrier Transport, 2nd Ed.*, Cambridge, UK: Cambridge University Press, 2000.
- [7] S. Takagi and A. Toriumi, "Quantitative understanding of inversion-layer capacitance in Si MOSFET's," *IEEE Trans. Electron Devices*, vol. 42, no. 12, pp. 2125-2130, Dec. 1995.
- [8] A. Rahman, M.S. Lundstrom, and A.W. Ghosh, "Generalized effective mass approach for cubic semiconductor n-MOSFETs on arbitrarily oriented wafers," J. Appl. Phys., vol. 97, no. 5, pp. 53702-1-12, Mar. 2005.
- [9] A. Khakifirooz, Private Communication, Feb. 2007.
- [10] M.V. Fischetti and S.E. Laux, "Band structure, deformation potentials, and carrier mobility in strained Si, Ge, and SiGe alloys," J. Appl. Phys., vol. 80, no. 4, pp. 2234-2252, Aug. 1996.
- [11] M.V. Fischetti, "Monte carlo simulation of transport in technologically significant semiconductors of the diamond and zinc-blende structures- part I: homogeneous transport," *IEEE Trans. Electron Devices*, vol. 38, no. 3, pp. 634-649, Mar. 1991.
- [12] S. Takagi and S. Sugahara, "Comparative study on influence of subband structures on electrical characteristics of III-V semiconductor, Ge and Si channel n-MISFETs," in SSDM Tech. Dig., 2006, pp. 1056-1057.
- [13] A. Rahman, G. Klimeck, and M. Lundstrom, "Novel channel materials for ballistic nanoscale MOSFETs- bandstructure effects," in *IEDM Tech. Dig.*, 2005, pp. 615-618.
- P. Bai, C. Auth, S. Balakrishnan, M. Bost, R. Brain, V. Chikarmane, R. Heussner, M. Hussein, J. Hwang, D. Ingerly, R. James, J. Jeong, C. Kenyon, E. Lee, S.-H. Lee, N. Lindert, M. Liu, Z. Ma, T. Marieb, A. Murthy, R. Nagisetty, S. Natarajan, J. Neirynck, A. Ott, C. Parker, J. Sebastian, R. Shaheed, S. Sivakumar, J. Steigerwald, S. Tyagi, C. Weber, B. Woolery, A. Yeoh, K. Zhang, and M. Bohr, "A 65nm logic technology featuring 35nm gate lengths, enhanced channel strain, 8 Cu interconnect layers, low-

k ILD and 0.57 µm² SRAM cell," in IEDM Tech. Dig., 2004, pp. 657-660.

- [15] Y. Taur and T.H. Ning, *Fundamentals of Modern VLSI Devices*. Cambridge, UK: Cambridge University Press, 1998.
- [16] T. Krishnamohan, D. Kim, C. Jungemann, Y. Nishi, and K.C. Saraswat, "Strained-Si, relaxed-Gc or strained-(Si)Ge for future nanoscale p-MOSFETs?," in VLSI Symp. Tech. Dig., 2006.
- [17] A. Pethe, T. Krishnamohan, D. Kim, S. Oh, H.-S. Wong, Y. Nishi, and K.C. Saraswat, "Investigation of the performance limits of III-V double-gate n-MOSFETs," in *IEDM Tech. Dig.*, 2005, pp. 619-622.
- [18] A. Satta, E. Simoen, T. Janssens, T. Clarysse, B. De Jaeger, A. Benedetti, I. Hoflijk, B. Brijs, M. Meuris, and W. Vandervorst, "Shallow junction ion implantation in Ge and associated defect control," *J. Electrochem. Soc.*, vol. 153, no. 3, pp. G229-G233, Mar. 2006.
- [19] A. Satta, T. Janssens, T. Clarysse, E. Simoen, M. Meuris, A. Benedetti, I. Hoflijk, B. De Jaeger, C. Demeurisse, and W. Vandervorst, "P implantation doping of Ge: diffusion, activation, and recrystallization," *J. Vac. Sci. Technol. B*, vol. 24, no. 1, pp. 494-498, Jan./Feb. 2006.
- [20] O.J. Gregory, L.A. Pruitt, E.E. Crisman, C. Roberts, and P.J. Stiles, "Native oxides formed on single-crystal germanium by wet chemical reactions," *J. Electrochem. Soc.*, vol. 135, no. 4, pp. 923-929, Apr. 1988.
- [21] K. Prabhakaran and T. Ogino, "Oxidation of Ge(100) and Ge(111) surfaces: an UPS and XPS study," *Surf. Sci.*, vol. 325, pp. 263-271, Mar. 1995.
- [22] W.A. Albers, E.W. Valyocsik, and P.V. Mohan, "Tetragonal germanium dioxide layers on germanium," J. Electrochem. Soc., vol. 113, no. 2, pp. 196-198, Feb. 1966.
- [23] E.E. Crisman, J.L. Lee, P.J. Stiles, and O.J. Gregory, "Characterisation of n-channel germanium MOSFET with gate insulator formed by high-pressure thermal oxidation," *Electron. Lett.*, vol. 23, no. 1, pp. 8-10, Jan. 1987.
- [24] D. Bodlaki, H. Yamamoto, D.H. Waldeck, and E. Borguet, "Ambient stability of chemically passivated germanium interfaces," *Surf. Sci.*, vol. 543, pp. 63-74, Oct. 2003.
- [25] B. Onsia, T. Conard, S. De Gendt, M. Heyns, I. Hoflijk, P. Mertens, M. Meuris, G. Raskin, S. Sioncke, I. Teerlinck, A. Theuwis, J. Van Steenbergen, and C. Vinckier, "A study of the influence of typical wet chemical treatments on the germanium wafer surface," *Diffus. Defect Data B, Solid State Phenom.*, vols. 103-104, pp. 27-30, Apr. 2005.
- [26] Y. Kamata, Y. Kamimuta, T. Ino, and A. Nishiyama, "Direct comparison of ZrO₂ and HfO₂ on Ge substrate in terms of the realization of ultrathin high-k gate stacks," *Jpn. J. Appl. Phys.*, vol. 44, no. 4B, pp. 2323-2329, Apr. 2005.
- [27] A. Delabie, R.L. Puurunen, B. Brijs, M. Caymax, T. Conard, B. Onsia, O. Richard, W. Vandervorst, C. Zhao, M.M. Heyns, M. Meuris, M.M. Viitanen, H.H.

Brongersma, M. de Ridder, L.V. Goncharova, E. Garfunkel, T. Gustafsson, and W. Tsai, "Atomic layer deposition of hafnium oxide on germanium substrates," *J. Appl. Phys.*, vol. 97, pp. 064104-1-10, Mar. 2005.

- [28] S. Van Elshocht, B. Brijs, M. Caymax, T. Conard, T. Chiarella, S. De Gendt, B. De Jaeger, S. Kubicek, M. Meuris, B. Onsia, O. Richard, I. Teerlinck, J. Van Steenbergen, C. Zhao, and M. Heyns, "Deposition of HfO₂ on germanium and the impact of surface treatments," *Appl. Phys. Lett.*, vol. 85, no. 17, pp. 3824-3826, Oct. 2004.
- [29] H. Shang, M.M. Frank, E.P. Gusev, J.O. Chu, S.W. Bedell, K.W. Guarini, and M. Ieong, "Germanium channel MOSFETs: opportunities and challenges," *IBM J. Res. Develop.*, vol. 50, no. 4/5, pp. 377-386, Jul./Sept. 2006.
- [30] Y. Kamata, Y. Kamimuta, T. Ino, R. Iijima, M. Koyama, and A. Nishiyama, "Dramatic improvement of Ge p-MOSFET characteristics realized by amorphous Zrsilicate/Ge gate stack with excellent structural stability through process temperatures," in *IEDM Tech. Dig.*, 2005.
- [31] M.V. Fischetti, D.A. Neumayer, and E.A. Cartier, "Effective electron mobility in Si inversion layers in metal-oxide-semiconductor systems with a high-k insulator: the role of remote phonon scattering," *J. Appl. Phys.*, vol. 90, no. 9, pp. 4587-4608, Nov. 2001.
- [32] G.S. Lujan, W. Magnus, L.- A. Ragnarsson, S. Kubicek, S. De Gendt, M. Heyns, K. De Meyer, "Modelling mobility degradation due to remote Coulomb scattering from dielectric charges and its impact on MOS device performance," *Microelectron. Reliab.*, vol. 45, no. 5/6, pp. 794-797, May. 2005.
- [33] W.P. Bai, N. Lu, J. Liu, A. Ramirez, D.L. Kwong, D. Wristers, A. Ritenour, M.L. Lee, and D. Antoniadis, "Ge MOS characteristics with CVD HfO₂ gate dielectrics and TaN gate electrode," in VLSI Symp. Tech. Dig., 2003, pp. 121-122.
- [34] P. Zimmerman, G. Nicholas, B. De Jaeger, B. Kaczer, A. Stesmans, L.-A. Ragnarsson, D.P. Brunco, F.E. Leys, M. Caymax, G. Winderickx, K. Opsomer, M. Meuris, and M.M. Heyns, "High performance Ge pMOS devices using a Si-compatible process flow," in *IEDM Tech. Dig.*, 2006, pp. 655-658.
- [35] D.J. Hymes and J.J. Rosenberg, "Growth and materials characterization of native germanium oxynitride thin films on germanium," J. Electrochem. Soc., vol. 135, no. 4, pp. 961-965, Apr. 1988.
- [36] F. Gao, S.J. Lee, J.S. Pan, L.J. Tang, D.-L. Kwong, "Surface passivation using ultrathin AlN_x film for Ge-metal-oxide-semiconductor devices with hafnium oxide gate dielectric," *Appl. Phys. Lett.*, vol. 86, pp. 113501-3, Mar. 2005.
- [37] J.J. Rosenberg and S.C. Martin, "Self-aligned germanium MOSFET's using a nitrided native oxide gate insulator," *IEEE Electron Device Lett.*, vol. 9, no. 12, pp. 639-640, Dec. 1998.
- [38] C.O. Chui, F. Ito, K. Saraswat, "Nanoscale germanium MOS dielectrics- part I: germanium oxynitrides," *IEEE Trans. Electron Devices*, vol. 53, no. 7, pp. 1501-

1508, Jul. 2006.

- [39] N. Wu, Q. Zhang, C. Zhu, C.C. Yeo, S.J. Whang, D.S.H. Chan, M.F. Li, B.J. Cho, A. Chin, D.-L. Kwong, A.Y. Du, C.H. Tung, and N. Balasubramanian, "Effect of surface NH₃ anneal on the physical and electrical properties of HfO₂ films on Gc substrate," *Appl. Phys. Lett.*, vol. 84, no.19, pp. 3741-3743, May 2004.
- [40] H. Shang, H. Okorn-Schimdt, J. Ott, P. Kozlowski, S. Steen, E.C. Jones, H.-S.P.Wong, and W. Haensch, "Electrical characterization of germanium p-channel MOSFETs," *IEEE Electron Device Lett.*, vol. 24, no. 4, pp. 242-244, Apr. 2003.
- [41] K. H. Kim, R. Gordon, A. Ritenour, and D.A. Antoniadis, "Atomic layer deposition of insulating nitride interfacial layers for germanium metal-oxide-semiconductor field effect transistors with high-permittivity oxide/ tungsten nitride gate stacks," accepted *Appl. Phys. Lett.*, Mar. 2007.
- [42] M. Xu, S. Wang, G. Yin, J. Li, Y. Zheng, L. Chen, and Y. Jiab, "Optical properties of cubic Ti₃N₄, Zr₃N₄, and Hf₃N₄," *Appl. Phys. Lett.*, vol. 89, pp. 151908-1-3, Oct. 2006.
- [43] C.H. Lai, A. Chin, B.F. Hung, C.F. Cheng, W.J. Yoo, M.F. Li, C. Zhu, S.P. McAlister, and D.-L. Kwong, "A novel program-erasable high-k AlN-Si MIS capacitor," *IEEE Electron Device Lett.*, vol. 26, no. 3, pp. 148-150, Mar. 2005.
- [44] N. Wu, Q. Zhang, C. Zhu, C. Shen, M.F. Li, D.S.H. Chan, and N. Balasubramanian, "BTI and charge trapping in germanium p- and n-MOSFETs with CVD HfO₂ gate dielectric," in *IEDM Tech. Dig.*, 2005, pp. 563-566.
- [45] M.L. Lee and E.A. Fitzgerald, "Optimized strained Si/ strained Ge dual-channel heterostructures for high mobility p- and n-MOSFETs," in *IEDM Tech. Dig.*, 2003, pp. 429-432.
- [46] G.W. Anderson, M.C. Hanf, P.R. Norton, Z.H. Lu, and M.J. Graham, "The Spassivation of Ge(100)-(1x1)," *Appl. Phys. Lett.*, vol. 66, no. 9, pp. 1123-1125, Feb. 1995.
- [47] M.M. Frank, S.J. Koester, M. Copel, J.A. Ott, V.K. Paruchuri, H. Shang, and R. Loesing, "Hafnium oxide gate dielectrics on sulfur-passivated germanium," *Appl. Phys. Lett.*, vol. 89, pp. 112905-1-3, Sept. 2006.
- [48] S.J. Whang, S.J. Lee, F. Gao, N. Wu, C.X. Zhu, J.S. Pan, L.J. Tang, and D.L. Kwong, "Germanium p- and n-MOSFETs fabricated with novel surface passivation (plasma-PH₃ and thin AlN) and TaN/HfO₂ gate stack," in *IEDM Tech. Dig.*, 2004, pp. 307-310.
- [49] C.O. Chui, H. Kim, D. Chi, B.B. Triplett, P.C. McIntyre, and K.C. Saraswat, "A sub-400°C germanium MOSFET technology with high-k dielectric and metal gate," in *IEDM Tech. Dig.*, 2002, pp. 437-440.
- [50] H. Shang, K.-L. Lee, P. Kozlowski, C. D'Emic, I. Babich, E. Sikorski, M. Ieong, H.-S. P. Wong, K. Guarini, and W. Haensch, "Self-aligned n-channel germanium MOSFETs with a thin Ge oxynitride gate dielectric and tungsten gate," *IEEE*

Electron Device Lett., vol. 25, no. 3, pp. 135-147, Mar. 2004.

- [51] D.S. Yu, A. Chin, C.C. Liao, C.F. Lee, C.F. Cheng, M.F. Li, W.J. Yoo, and S.P. McAlister, "Three-dimensional metal gate-high-k-GOI CMOSFETs on 1-Poly-6-Metal 0.18-µm Si Devices," *IEEE Electron Device Lett.*, vol. 26, no. 2, pp. 118-120, Feb. 2005.
- [52] N. Wu, Q. Zhang, D.S.H. Chan, N. Balasubramanian, and C. Zhu, "Gate-first germanium nMOSFET with CVD HfO₂ gate dielectric and silicon surface passivation," *IEEE Electron Device Lett.*, vol. 27, no. 6, pp. 479-481, Jun. 2006.
- [53] L. Niinisto, J. Paivasaari, J. Niinisto, M. Putkonen, and M. Nieminen, "Advanced electronic and optoelectronic materials by atomic layer deposition: an overview with special emphasis on recent progress in processing of high-k dielectrics and other oxide materials," *Phys. Stat. Sol. A*, vol. 201, no. 7, pp. 1443-1452, Jul. 2004.
- [54] R.L. Puurunen, "Surface chemistry of atomic layer deposition: a case study for the trimethylaluminum/water process," J. Appl. Phys., vol. 97, no. 4, pp. 121301-1-52, Jun. 2005.
- [55] Savannah 100 & 200 Atomic Layer Deposition System User Manual, Cambridge, MA: Cambridge NanoTech Inc., 2006.
- [56] D.M. Hoffman, S.P. Rangarajan, S.D. Athavale, D.J. Economou, J.-R. Liu, Z. Zheng, and W.-K. Chu, "Chemical vapor deposition of aluminum and gallium nitride thin films from metalorganic precursors," *J. Vac. Sci. Technol. A*, vol. 14, no. 2, pp. 306-311, Mar./Apr. 1996.
- [57] J.S. Becker, S. Suh, S. Wang, and R.G. Gordon, "Highly conformal thin films of tungsten nitride prepared by atomic layer deposition from a novel precursor," *Chem. Mater.*, vol. 15, no. 15, pp. 2969-2979, Jul. 2003.
- [58] C.G. Van de Walle, "Theory of hydrogen-related levels in semiconductors and oxides," in *IEDM Tech. Dig.*, 2005, pp. 411-414.
- [59] D.K. Schroder, Semiconductor Material and Device Characterization, 2nd Ed., New York: Wiley, 1998.
- [60] P.V. Gray and D.M. Brown, "Density of SiO₂-Si interface states," Appl. Phys. Lett., vol. 8, no. 2, pp. 31-33, Jan. 1966.
- [61] A. Ritenour, S. Yu, M.L. Lee, N. Lu, W. Bai, A. Pitera, E.A. Fitzgerald, D.L. Kwong, and D.A. Antoniadis, "Epitaxial strained germanium p-MOSFETs with HfO₂ gate dielectric and TaN gate electrode," in *IEDM Tech. Dig.*, 2003, pp. 433-436.
- [62] A. Ritenour, A. Khakifirooz, D.A. Antoniadis, R.Z. Lei, W. Tsai, A. Dimoulas, G. Mavrou, and Y. Panayiotatos, "Subnanometer-equivalent-oxide-thickness germanium p-metal-oxide-semiconductor field effect transistors fabricated using molecular-beam-deposited high-k/metal gate stack," *Appl. Phys. Lett.*, vol. 88, pp. 132107-1-3, Mar. 2006.
- [63] K. Kim, Private Communication, 2007.

- [64] E.A. Fitzgerald, "Dislocations in strained-layer epitaxy: theory, experiment, and applications," *Material Science Reports*, vol. 7, no. 3, pp. 87-142, Nov. 1991.
- [65] A. Dimoulas, G. Mavrou, G. Vellianitis, E.K. Evangelou, N. Boukos, M. Houssa, and M. Caymax, "HfO₂ high-k gate dielectrics on Ge (100) by atomic oxygen beam deposition," *Appl. Phys. Lett.*, vol. 86, pp. 032908-1-3, Jan. 2005.
- [66] W. Bai, N. Lu, A.P. Ritenour, M.L. Lee, D.A. Antoniadis, and D.-L. Kwong, "The electrical properties of HfO₂ dielectric on germanium and the substrate doping effect," *IEEE Trans. Electron Devices*, vol. 53, no. 10, pp. 2551-2558, Oct. 2006.
- [67] S. Takagi, "Re-examination of subband structure engineering in ultra-short channel MOSFETs under ballistic carrier transport," in VLSI Symp. Tech. Dig., 2003., pp. 115-116.
- [68] J.-P. Han, E.M. Vogel, E.P. Gusev, C. D'Emic, C.A. Richter, D.W. Heh, and J.S. Suehle, "Asymmetric energy distribution of interface traps in n- and p-MOSFETs with HfO₂ gate dielectric on ultrathin SiON buffer layer," *IEEE Electron Device Lett.*, vol. 25, no. 3, pp. 126-128, Mar. 2004.
- [69] N.S. Saks and A.K. Agarwal, "Hall mobility and free electron density at the SiC/SiO₂ interface in 4H-SiC," *Appl. Phys. Lett.*, vol. 77, no. 20, pp. 3281-3283, Nov. 2000.
- [70] S. Persson, D. Wu, P.-E. Hellstrom, S.-L. Zhang, and M. Ostling, "Quantifying hole mobility degradation in pMOSFETs with a strained-Si_{0.7}Ge_{0.3} surface-channel under an ALD TiN/Al₂O₃/HfAlO_x/Al₂O₃ gate stack," *Solid-State Electron.*, vol. 48, no. 5, pp. 721-729, May 2004.
- [71] R.P.H. Chang and A.T. Fiory, "Inversion layers on germanium with low-temperature-deposited aluminum-phosphorus oxide dielectric films," *Appl. Phys. Lett.*, vol. 49, no. 22, pp. 1534-1536, Dec. 1986.
- [72] G. Apostolopoulos, G. Vellianitis, A. Dimoulas, J. C. Hooker, and T. Conard, "Complex admittance analysis for La₂Hf₂O₇/SiO₂ high-k dielectric stacks," *Appl. Phys. Lett.*, vol. 84, no. 2, pp. 260-262, Jan. 2004.
- [73] E.H. Nicollian and J.R. Brews, MOS (Metal Oxide Semiconductor) Physics and Technology. New York: Wiley, 2003.
- [74] R.M. Wallace and G. Wilk, "Alternative gate dielectrics for microelectronics," MRS Bull., vol. 27, no. 3, pp. 192-197, Mar. 2002.
- [75] E.P. Gusev, V. Narayanan, and M.M. Frank, "Advanced high-k dielectric stacks with polySi and metal gates: recent progress and current challenges," *IBM J. Res. Develop.*, vol. 50, no. 4/5, pp. 387-410, Jul./Sept. 2006.
- [76] Intel Corporation, Press Release, Jan. 27, 2007.
- [77] L.-A. Ragnarsson, N.A. Bojarczuk, M. Copel, E.P. Gusev, J. Karasinski, and S. Guhab, "Physical and electrical properties of reactive molecular-beam-deposited aluminum nitride in metal-oxide-silicon structures," J. Appl. Phys., vol. 93, no. 7, pp. 3912-3919, Apr. 2003.

- [78] C.O. Chui, L. Kulig, J. Moran, W. Tsai, and K.C. Saraswat, "Germanium n-type shallow junction activation dependences," *Appl. Phys. Lett.*, vol. 87, pp. 091909-1-3, Aug. 2005.
- [79] H. Yano, T. Hirao, T. Kimoto, and H. Matsunami, "A cause for highly improved channel mobility of 4H-SiC metal-oxide-semiconductor field-effect transistors on the (1120) face," *Appl. Phys. Lett.*, vol. 78, no. 3, pp. 374-376, Jan. 2001.
- [80] K.H. Kim, D.B. Farmer, J.-S.M. Lehn, P.V. Rao, and R.G. Gordon, "Atomic layer deposition of gadolinium scandate films with high dielectric constant and low leakage current," *Appl. Phys. Lett.*, vol. 89, pp. 133512-1-3, Sep. 2006.